HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY, GEORGIA SPECIAL CALLED MEETING

December 21, 2010

7:30a.m.

AGENDA

I.	Call to Order	Ralph Rosenberg
II.	Open Meeting and Establish a Quorum	Ralph Rosenberg
III.	Executive Session	Ralph Rosenberg
IV.	Action Items, if any	Ralph Rosenberg
٧.	Adjournment	Ralph Rosenberg

MINUTES OF HOSPITAL AUTHORTY OF ALBANY-DOUGHERTY COUNTY SPECIAL CALLED MEETING December 21, 2010

Attendees:

Authority board Members: Ralph Rosenberg, Lamar Reese, Wilhelmina Hall, Charles Lingle, John Inman, Jr. M.D., Rev. H.B. Johnson, Fred Ghiglieri. Among those also present were Joel Wernick, Kerry Loudermilk, Joe Austin, Tommy Chambless, Annette Allen (recorder), regular legal counsel to the Authority, James Reynolds and Huff Croxton and special legal counsel to the Authority Robert Baudino and Kari Middleton.

Absent:

Jyotir Mehta, M.D., Rev. Eugene Sherman(Rev. unavoidability out of town)

Called To Order:

The special called meeting was called to order by Chairman, Rosenberg at 7:33 a.m. in the Willson

Board Room of Phoebe Putney Memorial Hospital

Topic/Discussion	Conclusion/Evaluation	Recommendation/Action	Follow-Up
Open Meeting and Establish a Quorum Mr. Rosenberg welcomed the Authority and he established that a quorum was present. Mr. Rosenberg also recognized and thanked Ms. Wilhelmina Hall for her many years of loyal and able service to the Authority. Mr. Rosenberg pointed out that Ms. Hall was rotating off the Authority in Jan 2011 and he also congratulated Dr. Lingle and Mr. Ghiglieri for their respective re-appointments by the			
Dougherty County Commission			
Approval of the Agenda The agenda was reviewed.		A motion was made and seconded to approve the agenda. The motion passed	

Topic/Discussion	Conclusion/Evaluation	Recommendation/Action	Follow-Up
		unanimously.	
Closing of the Meeting:		Dr. Lingle made a motion to close the meeting to discuss the future acquisition of real estate by the Authority; to discuss a potentially commercially valuable plan proposal or strategy that may be of competitive advantage in the operation of the Hospital or its medical facilities; and privileged consultation with legal counsel, including consultation pertaining to pending or potential litigation, settlement, claims, administrative proceedings or other judicial actions @ 7:34a.m. Ms. Hall seconded the motion.	
		Mr. Rosenberg polled the individual Authority members whose votes are shown below: Ralph Rosenberg Yes Charles Lingle Yes John Inman, Jr.MD. Yes Rev. H.B. Johnson Yes Fred Ghiglieri Yes Lamar Reese Yes Dr. Eugene Sherman absent Wilhelmina Hall Yes	

Topic/Discussion	Conclusion/Evaluation	Recommendation/Action	Follow-Up
		Dr. Jyotir Mehta - absent The motion passed unanimously.	
The open session reconvened			
Resolutions:			
At the request of Mr. Rosenberg, Mr. Reynolds presented and reviewed in detail, for consideration by the Authority certain Resolutions(attached) entitled: Resolutions of the Hospital Authority of Albany- Dougherty County, Georgia: Approving Proposed Transaction with Palmyra Park Hospital, Inc.; Authorizing Waiver of Lease Provision; Approving Management Agreement; And Declaring Official Intent Relative to Possible Tax Exempt Financing.		Lamar Reese made the motion to adopt and approve the aforementioned Resolutions. Rev. Johnson seconded the motion. The motion was seconded and passed unanimously.	
Mr. Reynolds next presented and reviewed another Resolution (attached) entitled: Resolution of The Hospital Authority of Albany-Dougherty County, George Regarding Voluntary Payments In Lieu of Taxes.		Mr. Ghiglieri made the motion to adopt and approve the Resolution. The motion was seconded and passed unanimously.	
Closing Remarks Mr. Rosenberg stated the next meeting will be in February 2011 and the calendar of quarterly meetings will be mailed.			

Topic/Discussion	Conclusion/Evaluation	Recommendation/Action	Follow-Up
Adjournment		With there being no further business, the meeting, adjourned	
		at 9;07a,m.	.,

STATE OF GEORGIA COUNTY OF DOUGHERTY

AFFIDAVIT RELATIVE TO CLOSED MEETING

Personally appeared before the undersigned, RALPH S. ROSENBERG, who having been duly sworn, deposes and states as follows:

- I am over the age of 18 years, I am suffering under no disabilities and I am competent to testify to the matters contained herein.
- 2. I am the Chairperson of the Board of the Hospital Authority of Albany-Dougherty County, Georgia (the "Authority")
- 3. On the 21st day of December, 2010, at a meeting of the Authority Board, a motion was duly approved for the Authority Board to go into closed session for the purposes shown in a, b and c below, and the Authority did go into closed session for such purposes:
 - a. To discuss the future acquisition of real estate by the Authority. Permitted under O.C.G.A §50-14-3(4).
 - b. To discuss a potentially commercially valuable plans, proposal or strategy that may be of competitive advantage in the operation of the Hospital or its medical facilities. Permitted under O.C.G.A. § 31-7-75.2.
 - c. Privileged consultation with legal counsel, including consultation pertaining to pending or potential litigation, settlement, claims, administrative proceedings or other judicial actions. Permitted under O.C.G.A. § 50-14-2(1).
- 4. To the best of my knowledge and belief, the business conducted during the closed session was devoted solely to the above matters, for which the closed session was called.

This the 21st day of December, 2010.

Chairperson Chairperson

Sworn to and subscribed before me this

21st day of December, 2010.

Dougherty County, Georgia

My Commission expires:

Annette M. Allen

Notary Public - State of Georgia Qualified in Dougherty County My Commission Expires January 28, 2013 RESOLUTIONS OF THE HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY, GEORGIA: APPROVING PROPOSED TRANSACTION WITH PALMYRA PARK HOSPITAL, INC.; AUTHORIZING WAIVER OF LEASE PROVISION; APPROVING MANAGEMENT AGREEMENT; AND DECLARING OFFICIAL INTENT RELATIVE TO POSSIBLE TAX EXEMPT FINANCING

WHEREAS, the Hospital Authority of Albany-Dougherty County, Georgia (the "Authority") is a public body, corporate and politic of the State of Georgia, created under the Hospital Authorities Law, O.C.G.A. Section 31-7-70, et seq (the "Act"), and jointly activated by the City of Albany, Georgia the ("City") and Dougherty County, Georgia (the "County") pursuant to a resolution jointly adopted by the City and the County on July 22, 1941 and August 4, 1941;

WHEREAS, the Act empowers the Authority, among other things, to acquire by purchase, lease or otherwise, "Projects" as defined in the Act, to operate the same, and if it so chooses, to finance part or all of the cost of the acquisition, equipping or upgrading such Projects by the issuance and sale of its negotiable Revenue Anticipation Certificates or other debt instruments;

WHEREAS, Phoebe Putney Health System, Inc. ("Health System") is a Georgia non-profit corporation and it wholly controls its affiliates, Phoebe Putney Memorial Hospital, Inc. ("PPMH") and Phoebe North, Inc. ("PNI"), each of which is also a Georgia non-profit corporation;

WHEREAS, at the recommendation of and with the financial assistance and backing of Health System, the Authority has determined that it wishes to take advantage of the opportunity to acquire substantially all of the assets (the "Purchased Assets") used exclusively in the operation of Palmyra Medical Center, as well as other facilities or real estate owned or leased by Palmyra Park Hospital, Inc. or its affiliates, all as more fully set forth in the Asset Purchase Agreement (the "APA") presented at this meeting of the Authority, and with the Purchased Assets being more fully defined and described in the APA;

WHEREAS, the Authority has found that the Purchased Assets constitute a Project under the Act, that the Purchased Assets (with the exception of one commercial lot in Terrell County, Georgia which by designation of the Authority will be acquired by Health System or an affiliated designee) lie within the Authority's "area of operation", as defined in the Act and that pursuant to the Act, the Authority may lawfully acquire and operate the Purchased Assets;

WHEREAS, the Authority has been assured by Health System that it can and will provide all required funds and capital resources needed by the Authority to carry out the Authority's financial obligations under the APA and to operate the Purchased Assets thereafter;

WHEREAS, Robert J. Baudino of the Soverign Group, LLC and Baudino Law Group, P.L.C., has provided the Authority a comprehensive review and explanation of the APA together with its exhibits and schedules, and executive officers of Health System have further reviewed with the Authority certain perceived risks, rewards, strategic and business aspects of entering into the APA, completing the acquisition, as well as the future management and operation of the Purchased Assets;

WHEREAS, the Authority is a party, as Lessor (sometimes referred to therein as the "Transferor"), to that certain Lease and Transfer Agreement, dated as of December 11, 1990, as amended (as so amended, the "Lease"), along with PPMH, as Lessee; and

WHEREAS, in the event the Authority successfully acquires the Purchased Assets and thereafter operates the same, it wishes to clarify with PPMH, that such operation of the Purchased Assets will not be considered a breach of the Authority's covenant to PPMH under Section 4.21 of the Lease, which reads as follows:

Section 4.21. <u>Competition with Hospital</u>. During the term of this Agreement, Transferor shall not own, manage, operate or control or be connected in any manner with the ownership, management, operation or control of any hospital or other health care facility other than the Hospital in Albany, Georgia and in Dougherty County, Georgia and in any other areas of operation of the Transferor as permitted under the act.

WHEREAS, the Authority wishes to retain the future option of utilizing tax exempt financing for the financing or reimbursement of the qualified capital expenditures made in connection with the proposed acquisition of the Purchased Assets and any related expenses, including the rehabilitation, equipping, modernizing or other such capital expenditures associated therewith, and accordingly, the Authority wishes to make a proper declaration of the Authority 's official intent with respect to such financing;

WHEREAS, it is the desire of the Authority that upon purchase of the Purchased Assets, the Authority will operate the same utilizing PNI as its agent, pursuant to a management contract (the "Management Agreement"), a copy of which has been presented and reviewed at this meeting.

NOW, THEREFORE, the Authority hereby adopts the following Resolutions:

1. Having reviewed and considered the same, the Authority has found and determined that the acquisition and operation of the Purchased Assets as contemplated in the APA is consistent with and advances the Authority's

mission and purposes of enhancing and increasing the health care facilities available to the citizens of Albany and Dougherty County, Georgia and/or reducing or controlling the cost of health care in Albany and Dougherty County, Georgia, in multiple ways, including:

- a. alleviating current as well as long term lack of capacity issues encountered in the operations of Phoebe Putney Memorial Hospital and providing for increased services based on the core needs of our citizens, and doing so in a manner which will be cost effective, more practical and less disruptive than attempting to expand the Authority's current facilities;
- b. helping assure and increase full and continuing community access to the acquired healthcare facilities and the associated health care services provided therein;
- c. the ability to lower or contain costs through economies of scale, as well as other synergies brought about as a result of the acquisition of the Purchased Assets; and
- d. increased opportunity for providing for our population through the provision of additional quality outcome collaborative healthcare, which can be provided at a lower total cost and in a manner preferred by third party payees such as Medicare.
- 2. The Authority hereby approves of the APA and authorizes and directs its Chairman to join with Health System and on behalf of the Authority to execute and deliver the APA as presented at this meeting. A copy of the APA as presented at this meeting is to be included with the minute book copy of this Resolution as an attachment.
- 3. The appropriate officers of the Authority are hereby authorized and directed to take any and all further action to execute and deliver on behalf of the Authority any and all additional documents or certificates which may be deemed necessary, desirable or appropriate in order to fully implement and carry out the anticipated Closing as required of the Authority under the APA, including execution of all closing documents and certificates and the payment of the Base Purchase Price of \$195,000,000, as the same may ultimately be adjusted pursuant to the APA and the assumption of the Assumed Liabilities as defined in the APA and the designation of Health System or an affiliated entity to take title to the Terrell County lot otherwise included in the Purchased Assets.

- 4. The Authority hereby approves the Management Agreement and authorizes and directs its Chairman, Vice-Chairman and the Secretary or Assistant Secretary to execute and deliver the Management Agreement in the name of and on behalf of the Authority, subject to such changes, insertions and omissions as may be approved by the authorized officer of the Authority executing the same. The proper execution of the Management Agreement shall be conclusive evidence of the approval of any such changes, insertions or omissions. A copy of the Management Agreement as presented at this meeting is to be included with the minute book copy of this Resolution as an attachment.
- 5. The Chairman of the Authority or in his absence, the Vice-Chairman of the Authority, is hereby authorized and directed to take all reasonable action to seek to join with PPMH in mutually waiving the provisions of Section 4.21 of the Lease as respects the Authority's operation of the Purchased Assets, and such officer is hereby authorized to join with PPMH to execute a written waiver of such provision.
- 6. The Authority hereby makes a declaration of its official intent that it reasonably expects to reimburse all or part of the expenditures for the acquisition of the Purchased Assets as contemplated in the APA, as well as capital expenditures, if any, incurred in the equipping or retrofitting or rehabilitating any part or all of such acquired property. The maximum principal amount of tax-exempt debt expected to be issued for such purposes as of the date of adoption of this Resolution is \$200,000,000. This declaration of official intent is intended to constitute a "declaration of official intent" under Treasury Regulations 1.150-2.

Duly adopted this 21st day of December, 2010.

HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY, GEORGIA

RV.

RALPH S. ROSENBERG, CHAIRMAN

[SEAL]

ATTEST:

REV. EUGENE SHERMAN, SECRETARY

JOHN S. INMAN, JR., ASST.

FACT Sheet for Acquisition of Palmyra Medical Center

Why is the Authority purchasing PMC?

Phoebe Putney Memorial Hospital is facing exceptional challenges of capacity to meet the growing healthcare needs of Southwest Georgians. Their needs were brought to us by the Health System, which asked that we seek to purchase Palmyra Medical Center. This assimilation allows us to address the capacity issues in the most efficient and cost effective manner. The unification of these institutions allows us to enhance access to quality medical while reducing costly duplication of services.

What will the name be?

Phoebe North. A Georgia not-for-profit corporation has been formed called Phoebe North, Inc. It is a wholly owned subsidiary of Phoebe Putney Health System Inc., and it will manage the hospital for the Authority.

What will happen to Palmyra employees?

The Phoebe Family will welcome all PMC employees who meet the current employment criteria for PPHS.

What will the campus be used for?

Although there are a number of viable options for use and development of the PMC campus as a part of the PPHS system, Phoebe management intends to quickly engage board, medical staff and leadership in a discussion about potential options. We will center on programs and services that enhance capacity, and improve throughput and clinical program development.

About competition and choice, will cost of care be impacted?

There is absolutely no expectation that cost of care will increase for our patients and communities as a result of this purchase. Instead there will be many opportunities to consolidate services, thereby reducing the burden on the consumer often brought about by duplication. Further benefits will evolve because PPHS can purchase supplies and materials in greater quantity and better pricing.

What becomes of the tax digest when the facility is converted to a not-for-profit hospital?

Technically, the hospital and its campus and holdings will become part of a governmental entity, the Hospital Authority of Albany and Dougherty County, GA., and the management of the hospital will be by the not-for-profit entity, Phoebe North. Neither is subject to ad valorem or sales taxes under Georgia law. However the authority in approving the action, made clear its intent that the revenue base of Albany, Dougherty County and the Dougherty County School Board not be adversely affected by the transaction. Its current lessee, Phoebe Putney Memorial Hospital, and its affiliates, including PPHS, have agreed that the Hospital Authority will formulate and enter into contracts with the taxing authorities to make voluntary payments to be used by those governing bodies for the same purposes that they use tax revenues. Phoebe officials have indicated an intent to accomplish that immediately and they have stated their intent is to be certain there is no gap in the voluntary payment and the payment will be no less than what would have been received had HCA/Palmyra continued to be a part of the tax base.

Will there be a separate board?

Yes, A newly created corporate entity, Phoebe North, which is wholly owned by Phoebe Putney Health System, will manage the hospital. As a separate not for profit corporation, it will have its own community-based board.

With years of rivalry, how did this happen?

Ownership was the foundation of the rivalry. Each of the "owners" had different needs that had to be met as benefits of ownership. The "owners" of Palmyra were investors (public at times, private at times) who had expectations of return on their investment in the form of cash.

The "owners" of Phoebe are the citizens of Dougherty County. Revenues that exceeded expenses (ie, profits) have been re-invested over the years resulting in new construction, advancements in technology, growth of services, addition of new medical specialties (recruitment of new physicians), and – of crucial economic importance, an expanding workforce – all to the benefit of Albany, Dougherty County, and Southwest Georgia.

Typical of normal marketplace behavior, for each of the two hospitals to be successful they had to compete with each other for business. There was a drive for each to match services provided by the other. This results in duplication of services, which adds cost to the marketplace, and is often unnecessary, especially in healthcare. If this duplication of services followed normal marketplace behavior, this could result in lower prices. The healthcare marketplace is atypical because pricing in competitive markets is often higher than in markets where the delivery of services is unified. Reducing duplication of operational support services and avoidance of redundant technology investments results in better cost control.

With payment reform driving healthcare providers to produce better outcomes with tighter cost controls, consolidation is necessary within marketplaces. It is simply too costly to continue in the business as usual approach. Unification of hospitals is not an Albany phenomenon; it is a national trend.

So the time has come for the rivalry to become a part of history. It no longer makes business sense. Elimination of the rivalry benefits the community.

What about physicians who may have been a part of the rivalry?

Because the physicians are key drivers of the culture of a hospital, hospitals that have different medical staffs often have different cultures. "Rivalries" between hospitals can emerge as a result of these physician-driven cultural differences. Here in Albany, physicians have historically been medical staff members at both hospitals. While some physicians have been aware of the rivalry, they have not been at the foundation of the rivalry, which historically is a business issue.

Will the Dougherty County Commission have any involvement with Phoebe North?

It is the same relationship that currently exists with PPMH. The Hospital Authority, which owns the assets of PPMH, is appointed by the Dougherty County Commission and will run the assets of Palmyra.

Will PMC doctors have changes?

No. Current medical staff will remain in place.

Will there by a change in services?

Our management will begin quickly evaluating options. Our intent is to make all entities the best they can be to improve access to services and eliminate unnecessary duplications.

What about the lawsuits between Phoebe and PMC.

All lawsuits will be dismissed.

Who will be the CEO of PMC?

An interim administrator will be named to serve the Phoebe North campus.

How will I apply for a job at PMC?

After the transition, go to the Phoebe website at Phoebeputney.com and click on careers.

If a doctor is not on staff at PMC, will he or she now be?

There will be no changes to medical staff at this time.

Why did it cost \$195 million?

PPHS must invest to increase capacity needs and to be able to recruit and attract needed physician, nursing and allied health employees. This creates a strong economic foundation and enhances the long-term security of assuring access to care into the future. The market sets cost and PPHS acknowledges it is a significant amount. But the cost is reasonable and within the hospital industry parameters for the value received by PPHS.

ASSET PURCHASE AGREEMENT

 \mathbf{BY}

AND

AMONG

THE HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY PHOEBE PUTNEY HEALTH SYSTEM, INC.

PHOEBE NORTH, INC.

AND

PALMYRA PARK HOSPITAL, INC.

Dated as of December 21, 2010

TABLE OF CONTENTS

			Page	
1.	DEEL	NITIONS	1	
1.	1.1	Definitions		
	1.2	Interpretation		
	1.2	The production,	,	
2.	SALE OF ASSETS AND CERTAIN RELATED MATTERS			
	2.1	Sale of Purchased Assets		
	2.2	Excluded Assets		
	2.3	Assumed Liabilities		
	2.4	Excluded Liabilities		
	2.5	Purchase Price		
	2.6	Determination of the Price Adjustment Amounts		
	2.7	Post-Closing Determinations of Price Adjustment Amounts	17	
	2.8	Proration	18	
3.		SING		
	3.1	Closing	19	
	3.2	Actions of Seller at Closing.	19	
	3.3	Actions of Buyer at Closing		
	3.4	Additional Acts	21	
4.	REPF	RESENTATIONS AND WARRANTIES OF SELLER	21	
	4.1	Incorporation; Qualification and Capacity		
	4.2	Corporate Powers; Consents; Absence of Conflicts With Other Agreements, Etc	22	
	4.3	No Outstanding Rights	22	
	4.4	Purchased Assets	22	
	4.5	Binding Agreement	22	
	4.6	Financial Information	23	
	4.7	Permits and Approvals	23	
	4.8	Intellectual Property	24	
	4.9	Medicare Participation/Accreditation	24	
	4.10	Regulatory Compliance		
	4.11	Assumed Contracts		
	4.12	Real Property		
	4.13	Personal Property	26	
	4.14	Insurance		
	4.15	Employee Benefit Plans		
	4.16	Employees and Employee Relations		
	4.17	Litigation or Proceedings		
	4.18	Tax Matters		
	4.19	Environmental Matters		
	4.20	Immigration Act		
	4.21	WARN Act		
	4.22	OSHA		
	4.23	Inventory	30	
	4.24	Absence of Changes		
	4.25	Solvency	31	
	4.26	Statements True and Correct	31	

	4.27	Medical Staff	31
5.	REPR	ESENTATIONS AND WARRANTIES OF BUYER	31
	5.1	Authority Capacity: Corporate Capacity.	31
	5.2	Powers: Consents: Absence of Conflicts With Other Agreements, Etc	, 32
	5.3	Binding Effect	32
	5.4	Litigation	32
	5.5	Ruver Acknowledgements	, 33
	5.6	Availability of Funds: Solvency	<i>33</i>
	5.7	Antitrust Matters	3 <i>3</i>
	5.8	Statements True and Correct	33
6.	COVE	ENANTS OF SELLER AND BUYER	34
	6.1	Completion of Schedules; Information; Assumed Contracts and Rejected Contracts.	34
	6.2	Operations	
	6.3	Negative Covenants	3 1
	6.4	Notification of Certain Matters	38
	6.5	HSR Notification	35
	6.6	Approvals: Consented Assignment	39
	6.7	Additional Financial Information	4t
	6.8	Books and Records.	4۱
	6.9	Non-Solicitation	4(
7.	CONI	DITIONS PRECEDENT TO OBLIGATIONS OF BUYER	4
	7.1	Compliance With Covenants	4
	7.2	Opinion of Counsel	4
	7.3	Pre-Closing Confirmations	4
	7.4	Action/Proceeding	42
	7.5	Title Policies and Surveys	4.
	7.6	Representations and Warranties	4.
8.	CON	DITIONS PRECEDENT TO OBLIGATIONS OF SELLER	4
	8.1	Compliance With Covenants	4
	8.2	Opinion of Counsel	4
	8.3	Action/Proceeding	4
	8.4	Pre-Closing Confirmations	4
	8.5	Representations and Warranties	4.
9.		NSITIONAL ARRANGEMENTS	4
	9.1	PIP; Transition Patients; Government Patient Receivables; and Misdirected	1
		Payments	4
	9.2	Seller's Cost Reports	4
	9.3	Employees.	4
10.		ITIONAL AGREEMENTS	4
	10.1	Break-Up Fee, Rescission Fee	4
	10.2	Allocation of Purchase Price.	5
	10.3	Termination Prior to Closing.	5
	10.4	Post-Closing Access to Information.	5
	10.5	Preservation and Access to Records After the Closing	5

	10.6	Capital Expenditures	. 52
	10.7	Reproduction of Documents	. 52
	10.8	Cooperation on Tax Matters	. 52
	10.9	Litigation Support	. 52
	10.10	HCA Post-Closing Services to be Provided Buyer	, 52
	10.11	No-Shop	. 53
	10.12	Post-Closing Operations	. 53
	10.13	Maintenance of Programs	. 53
	10.13	Provision of Certain Employees	. 53
	10.15	Non-Competition.	. 53
	10.15	Settlement and Release	. 54
	10.10	Sourcinent and Resease	
11.	MDEN	MNIFICATION	. 55
	11.1	Indemnification by Seller	. 55
	11.2	Indemnification by Buyer	. 56
	11.3	Limitation	. 56
	11.3	Notice and Control of Litigation	. 57
	11.5	Survival	. 57
	11.5	Materiality; De Minimis Claims	. 58
	11.7	Exclusive Remedy	. 58
	11.7	Mitigation	. 58
	11.9	Limitation on Damages	. 58
		Treatment of Payments	. 58
	11.10	Treatment of Fayments	
12.	GENE	RAL	58
. ~.	12.1	Consents Approvals and Discretion	58
	12.2	Legal Fees and Costs	59
	12.3	Governing Law; Arbitration	59
	12.4	Benefit; Assignment	60
	12.5	Accounting Date	61
	12.5	No Brokerage.	61
	12.7	Cost of Transaction	61
	12.7	Public Announcements; Confidentiality	61
	12.8	Waiver of Breach	62
	12.10	Notice	62
	12.11	Severability	63
	12.11	No Inferences	63
	12.12	Divisions and Headings	64
		No Third-Party Beneficiaries	64
	12.14	Tax and Medicare Advice and Reliance	64
	12.15	1 ax and Medicare Advice and Renance	64
	12.16	Entire Agreement; Amendment	0 .1 64
	12.17	Knowledge	UM 1/A
	12.18	FIRPTA Withholding	UH
	12.19	Other Owners of Purchased Assets	04
	12.20	Disclaimer of Warranties	03
	12.21	Guarantee	03
	12.22	Schedules	00

EXHIBITS

Exhibit A Form of Assignment and Assumption Agreement

Exhibit B Form of Partial Assignment and Assumption Agreement

Exhibit C Form of General Bill of Sale and Assignment

Exhibit D-1 Owned Real Property
Exhibit D-2 Leased Real Property
Exhibit E Other Receivables

Exhibit F Prepaid Expenses and Excluded Prepaid Expenses

Exhibit G Plaintiff's Motion for Order of Dismissal Without Prejudice

Exhibit H Appellant Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center's Motion

for Permission to Withdraw Appeal

Exhibit I Appellant Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center's Motion

for Permission to Withdraw Appeal

Exhibit J Letter withdrawing Certificate of Need application # 2008-081

SCHEDULES

Schedule 2.2	Excluded Assets
Schedule 4.3	No Outstanding Rights
Schedule 4.6	Historical Financial Information
Schedule 4.9	Medicare Participation/Accreditation
Schedule 4.10	Regulatory Compliance
Schedule 4.11	Defaults Under Assumed Contracts
Schedule 4.14	Insurance
Schedule 4.15	Employee Benefit Plans
Schedule 4.16	Employees and Employee Relations
Schedule 4.17(a)	Litigation or Proceedings
Schedule 4.17(c)	Outstanding Judgments, Orders or Decrees
Schedule 4.18	Tax Matters
Schedule 4.19	Environmental Matters
Schedule 4.21	WARN Act
Schedule 4.24	Absence of Changes
Schedule 4.24(d)	Capital Expenditures
Schedule 5.2	Permits and Approvals
Schedule 6.1(e)	Contracts to be Partially Assigned or Modified
Schedule 6.2	Exceptions to Affirmative Operating Covenants
Schedule 6.3	Exceptions to Negative Covenants
Schedule 7.5	Title Insurance
Schedule 10.2	Allocation of Purchase Price

4016376.6

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "Agreement") is made and entered into as of December 21, 2010 by and among the HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY, a hospital authority organized and existing pursuant to Georgia Code Annotated §§ 31-7-70 et seq. ("Buyer"), and PALMYRA PARK HOSPITAL, INC., a Georgia corporation ("Seller"). PHOEBE PUTNEY HEALTH SYSTEM, INC., a Georgia nonprofit corporation ("PPHS"), joins in this Agreement solely for the purposes set forth in Section 6.9, Section 10.1 and Section 12.21. PHOEBE NORTH, INC., a Georgia nonprofit corporation ("PNI"), joins this Agreement solely for the purposes set forth in Section 6.9 and Section 9.3(b).

WITNESSETH:

- A. Seller owns the assets and operations of Palmyra Medical Center, Albany, Georgia (the "Hospital") and the other Facilities (as defined below).
- B. Buyer desires to acquire substantially all of the assets used exclusively in the operation of the Hospital and the other Facilities, and assume certain liabilities relating to the Hospital and the other Facilities, all in accordance with the terms and conditions set forth herein.
- PPHS has formed PNI as its wholly controlled affiliate corporation. PNI will manage the C. Hospital and the other Facilities for Buyer pursuant to a written management agreement between Buyer and PNI. Pursuant to that management agreement and Section 9.3 of this Agreement, PNI will employ at Closing substantially all of Seller's employees employed at the Facilities as of Closing. Seller and PPHS as experienced operators of hospitals recognize the risks and financial undertaking required of Buyer to acquire and own and of PNI to operate the Hospital for Buyer. As a material inducement to Seller to enter into this Agreement, pursuant to Section 10.1, and subject to the terms and provisions of Section 10.1, PPHS will compensate Seller in the event the transactions contemplated by this Agreement are not consummated or if the parties are required to rescind the transactions contemplated by this Agreement after the consummation of such transactions. As a further material inducement to Seller to enter into this Agreement, PPHS will guarantee the financial obligations of Buyer and PNI under this Agreement and any ancillary documents hereto pursuant to the terms of Section 12.21, and PNI will offer employment to Seller's employees as contemplated by Section 9.3 of this Agreement. PPHS acknowledges that Seller would not enter into this Agreement but for PPHS's obligations under Section 10.1 and PPHS' financial guarantee of Buyer's obligations under this Agreement and any ancillary documents hereto. PNI acknowledges that Seller would not enter into this Agreement but for PNI's offer of employment to Seller's employees as contemplated by this Agreement.

NOW, THEREFORE, for and in consideration of the premises, and the agreements, covenants, representations and warranties hereinafter set forth, and other good and valuable consideration, the receipt and adequacy of which are forever acknowledged and confessed, the parties hereto agree as follows:

1. **DEFINITIONS**

1.1 <u>Definitions</u>. As used herein the terms below shall have the following meanings:

"AAA" has the meaning set forth in Section 12.3.

"AAA Rules" has the meaning set forth in Section 12.3(b).

"Accrued EIB" has the meaning set forth in Section 2.3(b).

"Accrued PTO" means Seller's liability as of the Closing Date for "paid time off" with respect to the employees of Seller and its Affiliates who are employed at the Facilities, but excluding Seller's liability for "paid time off" in respect of employees of Seller and its Affiliates who are not offered employment or who do not accept offers of employment by PNI as of Closing as contemplated by Section 9.3.

"Accrued PTO Amount" means the amount of Seller's liability as of the Closing Date for Accrued PTO, as determined (or agreed upon by Buyer and Seller) in accordance with Section 2.6 or Section 2.7.

"ADA Liabilities" means (a) any obligation or requirement to modify, repair, replace or otherwise alter any portion of the Facilities in order to cause the Facilities to be compliant with the Americans with Disabilities Act, as amended, (b) any liabilities and obligations related to claims, actions, suits, audits, proceedings or investigations brought after the Closing against any Person that owns, leases, operates or manages any of the Facilities after the Closing or any Affiliate of any such Person (including Buyer, PPHS and PNI) that seek to require, or require, modification, repair, replacement or alteration of any portion of the Facilities so that the Facilities are compliant with the Americans with Disabilities Act, as amended, (c) any judgments, orders or decrees entered with respect to any of the claims, actions, suits, audits, proceedings or investigations described in clause (b), above, and/or (d) any obligation or requirement to modify, repair, replace or otherwise alter any portion of the Facilities pursuant to any settlement agreement, judgment, decree or order listed in Schedule 4.17(c) with respect to the "Access Now Litigation" (as defined in Schedule 4.17(a)).

"Additional Adjustment Amount" has the meaning set forth in Section 2.6(e).

"Affiliate" means, as to the Person in question, any Person that directly or indirectly controls, is controlled by, or is under common control with, the Person in question and any successors or assigns of such Person; and the term "control" means possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person whether through ownership of voting securities or membership interests, the right to appoint the Person's directors or trustees, by contract or otherwise; provided that "Affiliate" shall not include (a) officers or directors of the Authority, PPHS, PNI, Seller or its Affiliates or (b) shareholders of HCA. The Authority is not an Affiliate of PNI or PPHS. PPHS and PNI are Affiliates of one another.

"Agency Receivables" means rights to settlement and retroactive adjustments, if any, for open cost reporting periods ending on or prior to the Closing (whether open or closed) arising from or against the U.S. Government under the terms of the Medicare program or TRICARE and against any state under its Medicaid program and against any third-party payor programs that settle on a cost report basis.

"Agreement" means this Agreement as amended or supplemented together with the Exhibits and Schedules hereto.

"Antitrust Claim" has the meaning set forth in Section 10.16(a)(i).

"Applicable Rate" means the "prime rate" as quoted in the "Money Rates" section of <u>The Wall</u> Street Journal on the Closing Date.

"Approval" means any approval, authorization, consent, notice, qualification or registration, or any extension, modification, amendment or waiver of any of the foregoing, of or from, or any notice,

statement, filing or other communication to be filed with or delivered to, any Governmental Entity (other than Buyer or its Affiliates).

- "Arbitration Notice" has the meaning set forth in Section 12.3(b).
- "Arbitrators" has the meaning set forth in Section 12.3(b).
- "Assignment and Assumption Agreement" means the Assignment and Assumption Agreement attached hereto as Exhibit A.
- "Assumed Contract" means (a) any Contract that is not an Excluded Contract and (b) to the extent partially assigned as contemplated by Section 6.1(e), any agreement, lease, sublease, license, sublicense, promissory note, evidence of indebtedness, or other contract that is partially assigned to Buyer.
- "Assumed Indebtedness" means Seller's liability for any indebtedness for borrowed money or capitalized lease obligations assumed or paid by Buyer, if any, including the current liability portion thereof.
- "Assumed Indebtedness Amount" means (a) the amount of Seller's liability as of the Closing Date for the Assumed Indebtedness, if any, as determined (or agreed upon by Buyer and Seller) in accordance with Section 2.6 or Section 2.7, plus (b) the Accrued PTO Amount.
 - "Assumed Liabilities" has the meaning set forth in Section 2.3.
 - "Authority" has the same meaning as Buyer as set forth in the Preamble.
 - "Balance Sheet Date" means November 30, 2010.
 - "Base Purchase Price" has the meaning set forth in Section 2.5.
 - "Basket Amount" has the meaning set forth in Section 11.3(b).
 - "Break-Up Fee" has the meaning set forth in Section 10.1(a).
 - "Buyer" has the meaning set forth in the Preamble.
 - "Buyer Indemnified Parties" has the meaning set forth in Section 11.1.
 - "Buyer Losses" has the meaning set forth in Section 11.3(c).
- "Capital Expenditures Amount" means the amount of capital expenditures that Seller or any of its Affiliates incurs with respect to the Facilities or the Purchased Assets between March 1, 2011 and the Closing Date for routine capital expenditures incurred in the ordinary course of business; provided, however, that the Capital Expenditures Amount shall not exceed \$170,000.00 multiplied by the number of months between March 1, 2011 and the Closing Date (prorated for any partial month) except to the extent that Buyer shall have approved any capital expenditures in excess of such amount.
 - "CERCLA" has the meaning set forth in the definition of Environmental Laws.
 - "Closing" has the meaning set forth in Section 3.1.
 - "Closing Balance Sheet" has the meaning set forth in Section 2.7(b).

"Closing Date" has the meaning set forth in Section 3.1.

"Code" means the Internal Revenue Code of 1986, as amended, and the rules and regulations promulgated thereunder.

"Collateral" has the meaning set forth in Section 12.21(b)(iv).

"Competing Business" has the meaning set forth in Section 10.15(f).

"Contract" means any agreement, lease, sublease, license, sublicense, promissory note, evidence of indebtedness, or other contract, to which Seller or one of its Affiliates is a party or by which assets of the Facilities are bound which relates exclusively to the Facilities, the Purchased Assets or the operation thereof.

"Deletions" has the meaning set forth in Section 7.5.

"EEOC" means the Equal Employment Opportunity Commission.

"Employee" has the meaning set forth in Section 9.3.

"Employment Loss" has the meaning set forth in Section 4.21.

"Encumbrance" means any claim, charge, easement, restrictive covenant, encumbrance, encroachment, security interest, mortgage, lien, pledge or restriction, whether imposed by agreement, understanding, Law, equity or otherwise.

"Endorsements" has the meaning set forth in Section 7.5.

"Environmental Condition" means any event, circumstance or conditions related in any manner whatsoever to (i) the current or past presence or spill, emission, discharge, disposal, release or threatened release of any hazardous, infectious or toxic substance or waste (as defined by any applicable Environmental Laws) or any chemicals, pollutants, petroleum, petroleum products or oil ("Hazardous Materials"), into the environment; or (ii) the on-site or off-site treatment, storage, disposal or other handling of any Hazardous Material originating on or from the Real Property; or (iii) the placement of structures, materials or substances into waters of the United States; or (iv) the presence of any Hazardous Substance, including friable asbestos, in any building, structure or workplace or on any portion of the Real Property; or (v) any violation of Environmental Laws at or on any part of the Real Property or arising from the activities of Seller at the Facilities involving Hazardous Materials or the presence of any underground storage container on or in any part of the Real Property.

"Environmental Laws" means all Laws relating to pollution or the environment, including the Comprehensive Environmental Recovery, Compensation, and Liability Act, as amended, 42 U.S.C. §9601, et seq. ("CERCLA"); the Resource Conservation and Recovery Act, as amended, 42 U.S.C. §9601, et seq. ("RCRA"), the Clean Air Act, 42 U.S.C. §7401, the Occupational Safety and Health Act, 29 U.S.C. §600, et seq. ("OSHA"), the Georgia Air Quality Act, as amended, O.C.G.A. §§ 12-9-1, et seq., the Georgia Groundwater Use Act, as amended, O.C.G.A. §§ 12-5-90, et seq., the Georgia Safe Drinking Water Act of 1977, as amended, O.C.G.A. §§ 12-5-170, et seq., the Georgia Water Supply Act, as amended, O.C.G.A. §§ 12-5-470, et seq., the Georgia Water Well Standards Act, as amended, O.C.G.A. §§ 12-8-60, et seq., the Georgia Underground Storage Act, as amended, O.C.G.A. §§ 12-13-1, et seq., the Georgia Radiation Control Act, as amended, O.C.G.A. §§ 31-13-1, et seq., the Georgia Environmental Protection

Division ("GEDP") Air Quality Control Rules and Regulations, Official Compilation Rules and Regulations of the State of Georgia ("Georgia Rules") rr. 391-3-1, et seq., GEPD Groundwater Use Rules and Regulations, Georgia Rules rr. 391-3-2, et seq., GEPD Safe Drinking Water Rules and Regulations, Georgia Rules rr. 391-3-5, et seq., GEPD Hazardous Waste Management Rules and Regulations, Georgia Rules rr. 391-3-11, et seq., GEPD Underground Storage Tank Management Rules and Regulations, Georgia Rules rr. 391-3-15, et seq., GEPD Radioactive Materials Rules and Regulations, Georgia Rules rr. 391-3-17, et seq., and all other Laws relating to emissions, discharges, releases, or threatened releases of pollutants, contaminants, chemicals, pesticides, or industrial, infectious, toxic or hazardous substances or wastes into the environment (including ambient air, surface water, groundwater, land surface or subsurface strata) or otherwise relating to the processing, generation, distribution, use, treatment, storage, disposal, transport, or handling of pollutants, contaminants, chemicals, or industrial, infectious, toxic, or hazardous substances or wastes.

"ERISA" has the meaning set forth in Section 4.15(a).

"ERISA Controlled Group" has the meaning set forth in Section 4.15(b).

"Excluded Assets" has the meaning set forth in Section 2.2.

"Excluded Contracts" means (a) the Rejected Contracts, (b) the Plans, (c) any Contracts which are only available to Seller and its Affiliates, and (d) all rights and obligations existing under any agreement, lease, sublease, license, sublicense, promissory note, evidence of indebtedness, or other contract that relates to both the Facilities or the Purchased Assets (or the operations thereof) and to one or more hospitals or healthcare facilities owned or operated by Affiliates of Seller and located outside the Territory (or the assets or operations thereof) that are not expressly assigned to Buyer pursuant to Section 6.1(e).

"Excluded Inventory" means the inventory, supplies, food, pharmaceuticals, janitorial and office supplies and other disposables of Seller and its Affiliates at the Facilities that (a) are not current and usable, or (b) are not included in the determination of the Inventory Amount.

"Excluded Liabilities" has the meaning set forth in Section 2.4.

"Excluded Prepaid Expenses" means Seller's prepaid expenses, claims for refunds and rights to offset in respect thereof relating to the Facilities that (a) are not assumable or useable by Buyer, or (b) are not included in the determination of the Prepaid Expenses Amount. Exhibit F indicates the types of such assets that would be classified as Excluded Prepaid Expenses if the Closing took place as of November 30, 2010.

"Excluded Receivables" means all accounts receivable of the Seller and the Facilities (including all Agency Receivables and Government Patient Receivables) other than the Other Receivables and, subject to the provisions of Section 9.1, the Transition Patient Receivables.

"<u>Facilities</u>" means (a) the Hospital, (b) all medical office buildings owned or leased by Seller or its Affiliates and located in the Territory, and (c) all such property of every description listed in <u>Exhibit</u> D-1 or D-2 (as supplemented or amended in accordance with Section 6.1(a)).

"First Arbitrator" has the meaning set forth in Section 12.3(b).

"FTC" means the United States Federal Trade Commission.

4016376.6

"Furniture and Equipment" means all equipment (including movable equipment), vehicles, medical instruments, machines, machinery, and furniture or furnishings, wherever located, that are held or used by Seller exclusively in connection with the business or operation of the Facilities, whether or not reflected in the Reference Balance Sheet, including all such equipment, vehicles, medical instruments, machines, machinery, furniture or furnishings that have been fully depreciated for accounting purposes and any such items that were acquired for use exclusively in the business or operation of the Facilities since the Balance Sheet Date, but excluding the Excluded Assets.

"GAAP" means, subject to the exceptions and modifications described in <u>Schedule 4.6</u>, United States generally accepted accounting principles and practices as in effect from time to time applied consistently by Seller in connection with the Facilities throughout the periods involved.

"General Bill of Sale and Assignment" means the General Bill of Sale and Assignment attached hereto as Exhibit C.

"Government Patient Receivables" means all accounts receivable arising from the rendering of services and provision of medicine, drugs and supplies to patients at the Facilities through the Closing Date and relating to Medicare, Medicaid, TRICARE and other third party patient claims due from beneficiaries or governmental third party payors; provided, however, that subject to the provisions of Section 9.1, Transition Patient Receivables are not within the definition of Government Patient Receivables.

"Government Programs" has the meaning set forth in Section 4.9(a).

"Governmental Entity" means any government or any agency, bureau, board, directorate, commission, court, department, official, political subdivision, tribunal or other instrumentality of any government, whether federal, state or local, domestic or foreign.

"Hazardous Materials" has the meaning set forth in the definition of Environmental Condition.

"HCA" means HCA Inc., a Delaware corporation.

"Historical Financial Information" has the meaning set forth in Section 4.6.

"Hospital" has the meaning set forth in the Recitals.

"Hospital Authorities Law" means the Hospital Authorities Law, Georgia Code Annotated §§ 31-7-70 et seq.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the related regulations and published interpretations.

"Immigration Act" means the Immigration Reform and Control Act of 1986.

"Indemnifiable Losses" means all losses, liabilities, damages, costs (including court costs and costs of appeal) and expenses (including reasonable attorneys' fees) actually incurred by an Indemnified Party.

6

"Indemnified Party" has the meaning set forth in Section 11.4(a).

"Indemnifying Party" has the meaning set forth in Section 11.4(a).

"Insurable Leaseholds" has the meaning set forth in Section 7.5.

"Intellectual Property" means, to the extent held or used in or ancillary to the business or operation of the Facilities, patents, trademarks, trade names (including Seller's interest in the names "Palmyra Medical Center" and "Palmyra Park Hospital" insofar as any of such names is used within 50 miles of the Hospital), service marks, copyrights and any applications therefor, mask works, net lists, schematics, technology, know-how, trade secrets, ideas, algorithms, processes, computer software programs and applications (in both source code and object code form), and tangible or intangible proprietary information or material, except as set forth on Schedule 2.2.

"Interim Balance Sheet" has the meaning set forth in Section 2.6(c).

"Inventory" means all inventory, supplies, food, pharmaceuticals, janitorial and office supplies and other disposables on hand or under order for use exclusively in the business or operation of Seller and its Affiliates at the Facilities, but excluding the Excluded Inventory.

"Inventory Amount" means the value of the Inventory as of the Closing Date, as determined (or agreed upon by Buyer and Seller) in accordance with <u>Section 2.6</u> or <u>Section 2.7</u>.

"Justice Department" means the United States Department of Justice.

"Law" means any constitutional provision, statute, ordinance or other law, rule, regulation, interpretation, judgment, decree or order of any Governmental Entity (other than Buyer or its Affiliates) or any settlement agreement or compliance agreement with any Governmental Entity (other than Buyer or its Affiliates).

"<u>Leased Real Property</u>" means real property owned by a third party which is subject to a leasehold or subleasehold estate in favor of Seller, described on <u>Exhibit D-2</u> (as supplemented or amended in accordance with <u>Section 6.1(a)</u>).

"Medicaid" has the meaning set forth in Section 4.9(a).

"Medicare" has the meaning set forth in Section 4.9(a).

"Obligations" has the meaning set forth in Section 12.21(a).

"OSHA" has the meaning set forth in the definition of Environmental Laws.

"Other Receivables" means Seller's promissory notes and accounts receivable made or generated in connection with the operation of the Facilities through at the Closing that are accounted for as "Other Receivables" in the Facilities' financial statements and included in the determination of the Other Receivables Amount. Exhibit E indicates the types of notes and accounts receivable that would be classified as Other Receivables (and the Other Receivables Amount) if the Closing took place as of November 30, 2010.

"Other Receivables Amount" means the value of the Other Receivables as of the Closing Date, as determined (or agreed upon by Buyer and Seller) in accordance with <u>Section 2.6</u> or <u>Section 2.7</u>.

"Owned Real Property" means the real property described on Exhibit D-1 (as supplemented or amended in accordance with Section 6.1(a)), together with all leases and subleases therein, improvements, buildings or fixtures located thereon or therein, all easements, rights of way, and other appurtenances

thereto (including appurtenant rights in and to public streets), and all claims and recorded or unrecorded interests therein, including any and all options to acquire such real property.

"PNI" has the meaning set forth in the Preamble.

"PPHS" has the meaning set forth in the Preamble.

"Palmyra Medical Center Plans" has the meaning set forth in Section 9.3(b)

"Partial Assignment and Assumption Agreement" means the Partial Assignment and Assumption Agreement attached hereto as Exhibit B.

"Payable Tax Items" has the meaning set forth in Section 4.18(b).

"Permit" means any license, permit or certificate of need required to be issued by any Governmental Entity (other than Buyer or its Affiliates).

"Permitted Encumbrances" means the following: (a)(i) liens for current ad valorem taxes not yet due and payable, (ii) all existing utility, access or other easements and rights of way of record provided that, unless otherwise deemed a Permitted Encumbrance, for example under clause (vii) or clause (viii) below, no encroachment of any such easement or right of way shall be considered a Permitted Encumbrance, (iii) Laws regulating the use or enjoyment of the Owned Real Property or Leased Real Property provided that, unless otherwise deemed a Permitted Encumbrance, for example under clause (vii) or clause (viii), no violation of any such Laws that would constitute an Encumbrance shall be considered a Permitted Encumbrance, (iv) all leases and subleases to third party tenants of space in or portions of the Owned Real Property or Leased Real Property, (v) liens securing obligations which are Assumed Liabilities, (vi) those matters set forth in those title commitments issued by Old Republic National Title Insurance Company to the Authority pursuant to Section 7.5 and on such surveys as are prepared in accordance with Section 7.5 that are either accepted or waived by Buyer pursuant to Section 7.5 (or insured over by the title insurance company or indemnified against by Seller as contemplated by Section 7.5) or that would not have the result of Buyer being unable to operate the Hospital and any other material Facilities in a manner consistent with the current use thereof by Seller or its Affiliates, (vii) any other Encumbrances which, either individually or in the aggregate, do not interfere with the use of the Purchased Assets in a manner consistent with the current use thereof by Seller or its Affiliates, and (viii) with respect to Leased Real Property, Encumbrances which encumber the fee interest in such property or, in the case of subleased Leased Real Property, any leasehold estate superior to that of the Seller; and (b) such other matters set forth in the title commitments and Survey contemplated by Section 7.5 that are either accepted or waived by Buyer in accordance with Section 7.5 or insured over by the title insurance company or indemnified against by Seller as contemplated by Section 7.5.

"Person" means an association, a corporation, a limited liability company, an individual, a partnership, a limited liability partnership, a trust, a hospital authority organized and existing under the Hospital Authorities Act, or any other entity or organization, including a Governmental Entity.

8

"Petitioner" has the meaning set forth in Section 12.3(b).

"PIP" has the meaning set forth in Section 9.1(a).

"Plans" has the meaning set forth in Section 4.15(a).

4016376 6

- "Prepaid Expenses" means Seller's prepaid expenses, claims for refunds and rights to offset in respect thereof relating to the Facilities, but excluding the Excluded Prepaid Expenses. Exhibit F indicates the types of such assets that would be classified as Prepaid Expenses and Excluded Prepaid Expenses (and the Prepaid Expense Amount) if the Closing took place as of November 30, 2010.
- "Prepaid Expense Amount" means the value of the Prepaid Expenses as of the Closing Date, as determined (or agreed upon by Buyer and Seller) in accordance with Section 2.6 or Section 2.7.
 - "Price Adjustment Amount" has the meaning set forth in Section 2.5.
 - "Price Adjustment Statement" has the meaning set forth in Section 2.6(c).
 - "Purchase Price" has the meaning set forth in Section 2.5.
 - "Purchased Assets" has the meaning set forth in Section 2.1.
 - "RCRA" has the meaning set forth in the definition of Environmental Laws.
 - "Real Property" means the Owned Real Property and the Leased Real Property.
 - "Reference Balance Sheet" has the meaning set forth in Section 4.6(a)(ii).
 - "Rejected Contract" has the meaning set forth in Section 6.1(c).
 - "Rescission Fee" has the meaning set forth in Section 10.1(b).
 - "Respondent" has the meaning set forth in Section 12.3(b).
 - "Retirement Plans" has the meaning set forth in Section 4.15(g).
 - "Returns" has the meaning set forth in Section 4.18(a).
 - "Second Arbitrator" has the meaning set forth in Section 12.3(b).
 - "Seller Cost Reports" has the meaning set forth in Section 9.2.
 - "Seller Indemnified Parties" has the meaning set forth in Section 11.2.
 - "Seller" has the meaning set forth in the Preamble.
- "Senior Management" means the persons holding the following titles with respect to the Hospital: Chief Executive Officer, Chief Financial Officer, Chief Operating Officer and Chief Nursing Officer.
 - "Survey" has the meaning set forth in Section 7.5.
 - "Survival Period" has the meaning set forth in Section 11.5.
 - "TJC" has the meaning set forth in Section 4.9(c).
 - "Territory" has the meaning set forth in Section 10.15(e).
 - "Third Arbitrator" has the meaning set forth in Section 12.3(b).

"Transition Patient Receivables" has the meaning set forth in Section 9.1(b).

"Transition Patients" means patients admitted to the Hospital prior to the Closing but not yet discharged as of the Closing.

"Transition Services" means the rendering of services and provision of medicine, drugs and supplies by the Hospital to Transition Patients whose care is reimbursed by Medicare, Medicaid, TRICARE or other third party programs on a prospective payment system, including the Medicare diagnostic related groups.

"TRICARE" means the Department of Defense's managed healthcare program for active duty military, active duty service families, retirees and their families and other beneficiaries.

Interpretation. In this Agreement, unless the context otherwise requires: (a) references 1.2 to this Agreement are references to this Agreement and to the Exhibits and Schedules to this Agreement; (b) references to Sections are references to sections of this Agreement; (c) references to schedules and/or exhibits are references to the Schedules and/or Exhibits to this Agreement; (d) references to any party to this Agreement shall include references to its respective successors and permitted assigns; (e) references to a judgment shall include references to any order, writ, injunction, decree, determination or award of any court or tribunal; (f) the terms "hereof," "herein," "hereby," and derivative or similar words will refer to this entire Agreement; (g) references to any document (including this Agreement and the Exhibits and Schedules to this Agreement) are references to that document as amended, consolidated, supplemented, novated or replaced by the parties from time to time; (h) unless the context requires otherwise, references to any Law are references to that Law as of the Closing Date, and shall also refer to all rules and regulations promulgated thereunder; (i) the word "including" means including without limitation; (j) references to time are references to Eastern Standard or Daylight time (as in effect on the applicable day) unless otherwise specified herein, (k) the gender of all words herein include the masculine, feminine and neuter, and the number of all words herein include the singular and plural; (1) provisions of this Agreement shall be interpreted in such a manner so as not to inequitably benefit or burden any party through "double counting" of assets or liabilities or failing to recognize benefits that may result from any matters that impose losses or burdens on any party, including in connection with the determination the Price Adjustment Amounts and the calculation of losses on casualty claims, and (m) the terms "date hereof", "date of this Agreement" or similar terms shall mean the date of this Agreement as indicated in the Preamble to this Agreement.

2. SALE OF ASSETS AND CERTAIN RELATED MATTERS.

- 2.1 <u>Sale of Purchased Assets</u>. Subject to the terms and conditions of this Agreement, Seller agrees to sell, convey, transfer and deliver to Buyer and Buyer agrees to purchase at Closing, all assets of every description, and whether real, personal or mixed, tangible or intangible, other than the Excluded Assets, owned or leased by Seller on the Closing Date and held or used exclusively for the business or operation of the Facilities, including the following items (collectively, the "<u>Purchased Assets</u>"):
- (a) fee simple title to the Owned Real Property, and, to the extent permitted by Law, any rights of Seller or any of its Affiliates against third parties under general or limited warranty deeds related to any such Owned Real Property;
 - (b) leasehold title to the Leased Real Property;
 - (c) all Furniture and Equipment;

- (d) all Inventory;
- (e) all Prepaid Expenses;
- (f) to the extent assignable under applicable Law, all good will and current financial, patient, medical staff records, employee records as of the Closing and any other records held or used by Seller exclusively for the business or operation of the Facilities;
 - (g) all of the interest of Seller in all Assumed Contracts;
- (h) all Permits and Approvals issued or granted by Governmental Entities to the extent assignable and which are held or used by Seller and relate exclusively to the ownership, development and business or operation of the Purchased Assets (including any pending Permits and Approvals that relate exclusively to any Purchased Assets);
 - (i) all Intellectual Property;
- (j) all computers and other data processing equipment held or used exclusively in the business or operation of the Facilities;
 - (k) any insurance proceeds to the extent provided in Section 2.6(e);
 - (l) the Other Receivables;
- (m) to the extent assignable by Seller, the telephone numbers used by Seller at the Facilities as of the Closing Date;
- (n) all rights to and all the interest of Seller in any causes of action, claims, demands, damages, debts, dues or any and all other rights set forth in or arising out of or related to a pending legal action entitled Palmyra Park Hospital, Inc., d/b/a Palmyra Medical Center v. Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Health System, Inc., Civil Action File No. 1:08-CV-00102-WLS, pending in the United States District Court for the Middle District of Georgia, Albany Division; and
- (o) to the extent not included in any of the foregoing, (A) any current assets included in the determination of the Price Adjustment Amounts or any non-current assets reflected on the Reference Balance Sheet, except for assets used, consumed, or disposed of in the ordinary course of business since the Balance Sheet Date, (B) any assets purchased or otherwise acquired since the Balance Sheet Date which are not reflected on the Reference Balance Sheet but which are held or used exclusively in the business or operation of the Facilities and (C) the Facilities. Seller shall be permitted, at its expense, to retain copies of all records that it reasonably determines may be required by Seller or its Affiliates in the future. Subject to the conditions set forth in this Agreement, at Closing, Seller will sell, convey, transfer, assign and deliver good and marketable title to the Purchased Assets free and clear of all Encumbrances other than Permitted Encumbrances. Buyer shall not be obligated to assume, and the term "Assumed Contracts" shall be deemed to exclude, any Contracts relating to the business or operation of the Facilities to the extent that any such Contract is an Excluded Contract.
- 2.2 <u>Excluded Assets</u>. Notwithstanding anything herein to the contrary, the following assets are not intended by the parties to be a part of the sale and purchase contemplated hereunder and are excluded from the Purchased Assets (the "Excluded Assets"):

- (a) restricted and unrestricted cash and cash equivalents, including investments in marketable securities, certificates of deposit, bank accounts and promissory notes, except to the extent the value of such assets are reflected in the Price Adjustment Amounts;
 - (b) all Excluded Receivables;
 - (c) all Excluded Prepaid Expenses;
 - (d) all Excluded Inventory;
- (e) all names and symbols used in connection with the Facilities and the Purchased Assets which include the name "HCA" or any variants thereof, or any other names which are proprietary to HCA or its Affiliates, unless exclusively used in connection with the business and operation of the Facilities:
- (f) all computer software and programs which are proprietary to HCA or its Affiliates;
- (g) all documents, records, operating manuals film and video pertaining to the Hospital proprietary to HCA or its Affiliates or which Seller is required by law to retain;
- (h) to the extent permitted by <u>Sections 6.2</u> and <u>6.3</u>, all assets disposed of or exhausted prior to Closing in the ordinary course of business, including Inventory, Prepaid Expenses, and Furniture and Equipment, except to the extent the value of such assets are included in the Price Adjustment Amounts;
- (i) all amounts included in the entry (or line item) on the Closing Balance Sheet entitled "Intercompany Debt";
- (j) all insurance proceeds arising in connection with the operation of the Facilities or the Purchased Assets prior to Closing, except to the extent provided in <u>Section 2.6(e)</u>;
- (k) any claims or rights related to the Purchased Assets (including the Assumed Contracts), contractual or otherwise, accruing or arising prior to the Closing, except to the extent (A) included in the determination of the Price Adjustment Amounts, (B) such claim or right would also relate to a period after Closing, but only to the extent such right or claim relates to periods after Closing or (C) such claim or right is described at Section 2.1(n) of this Agreement;
 - (l) all corporate or partnership records and stock books of Seller and its Affiliates;
- (m) all assets used by HCA and its Affiliates in rendering corporate services to the Seller or the Facilities that are located outside the Facilities, except to the extent such assets are reflected on the Reference Balance Sheet or are included in the Purchase Price;
 - (n) all Excluded Contracts;
- (o) any assets of HCA or its Affiliates not used exclusively in connection with the Facilities, including (i) assets used on a company-wide or region-wide basis, and (ii) assets of HCA and its Affiliates used at other hospitals and health care facilities owned or operated by HCA or its Affiliates, unless such assets are reflected on the Reference Balance Sheet or included in the Price Adjustment Amounts;

- (p) rights to refunds of taxes and all other tax assets for periods prior to the Closing; and
 - (q) assets listed on <u>Schedule 2.2</u>.
- 2.3 <u>Assumed Liabilities</u>. As of Closing, Buyer agrees to assume the future payment and performance of the following liabilities of Seller and its Affiliates (collectively, the "<u>Assumed</u> Liabilities"):
- (a) other than obligations and liabilities accruing, arising out of, or relating to acts or omissions of Seller and its Affiliates, all obligations and liabilities accruing, arising out of, or relating to the period after the Closing in connection with (i) the Assumed Contracts, (ii) the Facilities and (iii) the Purchased Assets;
- (b) obligations and liabilities as of the Closing in respect of (i) Accrued PTO (but only to the extent the liability for such Accrued PTO is reflected in the Accrued PTO Amount), (ii) accrued employee benefits for sick leave, extended illness and similar benefits (the "Accrued EIB") of employees of Seller and its Affiliates at each Facility as of the Closing Date, but excluding the accrued employee benefits for sick leave, extended illness and similar benefits of employees of Seller and its Affiliates who are not offered employment or who do not accept offers of employment by PNI as of Closing as contemplated by Section 9.3, (iii) obligations and liabilities assumed by Buyer under Section 9.3, and (iv) any payroll taxes related to the Assumed Liabilities described in clauses (i) through (iii) above;
- (c) any state and local transfer, sales and recording fees, documentary stamps and taxes which may arise upon the consummation of the transactions contemplated herein;
- (d) all obligations and liabilities accruing, arising out of or relating to acts or omissions of Buyer, PPHS, PNI or their respective Affiliates, before or after the Closing;
 - (e) the Permitted Encumbrances;
 - (f) the ADA Liabilities; and
- (g) the Assumed Indebtedness, but only to the extent such Assumed Indebtedness is reflected in the Assumed Indebtedness Amount.
- **Excluded Liabilities.** Except as expressly provided to the contrary in Section 2.3, Buyer is not obligated to pay or assume, and none of the Purchased Assets shall be or become liable for or subject to, any liability of Seller and its Affiliates, including the following, whether fixed or contingent, recorded or unrecorded, known or unknown and whether or not set forth on Schedules hereto (collectively, the "Excluded Liabilities"):
- (a) any obligation or liability accruing, arising out of, or relating to acts or omissions prior to Closing, including any acts or omissions in connection with (i) any Assumed Contract, (ii) the business or operation of the Facilities, or (iii) the Purchased Assets;
- (b) any obligation or liability accruing, arising out of, or relating to any act or omission by Seller or its Affiliates after Closing;

- (c) any obligation or liability accruing, arising out of, or relating to any Excluded Contract;
- (d) any liability or obligation for severance with respect to those members of Senior Management who are not employed by Buyer on the Closing Date;
- (e) any obligation or liability accruing, arising out of, or relating to any federal, state or local investigations of, or claims or actions against, Seller or any of its Affiliates or any of their employees, medical staff, agents, vendors or representatives with respect to acts or omissions prior to the Closing;
- (f) any civil or criminal obligation or liability accruing, arising out of, or relating to any acts or omissions of Seller, its Affiliates or their directors, officers, employees and agents claimed to violate any Laws;
- (g) any liabilities or obligations existing on the Closing Date which are required to be reflected on a balance sheet prepared in accordance with GAAP and which were not reflected on the Reference Balance Sheet, except for Assumed Liabilities;
- (h) indebtedness and other obligations or guarantees of Seller, other than with respect to Assumed Liabilities;
- (i) liabilities or obligations of Seller of every kind and nature, known and unknown, in respect of periods prior to and up to Closing arising under the terms of the Medicare, Medicaid, TRICARE or any other third party payor programs or health insurers, and any liability arising pursuant to the Medicare, Medicaid, TRICARE or any other third party payor program as a result of the consummation of the transactions contemplated herein, including, without limitation, recapture of excess depreciation arising from or relating to services rendered or failed to be rendered at or by, or reimbursement claims relating to such services submitted by or on behalf of Seller prior to the Closing Date, including claims for overpayments or other excessive reimbursement or non-covered services; and any liability of Seller under or arising prior to the Closing Date from any risk pools and other risk sharing agreements established in connection with any managed care and payor contract assumed by the Buyer hereunder;
- (j) federal, state or local tax liabilities or obligations of or assessments against Seller in respect of periods prior to Closing or resulting from the consummation of the transactions contemplated herein, including, without limitation, any income tax, any franchise tax, any tax recapture, real estate, ad valorem tax, any indigent care tax, and any FICA, FUTA, workers' compensation taxes and any and all other taxes or amounts due and payable by Seller;
- (k) liability for any and all claims by or on behalf of Seller's employees relating to periods prior to Closing, including, without limitation, liability for any compensation, benefits, pension, profit sharing, deferred compensation, or any other employee health and welfare benefit plans, liability for any EEOC claim, wage and hour claim, unemployment compensation claim or workers' compensation claim or personnel policy, including those relating to any termination of employment, and all employee wages and benefits, including, without limitation, accrued vacation, sick leave, extended illness benefits, holiday pay, severance pay and related taxes or other liability related thereto;
- (l) any obligation or liability attributable to periods prior to or as of Closing and asserted under the federal Hill-Burton program or other restricted grant and loan programs with respect to the ownership or operation of the Assets;

- (m) contracts and agreements between Seller and one or more of Seller's Affiliates with respect to insurance, management services, information services or systems, or cash management services;
- (n) any debt, obligation, expense or liability of Seller arising out of or incurred solely as a result of any transaction of Seller occurring after Closing;
- (o) liability arising out of the assignment at Closing of any contract due to breach by Seller prior to Closing;
- (p) liabilities for expenses incurred by Seller incidental to the preparation of this Agreement, preparation or delivery of materials or information requested by Buyer, or the consummation of the transactions contemplated hereby, including all broker, counsel and accounting fees or any account payable which is attributable to legal and accounting fees and similar costs incurred by Seller which are directly related to the sale of any of the Purchased Assets of Seller;
- (q) liabilities for the unpaid taxes of any Person under U.S. Treasury Regulation §1.1502-6 (or any similar provision of state, local or foreign Law), as a transferee or successor, by contract or otherwise;
- (r) liabilities arising from or in connection with (i) any administrative ruling or other order, stipulation or decree of any federal, state or local agency or (ii) the violation of any federal, state or local act, statute, rule or regulation, decree or ordinance, Medicare or Medicaid program integrity or compliance agreement either involving Seller, or relating to or arising in connection with the use, operation, ownership or possession of the Facilities or Hospital or the use, operation, ownership or possession of any of the other Purchased Assets prior to the Closing Date;
- (s) liabilities attributable to any assets, properties or contracts which are not among the Purchased Assets; and
- (t) any other liability, fixed or contingent, relating to the Facilities or Purchased Assets occurring prior to the Closing and not expressly assumed by Buyer pursuant to Section 2.3.
- 2.5 <u>Purchase Price</u>. The consideration to be paid by Buyer (the "<u>Purchase Price</u>") shall be an amount equal to (a) \$195,000,000.00 (the "<u>Base Purchase Price</u>"); plus (b) the Inventory Amount, the Other Receivables Amount and the Prepaid Expenses Amount, plus (c) the Capital Expenditure Amount, minus (d) the Assumed Indebtedness Amount. The Inventory Amount, the Other Receivables Amount, the Prepaid Expenses Amount, the Capital Expenditure Amount and the Assumed Indebtedness Amount are referred to collectively as the "<u>Price Adjustment Amounts</u>," and each of them is referred to individually as a "<u>Price Adjustment Amount</u>".

2.6 <u>Determination of the Price Adjustment Amounts</u>.

- (a) In the absence of an agreement by Buyer and Seller to the contrary (including an agreement as contemplated by Sections 2.6(b) (d)):
 - (i) the Inventory Amount, the Other Receivables Amount and the Prepaid Expenses Amount will be the value of the Inventory, the Other Receivables and the Prepaid Expenses as of the Closing Date as determined in accordance with GAAP, applied on a basis consistent with the Reference Balance Sheet;

- (ii) the Assumed Indebtedness Amount (including the Accrued PTO Amount) will be the amount of the liability for the Assumed Indebtedness (including the Accrued PTO) as of the Closing Date as determined in accordance with GAAP, applied on a basis consistent with the Reference Balance Sheet; and
- (iii) each of the Price Adjustment Amounts described in clauses (i) and (ii), above, will be the amounts of the relevant assets and liabilities as reflected on the Closing Balance Sheet, except to the extent that the Closing Balance Sheet includes the value of Excluded Inventory, Excluded Prepaid Expenses or Excluded Receivables or the amount of a liability that is not Assumed Indebtedness or Accrued PTO (e.g. the amount of Seller's liability for "paid time off" with respect to employees of Seller to whom PNI does not offer employment in accordance with Section 9.3 or who do not accept PNI's offer of employment) in which case the value of Excluded Inventory, Excluded Prepaid Expenses or Excluded Receivables or the amount of a liability that is not Assumed Indebtedness or Accrued PTO will be excluded from the applicable Price Adjustment Amounts.
- (b) Notwithstanding the provisions of <u>Section 2.6(a)</u>, Buyer and Seller acknowledge that it is their mutual desire and intent to determine the final Purchase Price at or prior to the Closing, even though the Closing Balance Sheet and all relevant data regarding the Capital Expenditure Amount will not be available until after the Closing. Consequently, Buyer and Seller agree to use commercially reasonable efforts to determine and agree upon the Price Adjustment Amounts at or prior to the Closing in accordance with <u>Sections 2.6(c)</u> and <u>(d)</u>.
- Not less than three (3) days prior to the Closing Date, Seller shall deliver to Buyer (i) the combined balance sheet of the Facilities as of the last day of the month immediately preceding the Closing Date, which balance sheet will be prepared in accordance with GAAP, applied on a basis consistent with the Reference Balance Sheet (the "Interim Balance Sheet"), (ii) the portions of the trial balance for the Interim Balance Sheet relating to the Price Adjustment Amounts, and (iii) a statement setting forth (A) Seller's proposed Purchase Price, (B) each of Seller's proposed Price Adjustment Amounts, (C) calculations showing the adjustments that Seller made to the applicable amounts shown on the Interim Balance Sheet to derive Seller's proposed Price Adjustment Amounts, and (D) a schedule of capital expenditures supporting the Capital Expenditure Amount (the "Price Adjustment Statement"). The Price Adjustment Statement will include (to the extent applicable) (1) Seller's designation of the prepaid expenses, claims for refunds and rights to offset that are Prepaid Expenses and Excluded Prepaid Expenses, respectively, (2) Seller's description of inventory, supplies, food, pharmaceuticals, janitorial and office supplies and other disposals that are Excluded Inventory; (3) Seller's list of employees whose "paid time off" liability is being excluded from the calculation of the Accrued PTO Amount, (4) Seller's adjustment to the relevant amounts as shown on the Interim Balance Sheet to reflect the exclusion of any such assets or liabilities from Seller's calculation of the Price Adjustment Amounts; (5) Seller's adjustment to the relevant amounts as shown on the Interim Balance Sheet to reflect the estimated change in the value of the relevant Purchased Assets or the amount of the Assumed Indebtedness between the date of the Interim Balance Sheet and the Closing Date; and (6) such other information regarding Seller's proposed Price Adjustment Amounts as Seller deems necessary or appropriate.
- (d) Seller shall (i) consult with Buyer and its representatives with respect to the Interim Balance Sheet and the Price Adjustment Statement, and (ii) permit Buyer and its representatives to review Seller's books and records relating thereto. Buyer and Seller will use commercially reasonable efforts to resolve any objections or proposed adjustments raised or proposed by Buyer with respect to the Interim Balance Sheet or the Price Adjustment Statement and will modify the Interim Balance Sheet and/or the Price Adjustment Statement as necessary to reflect the resolution of any such objections or proposed adjustments. As and to the extent Buyer and Seller agree in writing upon any of the Price

Adjustment Amounts at or prior to the Closing, (A) such agreed upon amount will be used to determine the amount of cash to be paid by Buyer at Closing under Section 3.3(a) with respect to such Price Adjustment Amount, and (B) the parties will not be required to determine or adjust such Price Adjustment Amount post-Closing as contemplated by Section 2.7. As and to the extent Buyer and Seller are unable to agree in writing upon any Price Adjustment Amount at or prior to the Closing, then, (1) for purposes of determining the amount of cash to be paid by Buyer at Closing under Section 3.3(a) with respect to such Price Adjustment Amount, the parties will use the amount set forth in Seller's Price Adjustment Statement, and (2) such unresolved Price Adjustment Amount will be resolved following the Closing in accordance with Section 2.7.

- If any material part of the Purchased Assets is damaged, lost or destroyed (whether by fire, theft, or other casualty event) prior to the Closing, Buyer may, at its option, either (i) reduce the Purchase Price by the (A) fair market value of the Purchased Assets damaged, lost or destroyed, such value to be determined as of the date immediately prior to such damage, loss or destruction or (B) by the estimated cost to replace or restore the damaged, lost or destroyed Purchased Assets, or (ii) require Seller upon the Closing to transfer to Buyer the proceeds (or the right to the proceeds) of any applicable insurance, including any business interruption insurance relating to periods after the Closing. Any determination of the fair market value of any Purchased Assets which are damaged, lost or destroyed shall be made in good faith jointly by Buyer and Seller prior to Closing. In the event that Buyer and Seller cannot reach a resolution regarding the fair market value of any such damaged, lost or destroyed Purchased Assets, the Closing shall occur and (x) the Purchase Price shall be subject to subsequent adjustment to reflect such damage, loss or destruction after the Closing Date (the amount of such adjustment, the "Additional Adjustment Amount") in accordance with the dispute resolution procedures in Section 12.3(c) and (y) Seller shall be obligated to pay to Buyer interest on the Additional Adjustment Amount from the Closing Date to the date of payment of the Additional Adjustment Amount at the Applicable Rate.
- (f) The amount of any liability under this Agreement that is owing from Buyer to Seller or Seller to Buyer and that has been agreed, resolved or liquidated prior to the Closing shall be treated appropriately as a further adjustment to the Purchase Price to be paid at Closing. The payment of any such amount that has not been so agreed, resolved or liquidated shall be made when agreed, resolved or liquidated; provided that the foregoing shall not affect Buyer's obligation to pay the entire amount of the Purchase Price at Closing.
- 2.7 <u>Post-Closing Determinations of Price Adjustment Amounts</u>. If Buyer and Seller have not agreed in writing upon any of the Price Adjustment Amounts at or before the Closing, the amount of such Price Adjustment Amount (and any adjustments to the amount of the Purchase Price paid pursuant to Section 3.3(a) at Closing) will be determined in accordance with this <u>Section 2.7</u>.
- (a) Following the Closing, Buyer and Seller will continue to use commercially reasonable efforts to determine and agree upon any unresolved Price Adjustment Amount in the manner described in Sections 2.6(c) and (d).
- (b) If Buyer and Seller have not agreed in writing upon any such Price Adjustment Amount within forty-five (45) days following the Closing, Seller shall deliver to Buyer (i) the combined balance sheet of the Facilities as of the Closing Date (the "Closing Balance Sheet"), which Closing Balance Sheet will be prepared in accordance with GAAP, applied on a basis consistent with the Reference Balance Sheet, except that as it relates to any unresolved Price Adjustment Amount, the Closing Balance Sheet will exclude the value (or amount) of (A) any Excluded Inventory, (B) any Excluded Prepaid Expenses, (C) any accrued liability for "paid time off" that is not within the definition of "Accrued PTO," and (D) any liability for indebtedness for borrowed money or capitalized lease

4016376.6

obligations that are not within the definition of "Assumed Indebtedness," (ii) the portions of the trial balance for the Closing Balance Sheet relating to the unresolved Price Adjustment Amounts, and (iii) a statement setting forth (A) Seller's proposed Purchase Price, (B) each of Seller's proposed Price Adjustment Amounts (for the unresolved Price Adjustment Amounts), (C) calculations showing the adjustments, if any, that Seller made to the applicable amounts shown on the Closing Balance Sheet to derive such proposed Price Adjustment Amounts and (D) a schedule of capital expenditures supporting the Capital Expenditure Amount. Such statement will include (to the extent applicable) (1) Seller's designation of the prepaid expenses, claims for refunds and rights to offset that are Prepaid Expenses and Excluded Prepaid Expenses, respectively; (2) Seller's description of inventory, supplies, food, pharmaceuticals, janitorial and office supplies and other disposals that are Excluded Inventory; (3) Seller's list of employees whose "paid time off" liability is being excluded from the calculation of the Accrued PTO Amount; (4) the list of any indebtedness for borrowed money or capitalized lease obligations that are "Assumed Indebtedness," and (5) such other information regarding Seller's proposed Price Adjustment Amounts as Seller deems necessary or appropriate.

- Within thirty (30) days after Seller's delivery of the Closing Balance Sheet and other information contemplated by Section 2.7(b), Buyer shall, in a written notice to Seller, either accept or describe in reasonable detail any proposed adjustments to the Closing Balance Sheet and Seller's proposed Price Adjustment Amounts and the reasons therefor, and shall include pertinent calculations. At all reasonable times following delivery by Seller to Buyer of the proposed Closing Balance Sheet and other information contemplated by Section 2.7(b), Seller shall make available to Buyer and its agents all books and records of Seller included in Excluded Assets that relate to the determination of the unresolved Price Adjustment Amounts. If Buyer fails to deliver notice of acceptance or objection to the Closing Balance Sheet and the Price Adjustment Amounts within such thirty (30) day period, Buyer shall be deemed to have accepted the Closing Balance Sheet and Seller's proposed Price Adjustment Amounts. In the event that Seller and Buyer are not able to agree on the Price Adjustment Amounts within thirty (30) days from and after the receipt by Seller of any objections raised by Buyer, Seller and Buyer shall each have the right to require that such disputed determination be submitted to such independent certified public accounting firm as Seller and Buyer may then mutually agree upon in writing for computation or verification in accordance with the provisions of this Agreement. Any such independent certified public accounting firm shall not have been engaged by either party in the preceding three years. The results of such accounting firm's report shall be binding upon Seller and Buyer, and such accounting firm's fees and expenses for each such disputed determination shall be borne equally by the parties. Appropriate payment of the difference between the amount of the Purchase Price paid pursuant to Section 3.3(a) at Closing and the amount of the Purchase Price as finally determined pursuant to this Section 2.7 shall be made by Buyer or Seller, as applicable, in immediately available funds promptly upon agreement of Buyer and Seller on the amount of the unresolved Price Adjustment Amounts (or upon the determination of the amount of such Price Adjustment Amounts in accordance with this Section 2.7). Any amounts due under this Section 2.7 shall bear interest from the Closing Date until paid at a rate equal to the Applicable Rate.
- 2.8 Proration. Within ninety (90) days after the Closing Date, Seller and Buyer shall prorate as of the Closing Date any amounts which become due and payable after the Closing Date with respect to (i) the Assumed Contracts, (ii) ad valorem taxes, if any, on the Purchased Assets, (iii) property taxes on the Purchased Assets and (iv) all utilities servicing any of the Purchased Assets, including water, sewer, telephone, electricity and gas service, in each case to the extent not included in Assumed Liabilities. Any such amounts which are not available within ninety (90) days after the Closing Date shall be similarly prorated as soon as practicable thereafter.

3. CLOSING

- 3.1 Closing. Subject to the satisfaction or waiver by the appropriate party of all the conditions precedent to Closing specified in Sections 7 and 8 hereof, the consummation of the sale and purchase of the Purchased Assets and the other transactions contemplated by and described in this Agreement (the "Closing") shall take place at the offices of the Authority at 10:00 a.m. not later than the fifth business day after the conditions set forth in Sections 7 and 8 have been satisfied or waived or at such other date and/or at such other location as the parties hereto may mutually designate in writing (the "Closing Date"). The parties shall use commercially reasonable efforts to cause the conditions set forth in Sections 7 and 8 to be satisfied so that the Closing will occur on January 31, 2011.
- 3.2 <u>Actions of Seller at Closing</u>. At the Closing and unless otherwise waived in writing by Buyer, Seller shall deliver to Buyer the following:
- (a) Subject only to the Permitted Encumbrances, (i) limited warranty deeds containing special warranty of title, duly executed by Seller, or by its applicable Affiliate, in recordable form, conveying to Buyer, or to such Person as designated by Buyer, good and marketable fee simple title to the Owned Real Property and/or (ii) Assignment and Assumption Agreements, fully executed by Seller, or its applicable Affiliate, in recordable form assigning to Buyer leasehold title to any Leased Real Property;
- (b) a certificate executed by a duly authorized officer of Seller certifying that the conditions set forth in Sections 7.1 and 7.6 have been satisfied;
- (c) (i) an affidavit to the effect that, other than as specified therein, no improvements or repairs have been made by, or for the account of, or at the instance of, Seller to or on the Property within ninety-five (95) days preceding the Closing Date for which payment in full has not been made; (ii) a FIRPTA certificate; (iii) a residency affidavit (if required by the title company or applicable Law); and (iv) the information to enable the closing agent to file a 1099 to report the sale;
- (d) The General Bill of Sale and Assignment, duly executed by Seller, conveying to Buyer or to such Affiliate of Buyer which is designated by Buyer valid title to all tangible assets which are a part of the Purchased Assets and valid title to all intangible assets which are a part of the Purchased Assets, free and clear of all Encumbrances other than the Assumed Liabilities and Permitted Encumbrances;
- (e) The Assignment and Assumption Agreement (and, if applicable, the Partial Assignment and Assumption Agreement), duly executed by Seller, conveying Seller's interest in the Assumed Contracts to Buyer or to such Affiliate of Buyer which is designated by Buyer;
- (f) A certificate of Seller certifying that the conditions set forth in Section 7.1 have been satisfied;
- (g) A certificate of incumbency for the officers of Seller executing this Agreement and any other agreements or instruments contemplated herein dated as of the Closing Date;
- (h) A certificate of existence and good standing of Seller from the State of Georgia, dated the most recent practical date prior to Closing;
- (i) The opinion of Seller's counsel as described in and provided by <u>Section 7.2</u> hereof,

- (j) Possession of the Purchased Assets, subject only to Permitted Encumbrances, to Buyer or such Affiliate of Buyer which is designated by Buyer; and
- (k) Copy of the resolution duly adopted by the board of directors of Seller authorizing and approving Seller's performance of the transactions contemplated hereby and the execution and delivery of this Agreement and the documents described herein, certified as true and of full force and effect as of Closing, by the appropriate officers of Seller;
- (l) Seller shall deliver an executed Plaintiff's Motion for Order of Dismissal Without Prejudice of Palmyra Park Hospital, Inc., d/b/a Palmyra Medical Center v. Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Health System, Inc., Civil Action File No. 1:08-CV-00102-WLS, pending in the United States District Court for the Middle District of Georgia, Albany Division in the form attached hereto as Exhibit G;
- (m) Seller shall deliver an executed Appellant Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center's Motion for Permission to Withdraw Appeal of the matter pending before the Court of Appeals for the State of Georgia appealing the decision rendered by the Superior Court of Sumter County, Georgia, in Phoebe Sumter Medical Center v. Ga. Department of Community Health and Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center, Case Number 10CV-30-P and an executed Appellant Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center's Motion For Permission to Withdraw Appeal of the matter pending before the Court of Appeals for the State of Georgia appealing the decision rendered by the Superior Court of Dougherty County, Georgia, in Phoebe Putney Memorial Hospital v. Ga. Department of Community Health and Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center, Case Number 10CV-097-1 in the forms attached hereto as Exhibits H and I, respectively;
- (n) Seller shall deliver an executed copy of the document in the form attached hereto as Exhibit J pursuant to which Seller will withdraw all pending applications for Certificates of Need in Dougherty County, Georgia including, but not limited to, Certificate of Need application # 2008-081; and
- (o) Such other instruments and documents as are reasonably necessary to satisfy the conditions precedent to Buyer's obligations hereunder and the consummation of the transactions described in this Agreement.
- 3.3 <u>Actions of Buyer at Closing</u>. At the Closing and unless otherwise waived in writing by Seller, Buyer shall deliver to Seller the following:
- (a) an amount equal to the Purchase Price by wire transfer in immediately available federal funds to an account of Seller's designation;
- (b) the Assignment and Assumption Agreements (and, if applicable, the Partial Assignment and Assumption Agreement), described in Sections 3.2(a)(iii) and 3.2(c), duly executed by Buyer, pursuant to which Buyer shall assume the future payment and performance of (i) the leases and other Assumed Contracts assigned thereby, and (ii) the Assumed Liabilities as herein provided;
 - (c) a copy of the General Bill of Sale and Assignment, duly executed by Buyer,
- (d) a certificate of Buyer certifying that the conditions set forth in <u>Section 8.1</u> have been satisfied:

- (e) certificates of incumbency for the respective officers of Buyer, PPHS and PNI executing this Agreement and any other agreements or instruments contemplated herein dated as of the Closing Date;
- (f) certificates of existence and good standing of PPHS and PNI from the State of Georgia, dated the most recent practical date prior to Closing;
 - (g) the opinion of Buyer's counsel as described in and provided by Section 8.2;
- (h) copies of resolutions duly adopted by the members of the Authority and the boards of directors of PPHS and PNI, authorizing and approving each such entity's performance of the transactions contemplated hereby and the execution and delivery of this Agreement and the documents described herein, certified as true and in full force as of Closing by an appropriate officer of Buyer, PPHS and PNI;
- (i) a certificate executed by a duly authorized officer of Buyer certifying that the conditions set forth in <u>Sections 8.1 and 8.5</u> have been satisfied; and
- (j) such other instruments and documents as are reasonably necessary to satisfy the conditions precedent to Seller's obligations hereunder and the consummation of the transactions described in this Agreement.
- Additional Acts. From time to time after Closing, Seller and its Affiliates shall execute and deliver such other instruments of conveyance and transfer, and take such other actions as Buyer reasonably may request, to convey and transfer full right, title and interest to, vest in, and place Buyer in legal and actual possession of, any and all of the Purchased Assets. In the case of Assumed Contracts and rights which cannot be transferred effectively without the consent of third parties, Seller with Buyer's assistance shall use commercially reasonable efforts to obtain such consents prior to the Closing and, if not so obtained, then promptly thereafter. Seller shall also furnish Buyer with such information and documents in its possession or under its control, or which Seller can execute or cause to be executed, as will enable Buyer to prosecute any and all petitions, applications, claims and demands relating to or constituting a part of the Purchased Assets. To the extent reasonably necessary, and subject to such limitations as Seller shall reasonably establish, (including limitations consistent with the provisions of Sections 6.1(b)(i), (ii) and (iii)), Seller and its Affiliates shall provide to Buyer access to (i) Seller's risk management information concerning the Hospital or the Facilities, and (ii) those policies or procedures of Seller pertaining to the pre-Closing operation of the Hospital or the Facilities (including employment policies, procedures, manuals and training aids) which Buyer needs to operate either the Hospital or the Facilities.

4. REPRESENTATIONS AND WARRANTIES OF SELLER

As of the date hereof and as of the Closing Date, Seller represents and warrants to Buyer the following (except (a) to the extent any of the following speaks as of a specific date, in which case Seller makes such representations and warranties as of the date specified, and (b) to the extent provided in any Schedules that Seller delivers, supplements or amends in accordance with Section 6.1(a), each of which shall have the same force and effect under this Agreement as though delivered simultaneously with the parties' execution and delivery of this Agreement):

4.1 <u>Incorporation: Qualification and Capacity</u>. Seller is a corporation validly existing, duly organized and in good standing under the Laws of the State of Georgia. Seller has the requisite power and authority to enter into this Agreement, perform its obligations hereunder and to conduct its

businesses as now being conducted. The execution and delivery by Seller of the Agreement and documents described herein, the performance by Seller of its obligations under the Agreement and documents described herein and the consummation by Seller of the transactions contemplated by the Agreement and documents described herein have been duly and validly authorized and approved by all necessary corporate actions on the part of Seller, none of which actions have been modified or rescinded and all of which actions remain in full force and effect.

- 4.2 <u>Corporate Powers; Consents; Absence of Conflicts With Other Agreements, Etc.</u>
 The execution, delivery and performance of the Agreement and documents described herein by Seller and the consummation by Seller of the transactions contemplated by the Agreement and documents described herein, as applicable:
 - (a) are not in contravention or violation of the terms of the articles or certificate of incorporation or bylaws of Seller;
 - (b) except for Approvals and Permits required for Buyer's operation of the Purchased Assets after Closing, do not require any material Approval or Permit of, or filing or registration with, or other action by, any Governmental Entity (other than Buyer or its Affiliates) to be made or sought by Seller or any of its Affiliates; provided that Seller makes no representation regarding any requirements of the HSR Act or any other state or federal antitrust regulations in connection with transactions contemplated hereby; and
 - obtained, will not conflict in any material respect with, or result in any violation of or default under (with or without notice or lapse of time or both), or give rise to a right of termination, cancellation, acceleration or augmentation of any obligation or to loss of a material benefit under, or result in the creation of any material Encumbrance (other than Permitted Encumbrances) upon any of the Purchased Assets under (A) any Contract or (B) any Law applicable to any of the Purchased Assets; provided that no representation or warranty is given with respect to consents required to assign any of the Assumed Contracts; and provided further that Seller makes no representation regarding any requirements of the HSR Act or any other state or federal antitrust regulations in connection with transactions contemplated hereby.
- 4.3 No Outstanding Rights. Except as set forth on Schedule 4.3, there are no outstanding rights (including any right of first refusal), options, Contracts made on behalf of Seller giving any Person any current or future right to require Seller or any of its Affiliates or, following the Closing Date, Buyer, to sell or transfer to such Person or to any third party any material interest in any of the Purchased Assets.
- 4.4 Purchased Assets. The Purchased Assets and the Excluded Assets (a) constitute all assets which are held or used by the Seller or any of its Affiliates exclusively for use by and necessary for the conduct of the business and operation of the Facilities in the manner conducted as of the date of this Agreement other than assets related to other services normally provided by the HCA corporate, division and regional offices (including group purchasing, supply chain services, revenue cycle services, information technology services and other corporate or centrally provided services), and (b) are sufficient to permit Buyer to carry on such business as currently conducted. There are no hospitals, medical office buildings or other health care related facilities reflected on the Reference Balance Sheet other than the Facilities and all of the Facilities are reflected on the Reference Balance Sheet.
- 4.5 <u>Binding Agreement</u> This Agreement and all Contracts to which Seller or any of its Affiliates will become a party hereunder are and will constitute the valid and legally binding obligations of Seller and/or such Affiliates and are and will be enforceable against them in accordance with the

respective terms hereof or thereof, except as enforceability may be restricted, limited or delayed by applicable bankruptcy or other Laws affecting creditors' rights generally and except as enforceability may be subject to general principles of equity.

4.6 Financial Information.

- (a) <u>Schedule 4.6</u> contains the following financial statements and financial information (collectively, the "Historical Financial Information"):
- (i) unaudited combined balance sheet of the Facilities dated as of December 31, 2009;
- (ii) unaudited combined balance sheet of the Facilities dated as of the Balance Sheet Date (the "Reference Balance Sheet");
- (iii) unaudited combined income statement of the Facilities for the twelvemonth period ended on December 31, 2009; and
- (iv) unaudited combined income statement of the Facilities for the ten (10) month period ended on the Balance Sheet Date.

The financial statements included in the Historical Financial Information have been prepared in accordance with GAAP, applied on a consistent basis throughout the periods indicated, and Seller has not changed any accounting policy or methodology in determining the obsolescence of inventory or in calculating reserves (including reserves for uncollected accounts receivable) throughout all periods presented. Subject to the exceptions to generally accepted accounting principles set forth in Schedule 4.6, such financial statements have been prepared from and (in all material respects) in accordance with the books and records of Seller with respect to the Facilities. Except as provided in Schedule 4.6, the balance sheets contained in the Historical Financial Information present fairly in all material respects the financial condition of the Facilities as of the dates indicated thereon. The income statements contained in the Historical Financial Information present fairly in all material respects the results of operations of the Facilities for the periods covered.

(b) Except for (i) liabilities that are disclosed in this Agreement, Contracts entered into in connection herewith and schedules and exhibits hereto and thereto, and (ii) liabilities that were incurred after the Balance Sheet Date in the ordinary course of business, as of the date hereof, there are no material liabilities of any nature of Seller or any of its Affiliates relating to the Facilities or the other Purchased Assets and Assumed Liabilities required in accordance with GAAP to be disclosed on the financial statements of Seller.

4.7 <u>Permits and Approvals</u>.

(a) Seller holds all material Permits and Approvals necessary for the conduct of the business and operation of the Facilities as currently conducted and the use of the Purchased Assets by the Seller as currently used. All such Permits and Approvals are in full force and effect and unimpaired. Each Facility's pharmacies, laboratories and all other material ancillary departments located at such Facility or operated for the benefit of such Facility and included within the Purchased Assets, which are required to be specially licensed or to have Permits and Approvals, are licensed by the appropriate Governmental Entity or have such Permits and Approval issued by the appropriate Governmental Entity. Not later than January 7, 2011, Seller shall provide to Buyer a complete list of (i) all healthcare, controlled substance or similar Permits and Approvals necessary for the conduct of the business and operation of the Facilities as

currently conducted and the use of the Purchased Assets by Seller as currently used, and (ii) all other material Permits and Approvals necessary for the conduct of the business and operation of the Facilities as currently conducted and the use of the Purchased Assets by Seller as currently used.

(b) Each Facility is in compliance in all material respects with all Permits and Approvals required by Law. There are no provisions in, or Contracts relating to, any such Permits and Approvals which precludes or limits in any material respect Seller from operating any of the Facilities as they are currently operated. There is not now pending nor, to the knowledge of Seller, threatened, any action by or before any Governmental Entity (other than Buyer or its Affiliates) to revoke, cancel, rescind, modify or refuse to renew any of the Permits and Approvals (and as of the date of this Agreement, there is not pending nor, to the knowledge of Seller, threatened, any such action by or before Buyer or any of its Affiliates), and all of the material Permits or Approvals are and shall be in good standing now and as of the Closing.

4.8 <u>Intellectual Property</u>. Except for Intellectual Property constituting Excluded Assets:

- (a) Seller owns or will own, or is licensed or will be licensed or otherwise possesses or will possess all necessary rights to use, all Intellectual Property used in the Facilities owned by Seller.
- (b) To the knowledge of Seller, there is no unauthorized use, disclosure, infringement or misappropriation of any Intellectual Property rights of Seller or any of its Affiliates, any trade secret material to Seller or any of its Affiliates, or any Intellectual Property right of any third party to the extent licensed by or through Seller or any of its Affiliates, by any third party, including any employee or former employee of Seller or any of its Affiliates, relating in any way to any of the Purchased Assets. Other than in the ordinary course of business or otherwise reflected in the Historical Financial Information, there are no royalties, fees or other payments payable by Seller or any of its Affiliates, to any Person by reason of the ownership, use, sale or disposition of Intellectual Property related to any of the Purchased Assets.
- (c) Seller is not nor will be as a result of the execution and delivery of this Agreement or any of the documents described herein or the performance of its obligations under this Agreement or any of the documents described herein, in material breach of any license, sublicense or other agreement relating to the Intellectual Property or the Intellectual Property rights of any third party related to any of the Purchased Assets.

4.9 Medicare Participation/Accreditation.

(a) The Hospital is eligible to receive payment without restriction under Title XVIII of the Social Security Act ("Medicare") and Title XIX of the Social Security Act ("Medicaid") and is a "provider" with one or more valid and current provider agreements and with one or more provider numbers with the federal Medicare, all applicable state Medicaid and successor programs (the "Government Programs") through intermediaries or contract administrators. Except as set forth on Schedule 4.9 each of the other Facilities (other than physician practices) that have historically received Medicare or Medicaid reimbursement is eligible to receive payment without restriction under Medicare and Medicaid and is a "provider" with valid and current provider agreements and with one or more provider numbers with the Government Programs through intermediaries or contract administrators. Except as set forth on Schedule 4.9 the Hospital and each of the other Facilities that have historically received payments under TRICARE and successor programs is a "provider" with valid and current provider agreements and with one or more provider numbers with TRICARE and successor programs through intermediaries or contract administrators. The Hospital and the other Facilities referred to above are in compliance with the conditions of participation for the Government Programs in all material

respects and have received all Approvals necessary for capital reimbursement on the Purchased Assets. Except as set forth on Schedule 4.9, there is not pending, nor to the knowledge of Seller threatened, any proceeding or investigation under the Government Programs involving Seller or any of the Purchased Assets. The cost reports of Seller and the Hospital and the other Facilities referred to above for the Government Programs and for payment or reimbursement of any other Agency Receivables for the fiscal years through December 31, 2009 required to be filed on or before the date hereof have been properly filed and are complete and correct in all material respects. Not later than January 7, 2011, Seller shall provide Buyer with (i) a complete copy of the last three years' filed cost reports for the Hospital and the other Facilities, (ii) a list of all provider numbers and provider agreements currently utilized by the Hospital or any of the other Facilities, (iii) the identity of all fiscal intermediaries or contract administrators for each such provider number or provider agreement, and (iv) a list of all material claims, actions or appeals pending before any fiscal intermediary or contract administrator or Governmental Entity with respect to any of the last five years' filed cost reports for the Hospital or any of the other Facilities. Except as disclosed on Schedule 4.9, Seller is in material compliance with filing requirements with respect to cost reports of the Hospital and the other Facilities referred to above and such reports do not claim, and neither the Hospital nor any of the other Facilities referred to above has received payment or reimbursement in excess of, the amount provided by Law or any applicable agreement, except where excess reimbursement was noted on the cost report. Seller is in compliance in all material respects with any and all Medicare. Medicaid or other third party corporate integrity or compliance agreements or consent orders applicable to the Facilities.

- (b) All billing practices of Seller with respect to the Facilities to all third party payors, including the Government Programs and private insurance companies, have been and are in compliance with all applicable Laws, regulations and polices of such third party payors and Government Programs in all material respects. Neither Seller nor the Facilities have billed or received any payment or reimbursement in excess of amounts allowed by Law.
- (c) The Hospital is duly accredited by The Joint Commission ("<u>TJC</u>"). Seller will provide by January 14, 2011 copies of the Hospital's TJC accreditation survey report and deficiency list, if any, and the Hospital's plan of correction, if any.
- 4.10 <u>Regulatory Compliance</u>. Except as set forth on <u>Schedule 4.10</u>, Seller is in compliance in all material respects with all applicable statutes, rules, regulations and requirements of, and all agreements and orders with, Governmental Entities having jurisdiction over each Facility and the Purchased Assets and the business operation of each Facility and the Purchased Assets; provided, however, that no representation is made with respect to the Facilities' compliance with the Americans with Disabilities Act, as amended. Seller has timely filed all material forms, applications, reports, statements, data and other information required to be filed with Government Entities.
- 4.11 <u>Assumed Contracts</u>. Each Assumed Contract (i) is valid and existing, and Seller has duly performed in all material respects its obligations under each Assumed Contract to which it is a party to the extent that such obligations to perform have accrued and (ii) no material breach or default, alleged material breach or default, or event which would (with the passage of time, notice or both) constitute a material breach or default under any Assumed Contract by Seller or, to the knowledge of Seller, and except as set forth on <u>Schedule 4.11</u>, any other party or obligor with respect thereto, has occurred.

4.12 Real Property.

(a) Neither Seller nor any Affiliate of Seller has created any Encumbrance (other than Permitted Encumbrances) which will materially interfere with Buyer's use of the Purchased Assets in a manner consistent with the current use by Seller and its Affiliates. Except for medical office leases

(including the ground lease of approximately 0.5 acres of the Main Hospital Parcel, as defined in Exhibit D-1, by Seller, as ground lessor, to Kanan Albany, LLC with respect to the Kanan Albany, LLC medical office building located on such property) and food services agreements, neither Seller nor any of Seller's Affiliates has leased all or any material portion of the Facilities to any other Person that is not a Seller. Seller or one of its Affiliates owns and will convey fee simple interest in the Owned Real Property subject only to Permitted Encumbrances, and Seller or its applicable Affiliate will assign or cause the applicable Seller to assign its leasehold interest in the Leased Real Property to Buyer subject only to Permitted Encumbrances. Seller agrees that title to the Real Property shall not be altered or encumbered between the date of this Agreement and Closing except to the extent not restricted by Sections 6.2 and 6.3.

- (b) The Leased Real Property and the Owned Real Property collectively constitutes all real property necessary for or used by Seller in the operation of the Facilities.
- 4.13 <u>Personal Property</u>. Seller or its Affiliates will hold on the Closing Date good and marketable title to all tangible personal property assets and valid title to all intangible assets included in the Purchased Assets free and clear of all Encumbrances, except Permitted Encumbrances and rights of owners under Assumed Contracts or under leases or licenses of assets leased or licensed in the ordinary course of business.
- 4.14 Insurance Schedule 4.14 sets forth a true and complete list of all insurance policies or self insurance funds maintained by Seller as of the date of this Agreement covering the ownership and operation of the Purchased Assets or any of the Facilities, indicating the types of insurance, policy numbers, terms, identity of insurers and amounts and coverages (including applicable deductibles). Seller has one or more "business interruption" insurance policies in customary form and amount covering the Facilities. All of such policies are now and will be until the Closing in full force and effect on an occurrence basis with no premium arrearages. Such policies of insurance shall not be assigned to Buyer as part of the Purchased Assets and Buyer acknowledges that all of the coverages listed on Schedule 4.14 with respect to the Purchased Assets will cease on the Closing Date.

4.15 Employee Benefit Plans.

- (a) Schedule 4.15 contains a true and complete list of all the following agreements, plans or other arrangements, covering any employee of the Facilities, which are presently in effect: (i) employee benefit plans within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") and (ii) any other employee benefit plan, program, policy, or arrangement, whether written or unwritten, formal or informal, which Seller currently sponsors, or to which it has any outstanding present or future obligations to contribute or other liability, whether voluntary, contingent or otherwise (collectively, the "Plans").
- (b) The Purchased Assets are not, and Seller does not reasonably expect them to become, subject to a lien imposed under the Code or under Title I or Title IV of ERISA including liens arising by virtue of Seller considered to be aggregated with another entity pursuant to Section 414 of the Code ("ERISA Controlled Group").
- (c) Except as set forth in <u>Schedule 4.15</u>, neither Seller nor any member of Seller's ERISA Controlled Group has sponsored, contributed to or had an "obligation to contribute" (as defined in ERISA Section 4212) to a "multiemployer plan" (as defined in ERISA Sections 4001(a)(3) or 3(37)(A)) on or after September 26, 1980 on behalf of any employees of the Facilities.
- (d) Neither Seller nor any member of Seller's ERISA Controlled Group has at any time sponsored or contributed to a "single employer plan" (as defined in ERISA Section 4001(a)(14)) to

which at least two or more of the "contributing sponsors" (as defined in ERISA Section 4001(a)(13)) are not part of the same ERISA Controlled Group.

- (e) Except as set forth on <u>Schedule 4.15</u>, there are no material actions, audits or claims pending or, to Seller's knowledge, threatened against Seller with respect to Seller's maintenance of the Plans, other than routine claims for benefits and other claims that are not material.
- (f) Seller and Seller's ERISA Controlled Group have complied with all of the continuation coverage requirements of Section 1001 of the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended, and ERISA Sections 601 through 608 and with the requirements of Section 5000 of the Code.
- (g) All of Seller's Plans that are intended to satisfy Section 401 of the Code ("Retirement Plans") from which assets may be involved in a "direct rollover" (as defined in Section 401(a)(31) of the Code) or other transfer to Buyer's Retirement Plans have complied with the requirements of Section 401(a) of the Code.
- (h) All IRS Form 5500's due to have been filed with respect to the Plans have been filed.

4.16 Employees and Employee Relations.

- (a) No changes in the basis for remuneration of employees of the Facilities have been made, promised or authorized by Seller since the Balance Sheet Date, except in the ordinary and usual course. Seller has no written employment Contracts, and no Contract of any nature that provides for employment for any particular period of time or that provides any restrictions upon Seller's right to terminate employment without any post-termination payment obligation, with any Person whomsoever relating to the Facilities. Other than in the ordinary course of business, no binding agreements have been made or entered into between Seller and any employee involved in any of the Facilities regarding changes in compensation, promotion or any other change in status.
- (i) Except as set forth on Schedule 4.16, there is no pending or, to the best of Seller's knowledge, threatened employee strike, work stoppage or labor dispute, (ii) to the best of Seller's knowledge, no union representation question exists respecting any employees of Seller, no demand has been made for recognition by a labor organization by or with respect to any employees of Seller, no union organizing activities by or with respect to any employees of Seller are taking place, and none of the employees of Seller are represented by any labor union or organization, (iii) no collective bargaining agreement exists or is currently being negotiated by Seller, (iv) there is no unfair practice claim against Seller before the National Labor Relations Board, or any strike, dispute, slowdown, or stoppage pending or, to the best of Seller's knowledge, threatened against or involving the Facilities and none has occurred. (v) Seller is in compliance in all material respects with all Laws and Contracts respecting employment and employment practices, labor relations, terms and conditions of employment, and wages and hours, (vi) Seller is not engaged in any unfair labor practices, (vii) there are no pending or, to the best of Seller's knowledge, threatened complaints or charges before any Governmental Entity regarding employment discrimination, safety or other employment-related charges or complaints, wage and hour claims, unemployment compensation claims, workers' compensation claims or the like and (viii) except as otherwise provided in this Agreement, neither Buyer nor PNI will be subject to any claim or liability for severance pay as a result of the consummation of the transactions contemplated by this Agreement through the Closing.

4.17 Litigation or Proceedings.

- (a) Schedule 4.17(a) contains an accurate list and summary description of all litigation and proceedings with respect to the Facilities and the Purchased Assets (including litigation and proceedings with respect to the Facilities' employees and the Hospital's medical staff) to which Seller or any of its Affiliates are a party, as well as settlements and conciliation agreements under which Seller or any of its Affiliates have current or future obligations with respect to the Facilities or Purchased Assets. Except to the extent set forth on Schedule 4.17(a), there are no claims, actions, suits, audits, compliance reports or information requests, proceedings or investigations pending, or to the knowledge of Seller, threatened against or affecting (i) Seller or any of its Affiliates with respect to the Facilities or the Purchased Assets, or (ii) any employee who is employed at any of the Facilities, at law or in equity, or before or by any Governmental Entity (other than Buyer or its Affiliates) wherever located. Neither the representations and warranties set forth in this Section 4.17(a) nor the disclosure in Schedule 4.17(a) is intended to address PRO information, National Data Bank reports, quality review information and other physician specific confidential information.
- (b) Neither Seller nor any of its Affiliates nor any of the Facilities are in violation in any material respect under any Law relating to the operation of the Facilities (provided, however, that no representation is made with respect to the Facilities' compliance with the Americans with Disabilities Act, as amended), or under any order of any court or federal, state, municipal or other Governmental Entity (other than Buyer or its Affiliates) wherever located.
- (c) Other than as set forth on <u>Schedule 4.17(c)</u>, Seller is not subject to any outstanding judgment, order or decree with respect to the Purchased Assets.
- (d) Seller has not engaged in any transaction that would reasonably be expected to subject Seller (or any successors in interest) to any avoidance action with respect to the Purchased Assets. Without limiting the generality of the foregoing, Seller has not with respect to the Purchased Assets (i) received any material payments from its or their account debtors outside the ordinary and usual course, (ii) acquired or sold any asset other than for reasonably equivalent value or (iii) conducted any business with any debtor-in-possession or bankrupt estate other than in the ordinary and usual course.

4.18 Tax Matters. Except as set forth on Schedule 4.18:

- (a) All tax returns, including income tax returns, sales tax returns, employee payroll tax returns, employee unemployment tax returns and franchise tax returns, for periods prior to and including Closing which are required to be filed by Seller (collectively "Returns") have been filed or will be filed within the time (including any valid extensions thereof) and in the manner provided by Law, and all Returns are or will be true and correct and accurately reflect the tax liabilities of Seller in all material respects, and all amounts shown due on such Tax Returns have been or will be paid on a timely basis;
- (b) All federal, state, county and local income, franchise, payroll, withholding, property, sales, use and all other taxes, penalties, interest, and any other statutory additions which have become or are due with respect to the Purchased Assets and any assessments received by Seller (collectively "Payable Tax Items"), have been or by the Closing Date will be paid regarding any period ended on or prior to the Closing Date or adequately reflected on the Closing Balance Sheet (or otherwise reserved for by a Seller or its Affiliates), whether shown on such returns or not;
 - (c) There are no tax liens on any of the Purchased Assets;
- (d) Proper and accurate amounts have been withheld by Seller for all periods prior to the Closing in compliance with the payroll tax and other withholding provisions of all applicable Laws, and all of such amounts have been duly and validly remitted to the proper taxing authority; and

- (e) No notice of a claim or pending investigation has been received, or to the knowledge of Seller, has been threatened, by any state, local or other jurisdiction, alleging that Seller has a duty to file tax returns and pay taxes or is otherwise subject to the taxing authority of any jurisdiction, nor has Seller received any notice or questionnaire from any jurisdiction which suggests or asserts that Seller may have a duty to file such returns and pay such taxes, or otherwise is subject to the taxing authority of such jurisdiction.
- 4.19 <u>Environmental Matters</u>. Except as set forth on <u>Schedule 4.19</u> or in any environmental report listed therein:
- (a) Seller has materially complied and is in material compliance (with respect to each of the Facilities) with, and the Real Property and all improvements on the Real Property are in material compliance with, all Environmental Laws.
- (b) Seller has no liability under any Environmental Law with respect to any of the Facilities or the Real Property, nor is Seller responsible for any liability of any other Person under any Environmental Law with respect to any of the Facilities or the Real Property. There are no pending or, to the knowledge of Seller, threatened actions, suits, orders, claims, legal proceedings or other proceedings based on, and neither Seller nor any of its Affiliates have received any formal or informal written notice of any complaint, order, directive, citation, notice of responsibility, notice of potential responsibility, or information request from any Governmental Entity or any other Person or knows or suspects any fact(s) which would reasonably be expected to form the basis for any such actions or notices arising out of or attributable to any Environmental Condition.
- (c) Seller has been duly issued, and currently has and will maintain through the Closing Date, all material Approvals and Permits required under any Environmental Law with respect to any of the Facilities. A true and complete list of such Permits, all of which are valid and in full force and effect, is set forth in Schedule 4.19. Seller is in material compliance (with respect to each Facility) with, and the Real Property and all improvements on the Real Property are in material compliance with, all Approvals and Permits. Except in accordance with such Approvals and Permits, there has been no release of material regulated by such Approvals and Permits at, on, under, or from the Real Property in violation of Environmental Laws.
- (d) The Real Property contains no underground improvements, including treatment or storage tanks, or underground piping associated with such tanks, used currently or in the past for the management of Hazardous Materials, and Seller has not used any portion of the Real Property as a dump or landfill.
- (e) Seller will promptly furnish to Buyer written notice of any material Environmental Condition or of any actions or notices described in this <u>Section 4.19</u>.
- (f) Except to the extent permitted under Environmental Laws, neither PCBs, lead paint, nor asbestos-containing materials are present on or in the Real Property.
- (g) No Encumbrance in favor of any Person relating to or in connection with any claim under any Environmental Law has been filed or has attached to the Real Property, other than Permitted Encumbrances.
- 4.20 <u>Immigration Act</u>. Seller is in compliance in all material respects with the terms and provisions of the Immigration Act in all material respects with respect to each of the Facilities. For each employee of Seller employed in any of the Facilities for whom compliance with the Immigration Act by

Seller is required, Seller has obtained and retained a complete and true copy of each such employee's Form 19 (Employment Eligibility Verification Form) and all other records or documents prepared, procured or retained by Seller pursuant to the Immigration Act to the extent Seller is required to do so under the Immigration Act. Seller has not been cited, fined, served with a Notice of Intent to Fine or with a Cease and Desist Order (as such terms are defined in the Immigration Act) at any of the Facilities, nor, to the knowledge of Seller, has any action or administrative proceeding been initiated or threatened against Seller in connection with any of the Facilities, by reason of any actual or alleged failure to comply with the Immigration Act.

- Employment Loss, and type of Employment Loss (termination, layoff or reduction in work hours) of each employee of Seller and its Affiliates employed in any of the Facilities who has experienced an Employment Loss in the ninety (90) days preceding the date of this Agreement. Except as set forth in this Agreement, Seller does not presently intend to take any action that would result in an Employment Loss by any employee of Seller employed in any of the Facilities between the date of this Agreement and the Closing Date. "Employment Loss" for this purpose shall mean (i) an employment termination, other than a discharge for cause, voluntary departure, or retirement, (ii) a layoff exceeding six (6) months or (iii) a reduction in hours of work of more than fifty percent (50%), and "employee" shall mean any employee, including officers, managers and supervisors, but excluding employees who are employed for an average of fewer than 20 hours per week or who have been employed for fewer than six of the preceding 12 months.
- 4.22 OSHA. Seller has complied with all applicable Laws relating to employee health and safety in all material respects, and Seller has not received any written notice from any Governmental Entity (other than Buyer or its Affiliates) that past or present conditions of the Facilities or Purchased Assets violate any applicable legal requirements or otherwise will be made the basis of any claim, proceeding, or investigation, based on OSHA violations or otherwise related to employee health and safety.
- 4.23 <u>Inventory</u>. All of the Inventory existing on the date hereof will exist on the Closing Date, except for (i) Inventory exhausted or added in the ordinary course of business between the date of this Agreement and the Closing Date, and (ii) any Excluded Inventory. Except to the extent of reserves reflected in the Facilities' financial statements as of the applicable date, all of the Inventory on hand on the date of this Agreement and to be on hand on the Closing Date, consists and will consist in all material respects of items of a quality usable or saleable for the services currently being offered through the Hospital or the Facilities. The quantities of all Inventory are reasonable and justified under the normal operations of each of the Facilities.
- 4.24 <u>Absence of Changes</u>. Except as set forth in <u>Schedule 4.24</u>, between the Balance Sheet Date and the date hereof, there has not been any transaction or occurrence in which Seller or any of its Affiliates, in connection with the Facilities and Purchased Assets, has:
- (a) suffered any material damage, destruction or loss with respect to or affecting any of the Purchased Assets;
- (b) written down or written up in any material amount the value of any Inventory (including write-downs by reason of shrinkage or markdowns), determined as collectible any material Other Receivable or any portion thereof which was previously considered uncollectible, or written off as uncollectible any material Other Receivable or any portion thereof, except for write-downs, write-ups, and write-offs in the ordinary course of business;

- (c) disposed of or permitted to lapse any right to the use of any Intellectual Property;
- (d) made any material capital expenditure or commitment for additions to property, plant, equipment, intangible or capital assets or for any other purpose, other than for emergency repairs or replacement, and except for the capital expenditures disclosed on <u>Schedule 4.24(d)</u>;
- (e) sold, transferred or otherwise disposed of any of the Purchased Assets except in the ordinary course of business;
- (f) granted or incurred any obligation for any increase in the compensation of any employee who is employed at the Facilities (including any increase pursuant to any bonus, pension, profit-sharing, retirement, or other plan or commitment) except in the ordinary course of business;
- (g) made any change in any method of accounting or accounting principle, practice, or policy;
- (h) taken any other action neither in the ordinary course of business nor provided for in this Agreement; or
- (i) agreed, so as to legally bind Buyer or affect the Purchased Assets, whether in writing or otherwise, to take any of the actions set forth in this <u>Section 4.24</u> and not otherwise permitted by this Agreement.
- 4.25 <u>Solvency</u>. Seller is not insolvent and Seller will not be rendered insolvent as a result of any of the transactions contemplated by this Agreement. For purposes hereof, the term "<u>solvent</u>" means that: (a) the fair salable value of Seller's tangible assets is in excess of the total amount of their liabilities (including for purposes of this definition all liabilities, whether or not reflected on a balance sheet prepared in accordance with generally accepted accounting principles, and whether direct or indirect, fixed or contingent, secured or unsecured, and disputed or undisputed); (b) Seller is able to pay its debts or obligations in the ordinary course as they mature; and (c) Seller has capital sufficient to carry on its businesses and all businesses in which it is about to engage.
- 4.26 <u>Statements True and Correct</u>. This Agreement and the Schedules hereto do not include, as of the date hereof, any untrue statement of a material fact or omit to state any material fact necessary to make the statements made in this Agreement with respect to the Seller and the Purchased Assets not misleading.
- 4.27 <u>Medical Staff</u>. Seller has provided to Buyer true, correct and complete copies of the bylaws, rules and regulations of the medical staff of the Hospital. Not later than January 7, 2011, Seller will provide to Buyer the names of all medical staff members, including allied health practitioners.

5. REPRESENTATIONS AND WARRANTIES OF BUYER

As of the date hereof and as of the Closing Date, Buyer represents and warrants to Seller the following (except to the extent any of the following speaks as of a specific date, in which case Buyer makes such representations and warranties as of the date specified):

5.1 Authority Capacity; Corporate Capacity.

(a) Buyer is a hospital authority validly existing, duly organized and in good standing under the Laws of the State of Georgia. Buyer has the requisite power and authority to enter into

this Agreement, perform its obligations hereunder and to conduct its businesses as now being conducted. The execution and delivery by Buyer of the Agreement and the documents described herein, the performance by Buyer of its obligations under the Agreement and the documents described herein and the consummation by Buyer of the transactions contemplated by the Agreement and the documents described herein have been duly and validly authorized and approved by all necessary hospital authority actions on the part of Buyer, none of which actions have been modified or rescinded and all of which actions remain in full force and effect. Buyer is and will be at the Closing the lessor or owner, directly or indirectly, of the hospitals owned, operated or managed by Buyer, PPHS, PNI and their respective Affiliates.

- organized and in good standing under the Laws of the State of Georgia. PPHS and PNI each has the requisite power and authority to enter into this Agreement, perform its obligations hereunder and to conduct its business as now being conducted. The execution and delivery by PPHS and PNI of the Agreement and the documents described herein, the performance by PPHS and PNI of their respective obligations under the Agreement and the documents described herein and the consummation by PPHS and PNI of the transactions contemplated by the Agreement and the documents described herein have been duly and validly authorized and approved by all necessary corporate actions on the part of PPHS and PNI, none of which actions have been modified or rescinded and all of which actions remain in full force and effect.
- 5.2 <u>Powers; Consents; Absence of Conflicts With Other Agreements, Etc.</u> The execution, delivery and performance of this Agreement by Buyer, PPHS or PNI and all other agreements referenced in or ancillary hereto to which Buyer, PPHS or PNI are parties and the consummation of the transactions contemplated herein by Buyer, PPHS and PNI:
- (a) are within the powers of Buyer conferred by the Hospital Authorities Law and have been approved by all requisite authority action;
- (b) are within the corporate powers of PPHS and PNI and are not in contravention of the terms of PPHS's or PNI's certificate of incorporation, bylaws, or other governing documents and have been approved by all requisite corporate action;
- (c) except as described in <u>Schedule 5.2</u>, do not require any Approval or Permit of, or filing with, any Governmental Entity bearing on the validity of this Agreement which is required by Law, and
- (d) will neither conflict with nor result in any material breach or contravention of, or the creation of any Encumbrance under, any indenture, agreement, lease, instrument or understanding to which Buyer, PPHS or PNI are parties or by which Buyer, PPHS or PNI are bound.
- 5.3 Binding Effect. This Agreement and all other agreements to which Buyer, PPHS, PNI and their respective Affiliates will become parties hereunder are and will constitute the valid and legally binding obligations of Buyer, PPHS, PNI and their respective Affiliates and are and will be enforceable against them in accordance with the respective terms hereof and thereof, except as enforceability against Buyer, PPHS, and PNI may be restricted, limited or delayed by applicable bankruptcy or other Laws affecting creditors' rights generally and except as enforceability may be subject to general principles of equity.
- 5.4 <u>Litigation</u>. There is no claim, action, suit, proceeding or investigation pending or, to the knowledge of Buyer, PPHS, or PNI, threatened against or affecting Buyer, PPHS or PNI that has or

would reasonably be expected to have a material adverse effect on Buyer's, PPHS's, or PNI's ability to perform this Agreement or any aspect of the transactions contemplated hereby.

5.5 Buyer Acknowledgements.

- (a) Buyer has: (i) such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of the transactions contemplated by this Agreement, including the purchase of the Facilities and the Purchased Assets and the assumption of the Assumed Liabilities; (ii) the ability to bear the economic risk in connection with the consummation of the transactions contemplated by this Agreement, including a complete loss of future revenue, income or profits related to the Facilities and the Purchased Assets and (iii) been furnished with and has had access to such information as it has considered necessary to make a determination as to the purchase of the Facilities and the Purchased Assets and the Assumed Liabilities.
- (b) Buyer's decision to purchase the Facilities and the Purchased Assets and to assume the Assumed Liabilities, has been (i) made voluntarily and of its own accord, based upon, Buyer's (A) extensive knowledge and experience in financial and business matters relating to owning and operating general acute care hospitals, (B) consultations with its advisors and (C) investigation of the business, assets, risks and prospects of the Facilities and Purchased Assets and (ii) made without relying on any statement (whether oral or written), or any representation or warranty of Seller or any affiliate, officer or director of Seller, other than the representations and warranties expressly contained in this Agreement and the other Contracts executed in connection herewith.

5.6 Availability of Funds; Solvency.

- (a) Buyer has the ability to obtain funds in cash in amounts equal to the Purchase Price and will at the Closing have immediately available funds in cash, which are sufficient to pay the Purchase Price and to pay any other amounts payable pursuant to this Agreement and to consummate the transactions contemplated by this Agreement.
- (b) Neither Buyer, PPHS nor PNI is insolvent and neither Buyer, PPHS nor PNI will be rendered insolvent as a result of any of the transactions contemplated by this Agreement. For purposes hereof, the term "solvent" means that: (a) the fair salable value of Buyer's tangible assets is in excess of the total amount of its liabilities (including for purposes of this definition all liabilities, whether or not reflected on a balance sheet prepared in accordance with generally accepted accounting principles, and whether direct or indirect, fixed or contingent, secured or unsecured, and disputed or undisputed); (b) Buyer is able to pay its debts or obligations in the ordinary course as they mature; and (c) Buyer has capital sufficient to carry on its businesses and all businesses which Buyer is about to engage.
- 5.7 Antitrust Matters. (a) Neither the execution and delivery of this Agreement nor the consummation or Closing of the transactions contemplated hereby will violate any applicable state or federal antitrust or similar laws, and (b) the closing of the transaction does not require the filing of a premerger notification and report form pursuant to the HSR Act.
- 5.8 <u>Statements True and Correct</u>. This Agreement and the Schedules do not include, as of the date hereof, any untrue statement of a material fact or omit to state any material fact necessary to make the statements made in this Agreement with respect to Buyer and the Purchased Assets not misleading.

33

6. COVENANTS OF SELLER AND BUYER

6.1 Completion of Schedules; Information; Assumed Contracts and Rejected Contracts.

- (a) (i) Any Schedule required to be delivered by Seller under this Agreement shall be delivered by Seller to Buyer as promptly as practicable but in no event later than January 14, 2011. Seller shall have the right to supplement or amend any such Schedule until January 14, 2011. Additionally, Seller shall have the right to supplement or amend Exhibits D-1 and D-2 until January 7, 2011, in order to reflect (A) the addition of any Owned Real Property or Leased Real Property not reflected on such exhibits as of the date of this Agreement, (B) the deletion of any property (other than the Main Hospital Parcel, as identified in Exhibit D-1) from Exhibit D-1 that Seller or its applicable Affiliate shall have conveyed prior to the date hereof, or (C) the deletion of any property from Exhibit D-2 if the lease or sublease with respect to such property shall have terminated or expired as of the date hereof (or to reflect the fact that the lease or sublease applicable to any such property will terminate or expire in accordance with its terms before the anticipated Closing Date).
- (ii) Seller shall deliver to Buyer by no later than January 7, 2011 its proposed form of Exhibits E and E. The parties will use good faith efforts to agree upon Exhibits E and E by January 14, 2011 (and in any event by the Closing Date). If the parties are unable to agree upon Exhibit E or \underline{F} by the Closing, the parties will resolve any such matter in accordance with Sections 2.6 and 2.7.
- (iii) Buyer shall have the right to supplement or amend <u>Schedule 5.2</u> until January 14, 2011, to the extent any such supplement or amendment is required as a result of (or in response to) Seller's delivery of Approvals or Permits to Buyer as contemplated by <u>Section 4.7(a)</u>.
- (iv) The parties will use good faith efforts to agree upon <u>Schedule 10.2</u> by January 14, 2011 (and in any event by the Closing Date).
- (v) Schedules that Seller or Buyer delivers, supplements or amends in accordance with this Section 6.1(a) (or, with respect to Schedule 10.2, upon which Buyer and seller agree in accordance with this Section 6.1(a)) shall have the same force and effect under this Agreement as though delivered simultaneously with the parties' execution and delivery of this Agreement.
- (vi) Buyer and Seller agree that the right of the parties to deliver, supplement or amend the schedules and exhibits to this Agreement following the execution and delivery hereof does not limit or impact the binding nature of this Agreement or the respective representations and warranties of Buyer at Section 5.3 and Seller at Section 4.5.
- (b) In addition to Buyer's receipt of the financial statements and financial information contained on <u>Schedule 4.6</u> hereto, by December 28, 2010, Buyer will furnish a list of financial and operating data and other information pertaining to the business and properties of Seller relating to the Purchased Assets which Buyer requires in order to have an effective transition of ownership and operations at Closing. Not later than January 14, 2011, Seller shall use its best efforts to provide the requested information, subject to the restrictions set forth below in this <u>Section 6.1(b)</u>. Between the date of this Agreement and the Closing Date, to the extent permitted by Law, Seller shall afford to the authorized representatives and agents of Buyer fair and reasonable access to and the right to inspect the plants, properties, books and records of Seller relating to the Purchased Assets, and will

furnish Buyer with such additional financial and operating data and other information as to the business and properties of Seller relating to the Purchased Assets as Buyer may from time to time reasonably request including all items set forth on any of Seller's Schedules, Exhibits or deliveries under this Agreement. Buyer's right of access and inspection shall be made in such a manner as not to interfere unreasonably with the operation of the Purchased Assets. In this regard, Buyer agrees that such inspection shall not take place, and no employees or other personnel at any Facility shall be contacted by Buyer's representatives, without first coordinating such contact or inspection with Gregg Gerken or his designee at least 24 hours prior to such requested inspection or contact. Notwithstanding the foregoing, Buyer understands that (i) with respect to documents and information deemed by Seller in good faith to be market sensitive or competitive in nature, (A) Seller will identify such documents and information to Buyer, (B) if requested by Buyer, Seller will provide such documents and information to Buyer's outside attorneys and accountants (who will be bound by confidentiality agreements) for their review, and (C) any report by such attorneys and accountants to Buyer with respect to such documents and information will be in writing and subject to prior review and reasonable approval by Seller to confirm that any market sensitive or competitive information is not made available to Buyer, (ii) litigation and other materials (including internal/external legal audit letters or reviews, PRO information, National Data Bank reports, quality review information and other physician specific confidential information, and information subject to the confidentiality requirements of HIPAA) that are deemed privileged or confidential by Seller will not be made available to Buyer, and (iii) Seller shall not be obligated to generate or produce information in any prescribed format not customarily produced by Seller.

Not later than January 14, 2011, Seller will deliver to Buyer a list of all material Contracts. Except as provided in the last sentence of Section 6.1(b), all items set forth on such lists will be made available to Buyer pursuant to Section 6.1(b). Buyer may reject any Contract (i) that violates applicable federal or state Law, (ii) if the assumption of the contract or lease would cause Buyer to violate applicable Georgia law, including, without limitation, the Hospital Authorities Law, (iii) if the assumption of the Contract would cause PPHS to lose its tax exempt status, (iv) that imposes material and adverse limitations on Buyer's ability to operate the Purchased Assets in the same manner as they were operated by Seller and its Affiliates, (v) if the Contract contains exclusivity provisions, covenants not to compete or similar types of contractual terms that would, by its terms, limit Buyer's activities at its other health care facilities to any material and adverse extent, and (vi) that contains material and adverse obligations (either monetary or non-monetary) after the Closing that are extraordinary relative to the obligations of Seller and its Affiliates prior to the Closing (i.e. "balloon payments" or "balloon obligations"), but only to the extent that such future obligations are not reflected in the Historical Financial Information (any of the foregoing Contracts ultimately not assumed by Buyer in accordance with provisions of this Section 6.1(c) (after the final resolution of the arbitration procedures) are collectively referred to as "Rejected Contracts"). If Buyer seeks to reject any Contract, not later than January 26, 2011, Buyer shall notify Seller in writing of its intention to reject one or more Contracts and its reasons therefor. For two (2) business days after receipt of such notice, Buyer and Seller shall consult in good faith as to what action, if any, should be taken with respect to any such Contract to address concerns raised by Buyer. If Buyer and Seller do not jointly agree on whether any contract falls within the rejection criteria set forth in this Section 6.1(c) within such two (2) business day period, such Contract shall be deemed an Assumed Contract and not a Rejected Contract and the matter will be submitted to arbitration pursuant to the provisions of Section 12.3(b); provided that, if the Contract in question is one that Buyer reasonably believes violates applicable Law or would likely result in the revocation of Buyer's tax exempt status, then such Contract shall be deemed a Rejected Contract not an Assumed Contract and the matter will be submitted to arbitration pursuant to the provisions of Section 12.3(b). Notwithstanding the foregoing, if Buyer seeks to reject any Contract, Seller has the right to address any concerns that Buyer has raised with respect to such Contract by negotiating a related amendment or modification of such Contract with the other party to such Contract or, where applicable, by providing a legal opinion that the Contract in question is compliant with applicable Law or will not result in the revocation of Buyer's tax exempt

status, as the case may be, and if such amendment or modification remedies the basis for rejection or such opinion is obtained, then such Contract shall be an Assumed Contract and shall not be subject to arbitration hereunder.

- With respect to physician agreements that are Contracts, the parties agree as follows: (i) if (A) any such Contract contains an exclusivity provision that conflicts with a physician agreement at another health care facility owned by Buyer and (B) the physician agreement at such other health care facility does not contain an exclusivity provision that conflicts with the physician agreement that is a Contract, Seller shall assume and be responsible for any liability incurred by Seller and Buyer in connection with the resulting breach or the termination of such other agreement, (ii) if (A) any such Contract contains an exclusivity provision that conflicts with a physician agreement at another health care facility owned by Buyer and (B) the physician agreement at such other health care facility also contains an exclusivity provision that conflicts with the physician agreement that is a Contract, Seller (collectively) and Buyer shall each assume and be responsible for one-half of any liability incurred by Buyer and Seller in connection with the resulting breach or termination of such Contract or, at Buyer's option, such agreement, and (iii) if (A) Buyer has a physician agreement at another health care facility owned by Buyer that conflicts with the physician agreement that is a Contract and (y) the physician agreement that is a Contract does not contain an exclusivity provision that conflicts with the physician agreement at Buyer's other health care facility, Buyer shall assume and be responsible for any liability incurred by Seller and Buyer in connection with the resulting breach or the termination of such other agreement or, at Buyer's option, such Contract.
- (e) With respect to any Contract set forth on Schedule 6.1(e), Seller has the right to cause such Contract to be partially assigned, amended, modified or replicated so as to result in Buyer being a party to a Contract with such third-party containing terms substantially similar, in the aggregate, to the terms of such Contract as they presently exist with respect to the Facilities; provided, however, that no such partial assignment, amendment, modification or replication of any such Contract is a condition to the Closing.
- 6.2 Operations. From the date hereof until the Closing Date, except as set forth in Schedule 6.2, Seller shall, with respect to the Purchased Assets (unless prior written consent of Buyer is received, which will not be unreasonably withheld):
- (a) carry on its business related to the Purchased Assets in substantially the same manner as it has heretofore and not make any material change in personnel, operations, finance, accounting policies, or the Purchased Assets other than in the ordinary course of business;
- (b) use its commercially reasonable efforts to maintain the Purchased Assets and all parts thereof in as good working order and condition as at present, ordinary wear and tear excepted;
- (c) use its commercially reasonable efforts to perform in all material respects all of its obligations under Assumed Contracts or Excluded Contracts relating to or affecting the Purchased Assets and the Facilities' business and operation;
- (d) use its commercially reasonable efforts to keep in full force and effect present insurance policies or other comparable insurance on the Purchased Assets;
- (e) use its commercially reasonable efforts to maintain and preserve its business organization with respect to the Facilities intact, retain its present employees at the Facilities and maintain its relationship with physicians, medical staff, suppliers, customers and others having business relations with the Facilities;

- (f) permit and allow reasonable access by Buyer to make offers of post-Closing employment to any of Seller's personnel, which personnel shall be allowed to accept such offers without penalty, competing offer or interference, and to establish relationships with physicians, medical staff and others having business relations with Seller; provided that Seller shall have complied with the terms of Section 6.1(a) in connection with such access;
- (g) use its commercially reasonable efforts to render title to the Purchased Assets free and clear of all Encumbrances (except for the Permitted Encumbrances), and to obtain appropriate consents from all Persons and Governmental Entities (other than Buyer or its Affiliates) necessary for the assignment to Purchaser of the Purchased Assets and consummation of the transactions contemplated by this Agreement;
- (h) use its commercially reasonable efforts to cure any deficiencies cited by any Governmental Entity (other than Buyer or its Affiliates) or TJC in the most recent surveys conducted by each or develop and timely implement a plan of correction that is acceptable to any Governmental Entity (other than Buyer or its Affiliates) or TJC;
- (i) use its commercially reasonable efforts to keep available for Buyer the services of the present officers, employees, medical staff, consultants, sales agents and representatives of the Facilities:
- (j) comply in all material respects with all Laws applicable to the conduct of the business and operation of the Facilities;
 - (k) maintain the levels and quality of Inventory existing on the date hereof;
- (l) continue to collect accounts receivable and pay accounts payable with respect to the Facilities in the ordinary course of business;
- (m) use its commercially reasonable efforts to maintain all Approvals and Permits relating to the Facilities, Purchased Assets and Assumed Liabilities in good standing; and
- (n) promptly notify Buyer of any material and adverse change to the Purchased Assets.
- 6.3 <u>Negative Covenants</u>. From the date hereof to the Closing Date, except as set forth in Schedule 6.3, Seller will not, with respect to the business or operation of the Facilities or otherwise regarding the Purchased Assets, without the prior written consent of Buyer, which will not be unreasonably withheld:
- (a) enter into any contract or commitment, or incur or agree to incur any liability, except where such contract or commitment or other matter (i) is in the ordinary course of business, and (ii) is either (A) terminable without cause or penalty within ninety (90) days following Closing, (B) would not result in a payment or penalty in excess of \$50,000 if terminated following Closing, or (C) in the event the Closing occurs after January 31, 2011, either (1) if such Contract must have a minimum term, renewal term or extension period without optional termination rights under Applicable Law, is a Contract that includes such minimum term or period, or (2) is a renewal in the ordinary course of business of a Contract that exists as of the date of this Agreement, including the exercise of an optional extension or renewal thereof;

- (b) increase compensation payable or to become payable or make a bonus payment to or otherwise enter into one or more bonus agreements with any employee (other than members of Senior Management) or agent or under any personal services contract, except in the ordinary course of business in accordance with existing personnel policies or to induce employees to remain through the Closing; provided, however, that notwithstanding the foregoing Seller may not generally increase rates of compensation in order to induce employees to remain through Closing;
- (c) sell, assign or otherwise transfer or dispose of any Purchased Assets, except in the ordinary course of business;
- (d) except in the ordinary course of business (i) materially amend, modify or terminate any Assumed Contract; (ii) by action or inaction, abandon, terminate, cancel, forfeit, waive or release Seller's material rights, in whole or in part, with respect to the Purchased Assets, or encumber any of the Purchased Assets, except for Permitted Encumbrances; (iii) effect any corporate merger, business combination, reorganization or similar transaction or take any other action, corporate or otherwise, which could reasonably be expected to affect adversely Seller's ability to perform in accordance with this Agreement; (iv) cancel or permit the cancellation or lapse of insurance coverage on the Purchased Assets or the Facilities; or (v) settle any dispute or threatened dispute with any Governmental Entity regarding the Purchased Assets in a manner that materially and adversely affects Buyer;
- (e) except Permitted Encumbrances, create, assume or permit to exist any new Encumbrance upon any of the Purchased Assets;
- (f) take any other action materially outside the ordinary course of business except as otherwise permitted hereunder; or
- Buyer following the Closing in excess of \$50,000.00 for additions to property, plant, equipment, intangible, or capital assets or for any other purpose, other than for emergency repairs or replacement (the parties acknowledge that this clause (g) does not limit Seller's right to make capital expenditures or commitments for capital expenditures that could result in the Purchase Price being increased by the Capital Expenditures Amount or to make capital expenditures or commitments for capital expenditures for which Seller shall be economically responsible).

6.4 Notification of Certain Matters.

- (a) At any time from the date of this Agreement to the Closing Date, Seller shall give prompt written notice to Buyer of (i) the occurrence, or failure to occur, of any event that has caused any representation or warranty of Seller contained in this Agreement to be untrue in any material respect and (ii) any failure of Seller to comply with or satisfy, in any material respect, any covenant, condition or agreement to be complied with or satisfied by it under this Agreement. Such notice shall provide a reasonably detailed description of the relevant circumstances and shall include the amount which Seller believes, based on facts known to it, would be payable by Seller pursuant to the indemnification provisions hereof.
- (b) At any time from the date of this Agreement to the Closing Date, Buyer shall give prompt notice to Seller of (i) the occurrence, or failure to occur, of any event that has caused any representation or warranty of Buyer contained in this Agreement to be untrue in any material respect and (ii) any failure of Buyer to comply with or satisfy, in any material respect, any covenant, condition or agreement to be complied with or satisfied by it under this Agreement. Such notice shall provide a reasonably detailed description of the relevant circumstances and shall include the amount which Buyer

believes, based on facts known to it, would be payable by Buyer pursuant to the indemnification provisions hereof.

- 6.5 <u>HSR Notification</u> Notwithstanding Buyer's representations and warranties regarding the HSR Act and other antitrust laws in Article 5, if the parties hereafter agree that the consummation of the transactions contemplated hereby requires the parties to file premerger notification reports with the FTC and the Justice Department (or filings or notifications with a state agency under any applicable state antitrust Law), the provisions of <u>Section 6.5(a)</u> and (b) will apply to the parties' obligations.
- Within 15 business days following the parties' agreement that the consummation of the transactions contemplated hereby requires the parties to file premerger notification reports with the FTC and the Justice Department (or filings or reports with a state agency under any applicable state antitrust Law), Seller and Buyer shall each file, as applicable, a premerger notification with the FTC and the Justice Department or the necessary filings or notifications with the applicable state agency. From the date of such filing until the Closing Date, Seller and Buyer shall file all reports or other documents required or requested by the FTC or the Justice Department (or such state agency) under the HSR Act or otherwise, and will comply promptly with any requests by the FTC or the Justice Department (or such state agency) for additional information concerning such transactions, so that the waiting period specified in the HSR Act (or the applicable state Law) will expire as soon as reasonably possible after the execution and delivery of this Agreement. Buyer agrees to pay all application fees required in connection with any filing required under the HSR Act (or any such state Law), and Buyer and Seller agree to take all commercially reasonable actions necessary to assure that the waiting period imposed under the HSR Act (or such state Law) terminates or expires as soon as commercially practicable, but in no event later than the date specified in Section 10.3(a)(ii). Without limiting the foregoing, Seller and Buyer agree to use commercially reasonable efforts to cooperate and oppose any preliminary injunction sought by any Governmental Entity (other than Buyer or its Affiliates) preventing the consummation of the transactions contemplated by this Agreement.
- (b) Seller and Buyer shall cause their respective counsel to furnish each other such necessary information and reasonable assistance as the other may reasonably request in connection with its preparation of necessary filings or submissions under the provisions of the HSR Act or any applicable state Law. Seller and Buyer will cause their respective counsel to supply to each other copies of all correspondence, filings or written communications by such party or its Affiliates with any Governmental Entity or staff members thereof, with respect to the transactions contemplated by this Agreement and any related or contemplated transactions, except for documents filed pursuant to Item 4(c) of the Hart-Scott-Rodino Notification and Report Form (or the applicable state Law), or communications regarding the same documents or information submitted in response to any request for additional information or documents pursuant to the HSR Act (or the applicable state Law) which reveal Seller's or Buyer's negotiating objectives or strategies or purchase price expectations.

6.6 Approvals; Consented Assignment.

Seller will (i) cooperate with one another and take all reasonable steps to obtain, as promptly as practicable, all Approvals and Permits of any Governmental Entities required of either party to consummate the transactions contemplated by this Agreement and (ii) provide such other information and communications to any Governmental Entity as may be reasonably requested. The parties acknowledge and agree that Seller shall be responsible for obtaining all Approvals and Permits required for Seller to transfer the Purchased Assets to Buyer, that Buyer shall be responsible for obtaining all Approvals and Permits required for Buyer to acquire the Purchased Assets from Seller and to operate the Purchased Assets following the Closing (e.g., hospital licenses for the Hospital), and that, upon request, Seller and

Buyer each will cooperate with the other party in all reasonable respects in such party's efforts to obtain the Approvals and Permits for which such party is responsible. Additionally, between the date of this Agreement and the Closing Date, Seller and Buyer agree to use commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective as promptly as practicable the transactions contemplated by this Agreement and to cooperate with the other in connection with the foregoing, including using commercially reasonable efforts (x) to prevent entry of, or lift or rescind any order, decree, ruling issued by (or any other action taken by) any court or other Governmental Authority of competent jurisdiction restraining, enjoining or otherwise prohibiting the transactions contemplated hereby or the consummation of such transactions, and (y) to fulfill all conditions to this Agreement. Buyer and Seller shall notify and keep the other advised as to any material communication from any Governmental Authority regarding any of the transactions contemplated hereby and any proceeding or investigation pending and known to such party or, to its knowledge, threatened, which challenges the transactions contemplated hereby.

- Contract is required in connection with the transactions contemplated by this Agreement, Seller shall use commercially reasonable efforts to obtain such consent with the assistance of Buyer. Buyer shall use commercially reasonable efforts to cooperate with Seller to obtain such consent prior to the Closing Date. Buyer shall have no liability to Seller for losses, liabilities, damages, costs (including court costs and costs of appeal) and expenses (including attorney's fees) incurred as the result of any action, proceeding, claim or demand brought by any third party to an Assumed Contract relating to the failure of Seller to obtain any consent to assign an Assumed Contract in connection with the transactions contemplated by this Agreement. In the event that any such consent is not obtained as of the Closing, (i) such Assumed Contract will be assigned to and assumed by Buyer under this Agreement effective on the Closing Date, and (ii) at the request of Buyer, Seller will use commercially reasonable efforts to cooperate with Buyer to obtain for Buyer the benefits of such Assumed Contract; provided, however, that Seller will not be responsible to Buyer for any loss of benefit or similar damages that Buyer may suffer if Buyer is not able to obtain the benefits of such Assumed Contract.
- 6.7 Additional Financial Information. Within thirty (30) days following the end of each calendar month prior to the Closing Date, Seller will deliver to Buyer, copies of the unaudited balance sheets and the related unaudited income statements relating to the Facilities for each month then ended. Such financial statements shall have been prepared in accordance with GAAP and the books and records of the Facilities, the balance sheets contained therein shall fairly present in all material respects the financial position of the Facilities at the date of such balance sheet and the income statements contained therein shall present fairly in all material respects the results of operations of the Facilities for the period indicated.
- 6.8 Books and Records. Until the later to occur of (i) the final adjudication of any dispute or investigation involving liabilities, federal, state or local taxes or under the Medicare or Medicaid programs arising out of the business, operations or affairs of Seller relating to the Purchased Assets before the Closing Date, or (ii) the running of applicable statutes of limitations, Buyer will maintain in the ordinary course of business all books and records of Seller and its Affiliates constituting a part of the Purchased Assets which are maintained at or by the Facilities and which relate to the pre-Closing business, operations and affairs of Seller relating to the Purchased Assets to the extent reasonably necessary in connection with any tax, Medicare or Medicaid liability or other matter reasonably relating to the Purchased Assets for any period ending at or before the Closing Date.
- 6.9 <u>Non-Solicitation</u>. If the transactions contemplated by this Agreement are not consummated, neither Buyer, PPHS, PNI nor their respective Affiliates will, directly or indirectly, solicit for employment or hire any employee of Seller or the Facilities for a period of one year from and after

termination of this Agreement; provided, however, that with respect to employees other than officers or department heads, this clause will not prohibit Buyer, PPHS, PNI or their respective Affiliates from hiring any such person who (a) initiates discussions with Buyer, PPHS, PNI and their respective Affiliates regarding employment without any direct or indirect solicitation by or on behalf of Buyer, PPHS, PNI or their respective Affiliates, (b) responds to any public advertisement or general solicitation placed by or on behalf of Buyer, PPHS, PNI or their respective Affiliates, or (c) has been terminated by Seller or resigned from the Facilities prior to commencing employment discussion with Buyer, PPHS, PNI or their respective Affiliates.

7. CONDITIONS PRECEDENT TO OBLIGATIONS OF BUYER

The obligations of Buyer hereunder are subject to the satisfaction, on or prior to the Closing Date, of the following conditions unless waived in writing by Buyer:

- obligations and complied with all covenants. Seller shall have in all material respects performed all obligations and complied with all covenants and conditions required by this Agreement to be performed or complied with by it at or prior to the Closing Date; provided that this condition will be deemed to be satisfied unless both (a) Seller was given written notice of such failure to perform or comply and did not or could not cure such failure to perform or comply within thirty (30) days after receipt of such notice and (b) the respects in which such covenants and obligations have not been performed have had, or will likely have, a material adverse effect on the business, financial condition or results of operations of the Purchased Assets, taken as a whole.
- Opinion of Counsel. Buyer shall have received from counsel to Seller an opinion or opinions dated as of the Closing Date and addressed to Buyer, in form and substance satisfactory to Buyer, to the effect that: (i) Seller is a corporation duly organized and validly existing in good standing under the laws of the state of its organization; (ii) Seller has full power and authority to execute, deliver and perform this Agreement; (iii) that all corporate proceedings required to be taken by Seller to authorize the execution, delivery and performance of this Agreement, and to sell, convey, assign, transfer and deliver the Purchased Assets as herein contemplated, have all been taken; and (iv) this Agreement constitutes the valid and binding obligation of Seller, enforceable in accordance with its terms. Such opinion shall contain usual and customary qualifications and assumptions.
- 7.3 <u>Pre-Closing Confirmations</u>. Buyer shall have obtained documentation or other evidence reasonably satisfactory to Buyer that Buyer has:
- (a) received all required Approvals and Permits from all Governmental Entities whose approval is required (or received reasonable assurance from the applicable Governmental Entity that such Approvals and Permits shall be issued promptly following the Closing and be effective as of the Closing Date) that (i) are necessary for Buyer to (A) consummate the transactions herein contemplated or (B) operate or use the Purchased Assets or Facilities as currently operated or used by Seller, and (ii) are described in Schedule 5.2, except for any such Approvals and Permits the failure of which to obtain would not have a material adverse effect on the business, financial condition or results of operations of the Purchased Assets, taken as a whole; and
- (b) complied with all waiting periods under the HSR Act and any similar state Law, (i) if applicable, and (ii) if the parties have agreed that the consummation of the transactions contemplated hereby requires the parties to file premerger notification reports with the FTC and the Justice Department (for filings or notifications with a state agency under any applicable state antitrust Law).

7.4 <u>Action/Proceeding</u>. No court or other Governmental Entity of competent jurisdiction (other than Buyer or its Affiliates) shall have issued an order, decree or ruling or taken any other action enjoining, restraining or otherwise prohibiting the consummation of the transactions contemplated by this Agreement that remains outstanding and effective as of the Closing Date.

7.5 Title Policies and Surveys.

Buyer shall have received from an agent of its choice a commitment from Old Republic National Title Insurance Company to issue as of the Closing Date a standard ALTA 06-17-06 owner's policy of title insurance, in the customary form prescribed for use in the State of Georgia, with standard exceptions deleted and with the following endorsements: ALTA 3 Zoning, ALTA 9 Comprehensive, ALTA 17 Access, ALTA 18 Single Tax Parcel, CLTA 116 Same as Survey and CLTA 116.4 Contiguity (the "Endorsements"), for the Main Hospital Parcel (as defined in Exhibit D-1 attached hereto), together with improvements, buildings and fixtures thereon, in an amount equal to the reasonable value assigned to and agreed upon for such Main Hospital Parcel by Buyer and Seller and in the customary form prescribed for use in the State of Georgia, but with any mandatory arbitration provision deleted therefrom (the "Deletions"). The commitment shall provide for the issuance of such policy (or policies) to Buyer as of Closing and shall insure fee simple title to that portion of the Main Hospital Parcel owned by Seller with the required Endorsements and Deletions subject only to Permitted Encumbrances. Buyer shall cause the title agent and the surveyor described below to deliver the title commitment and the Survey contemplated by this Section 7.5(a) by January 13, 2011. Buyer shall, within seven (7) days from the date Buyer receives the later of (i) the title commitment for the Main Hospital Parcel or (ii) the Survey (as defined below) for the Main Hospital Parcel, notify Seller in writing of any title exceptions or defects pertaining to the Main Hospital Parcel which are not Permitted Encumbrances (as defined in clause (a) of the definition of the term "Permitted Encumbrance") and which would result in the inability of Buyer to receive a title policy sufficient to satisfy this Section 7.5. Not less than six (6) days after receiving Buyer's notice, Seller shall notify Buyer in writing of any such exceptions to title or defects in title which Seller is unable or unwilling to cause to be removed, insured against by the title insurance company or indemnified over by Seller at or prior to or at Closing. If any such exceptions or defects which Seller is unable or unwilling to cause to be removed, insured against by the title insurance company or indemnified over by Seller at or prior to or at Closing would have the result of Buyer being unable to operate the Hospital in a manner consistent with the current use thereof by Seller or its Affiliates, then Buyer shall elect, by giving written notice to Seller within two (2) business days thereafter, whether to (1) terminate this Agreement or (2) consummate the Closing with respect to all Purchased Assets and Assumed Liabilities without waiving any rights to indemnification with respect to such exceptions and defects. Seller agrees to deliver any information, consistent with a limited warranty of title as to the Main Hospital Parcel and the Permitted Encumbrances, as may be reasonably required by the title insurance company under the requirements section of the title insurance commitment or otherwise in connection with the issuance of Buyer's title insurance policy. Seller also agrees to provide an affidavit of title and/or such other information, consistent with a limited warranty of title as to the Main Hospital Parcel and the Permitted Encumbrances, as the title insurance company may reasonably require in order for the title insurance company to insure title to such Main Hospital Parcel and cause the title insurance company to delete all standard exceptions from the final title insurance policy. Buyer shall also have received from Seller a current survey of the Main Hospital Parcel (the "Survey") reasonably acceptable to Buyer from Lanier Engineering Company, a surveying company licensed under the laws of the State of Georgia, or other Georgia licensed surveyor selected by Seller and reasonably acceptable to Buyer, in standard ALTA format so as to enable the issuance of the required Endorsements and reflecting all boundaries and improvements visible on the grounds, and all easements and rights of way of record or on the grounds, and indicating whether or not the property appears on any U.S. Department of H.U.D. Flood Insurance Boundary Map and, if so, further indicating the map number and whether the property appears in the "Flood Hazard" shown on the map. Any exceptions or encroachments shown on the Survey other

than Permitted Encumbrances (as defined in clause (a) of the definition of the term "Permitted Encumbrance") to which Buyer objects in writing as set forth above shall be treated as exceptions to title or defects as above described in this Section 7.5. The Survey shall be certified to Buyer and to the title insurance company. The costs of such title policy (including without limitation, costs of the Deletions and Endorsements) and Survey shall be borne by Buyer. It is the intent of the parties that Buyer take the lead role in satisfying this condition, with the cooperation and assistance of Seller.

- With respect to all Real Property other than the Main Hospital Parcel, Buyer may elect to (i) obtain from an agent of its choice commitments from Old Republic National Title Insurance Company to issue as of the Closing Date a standard ALTA 06-17-06 owner's policy of title insurance, in the customary form prescribed for use in the State of Georgia, with standard exceptions deleted and with the Endorsements, for such Real Property (including leasehold policies with respect to those portions of the Real Property leased by Seller that are ground leases, leases of entire buildings or condominium units for which short forms or memoranda of leases have been recorded (such portions of the Leased Real Property being referred to herein as the "Insurable Leaseholds")) and (b) a Survey for such Real Property; provided, however, obtaining such title commitment or issuance of a policy of title insurance for such Real Property or obtaining a Survey for such Real Property shall not be a condition precedent to Buyer's obligation to purchase the Main Hospital Parcel or any other Real Property. Should Buyer elect to secure title commitments for such Real Property, Seller agrees to deliver any information, consistent with a limited warranty of title as to such Real Property, as may be reasonably required by the title insurance company under the requirements section of the title insurance commitment or otherwise in connection with the issuance of Buyer's title insurance policy. Seller also agrees to provide an affidavit of title and/or such other information, consistent with a limited warranty of title as to such Real Property as the title insurance company may reasonably require in order for the title insurance company to insure title to such Real Property and cause the title insurance company to delete all standard exceptions from the final title insurance policy. The costs of such title policy (including without limitation, costs of the Deletions and Endorsements) and Survey shall be borne by Buyer. It is the intent of the parties that Buyer take the lead role in satisfying this condition, with the cooperation and assistance of Seller. Buyer's failure or inability to obtain a title commitment with respect to any Real Property other than the Main Hospital Parcel as contemplated by this Section 7.5(b) shall not limit Buyer's right to seek indemnification following the Closing for a breach by Seller of any of the representations or warranties with respect to Seller's title to such Real Property. Additionally, to the extent Buyer seeks a title commitment with respect to any Real Property other than the Main Hospital Parcel but is either unable to obtain such title commitment or such title commitment includes an Encumbrance other than a Permitted Encumbrance (including any such Encumbrance not insured over by the title insurance company or indemnified by Buyer), Seller agrees that the Basket Amount under Section 11.3(b) will not apply to Buyer's right to indemnification for such Encumbrances.
- Representations and Warranties. Seller's representations and warranties contained in the following sections of this Agreement (or the following specified portions thereof) shall be true and correct in all material respects on the Closing Date: (i) the first sentence of Section 4.7(a) (but only to the extent that the accuracy of such representation is relevant to Buyer's ability to obtain all material Permits and Approvals necessary for Buyer to conduct the business and operation of the Hospital following the Closing as currently conducted by Seller or Buyer's use of the Hospital following the Closing as currently used by Seller), (ii) the first sentence of Section 4.9(a), and (iii) Section 4.12(a) (but solely with respect to the Main Hospital Parcel, as identified in Exhibit D-1). For the avoidance of doubt, except as provided above with respect to the first sentence of Section 4.7(a), the first sentence of Section 4.9(a) and Section 4.12(a), Buyer agrees that (A) Buyer shall not be excused from consummating the transactions contemplated by this Agreement based on the failure or alleged failure of any representation or warranty made by Seller to be true as a basis for not consummating the transactions contemplated by this

Agreement and (C) the sole remedy of Buyer for any failure of a representation and warranty made by Seller to be true is to be indemnified as and to the extent set forth in Section 11. In no event shall any dispute over a reduction in the Purchase Price based on the failure or alleged failure of any representation and warranty (except to the extent that any of the representations and warranties identified above in this Section 7.6 is not true and correct in all material respects) made by Seller to be true delay or postpone the Closing.

8. CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER

The obligations of Seller hereunder are subject to the satisfaction, on or prior to the Closing Date, of the following conditions unless waived in writing by Seller:

- 8.1 Compliance With Covenants. Buyer shall have in all material respects performed all obligations and complied with all covenants and conditions required by this Agreement to be performed or complied with by it at or prior to the Closing Date; provided that, except for Buyer's obligation to pay the Purchase Price at Closing, this condition will be deemed to be satisfied unless both (a) Buyer was given written notice of such failure to perform or comply and did not or could not cure such failure to perform or comply within thirty (30) days after receipt of such notice and (b) the respects in which such covenants have not been performed have had a material adverse effect on Buyer's ability to perform this Agreement or any aspect of the transactions contemplated hereby.
- Opinion of Counsel. Seller shall have received from counsel to Buyer, PPHS and PNI an opinion or opinions dated as of the Closing Date and addressed to Seller, in form and substance satisfactory to Seller, to the effect that: (i) Buyer is a hospital authority duly organized and validly existing under the laws of Georgia and PPHS and PNI are each nonprofit corporations duly organized and validly existing in good standing under the laws of Georgia; (ii) Buyer, PPHS and PNI each have full power and authority to execute, deliver and perform this Agreement; (iii) that hospital authority proceedings and corporate proceedings required to be taken by each of Buyer PPHS and PNI, respectively, to authorize the execution, delivery and performance of this Agreement, and the acquisition of the Purchased Assets by Buyer as herein contemplated, have all been taken; and (iv) this Agreement constitutes the valid and binding obligation of Buyer, PPHS and PNI, in each case enforceable in accordance with its terms. Such opinion shall contain usual and customary qualifications and assumptions.
- 8.3 <u>Action/Proceeding</u>. No court or other Governmental Entity of competent jurisdiction (other than Buyer or its Affiliates) shall have issued an order, decree or ruling or taken any other action enjoining, restraining or otherwise prohibiting the consummation of the transactions contemplated by this Agreement that remains outstanding and effective as of the Closing Date.
- 8.4 <u>Pre-Closing Confirmations</u>. Seller shall have obtained documentation or other evidence reasonably satisfactory to Seller that Seller has:
- (a) received all required Approvals and Permits from all Governmental Entities whose approval is required to consummate the transactions herein contemplated, except for any such Approvals and Permits the failure to obtain would not have a material adverse effect on the business, financial condition or results of operations of the Purchased Assets, taken as a whole; and
- (b) complied with all waiting periods under the HSR Act and any similar state Law, if applicable.

8.5 Representations and Warranties. Buyer's representations and warranties shall be true and correct in all material respects on the Closing Date.

9. TRANSITIONAL ARRANGEMENTS

9.1 PIP; Transition Patients; Government Patient Receivables; and Misdirected Payments.

- PIP Payments. The Hospital is reimbursed on an interim basis under the (a) Medicare program on a periodic interim payment ("PIP") basis. If Buyer or any of its Affiliates receives any PIP payments from the Medicare program associated with the operations of the Facilities relating solely to periods ending at or prior to the Closing, Buyer will pay Seller an amount equal to such PIP payments received by Buyer or its Affiliates. If Buyer or any of its Affiliates receives any PIP payments from the Medicare program associated with the operations of the Facilities relating to periods both prior to and after the Closing, Buyer will pay Seller an amount equal to the PIP payments actually received by Buyer or its Affiliates for such period multiplied by a fraction, the numerator of which shall be the total number of days prior to the Closing attributable to such PIP payments and the denominator of which shall be the total number of days attributable to such PIP payments. Likewise, if Seller or any of its Affiliates receive or received any PIP payments from the Medicare program associated with the operations of the Facilities relating solely to periods ending after the Closing, Seller will pay Buyer an amount equal to such PIP payments received by Seller or its Affiliates. If Seller or any of its Affiliates receive any PIP payments from the Medicare program associated with the operations of the Facilities relating to periods both prior to and after the Closing, Seller will pay Buyer an amount equal to the PIP payments actually received by Seller or its Affiliates for such period multiplied by a fraction, the numerator of which shall be the total number of days after the Closing attributable to such PIP payments and the denominator of which shall be the total number of days attributable to such PIP payments. It is the intent of the parties that Buyer and Seller are entitled to the portion of PIP payments applicable to the period of time the Facilities are owned by such party. Buyer and Seller will make all payments they are required to make under this Section 9.1(a) promptly, and in any event within five business days of receipt of the applicable PIP payments.
- Buyer or its Affiliates with respect to Transition Services provided to Transition Patients, Buyer will prepare claims (the accounts receivable resulting from such claims, the "Transition Patient Receivables") for the Transition Patients following their discharge from the Hospital. If Buyer or any of its Affiliates receives any payments on any Transition Patient Receivable, Buyer will pay Seller an amount equal to (x) the payments (including deposits, deductibles and co-payments paid, whether received by Buyer or Seller) with respect to the applicable Transition Patient Receivable multiplied by a fraction, the numerator of which shall be the total charges for the Transition Services provided to the applicable Transition Patient prior to the Closing Date, and the denominator of which shall be the sum of the total charges of the Transition Services provided to the applicable Transition Patient prior to and after the Closing Date (including charges for medicine, drugs and supplies), minus (y) any deposits, deductibles or co-payments paid by the applicable Transition Patient prior to the Closing Date and included in the Excluded Assets. Buyer will make all payments it is required to make under this Section 9.1(b) promptly, and in any event within five business days of receipt of the applicable payments.
- (c) <u>Split Bill Transition Patients</u>. Notwithstanding the provisions of <u>Section 9.1(b)</u>, to the extent a third party payor requires the parties to submit split bills for Transition Patients (i.e., a bill for the portion of the Transition Services provided prior to the Closing and a separate bill for the portion of the Transition Services provided after the Closing), (i) the provisions of <u>Section 9.1(b)</u> will not apply to billing and collection for such Transition Services, (ii) Seller will be responsible for billing and collection

for all such pre-Closing Transition Services (and the accounts receivable for such Transition Services will be Excluded Receivables), and (iii) Buyer will be responsible for billing and collection for all such post-Closing Transition Services (and the accounts receivable for such Transition Services will not be Excluded Receivables).

- (d) Government Patient Receivables. Seller hereby appoints Buyer, and Buyer agrees to act, as Seller's collection agent with respect to the Government Patient Receivables that are unpaid as of the Closing. After the Closing, Buyer will pay Seller all amounts received by Buyer or its Affiliates as payments with respect to any of the Government Patient Receivables. Buyer will make all payments that it is required to make under this Section 9.1(d) promptly, and in any event within five business days of receipt of the applicable payments.
- (e) <u>Misdirected Payments</u>. In addition to the other payments contemplated by this <u>Section 9.1</u> (or elsewhere in this Agreement), (i) if Seller or any of its Affiliates receive any amount from patients or third-party payors which relate to services rendered by the Facilities after the Closing, Seller will remit such amount to Buyer within five business days of receiving such amount; and (ii) if Buyer or any of its Affiliates receives any amount from patients or third-party payors which relate to services rendered by the Facilities prior to the Closing, Buyer will remit such amount to Seller within five days of receiving such amount.
- 9.2 Seller's Cost Reports. Seller will timely prepare all cost reports relating to Seller for periods ending on or prior to the Closing Date or required as a result of the consummation of the transactions set forth herein, including terminating cost reports for the Medicare, Medicaid and TRICARE programs (the "Seller Cost Reports"). Buyer shall fully cooperate with Seller in the preparation of Seller Cost Reports, including making available to Seller all information and records necessary for the preparation of Seller Cost Reports, providing the services of Buyer's employees as required to prepare such reports, coordinating with Seller in regards to Medicare and Medicaid exit conferences or meetings, and providing to appropriate parties (including the Provider Reimbursement Review Board), as determined to be reasonably necessary by Seller, a letter acknowledging that Seller has retained all rights to such appeals and that Buyer agrees that Seller has the right to pursue such appeals, either on Seller's behalf or, to the extent required by law, as Buyer's representative. In the event that Buyer provides personnel pursuant to this section, it shall be entitled to reimbursement from the requesting party for all reasonably incurred out-of-pocket costs and expenses, but not including internal time charges. Buyer shall forward to Seller any and all correspondence relating to Seller Cost Reports and the Excluded Receivables within three business days after receipt by Buyer. Buyer shall remit any receipts of funds relating to Seller Cost Reports or Excluded Receivables within five business days after receipt by Buyer and shall forward to Seller any demand for payments with respect to Seller Cost Reports and the Excluded Receivables within three business days after receipt by Buyer. Seller shall retain all rights to Agency Receivables and to Seller Cost Reports including any amounts receivable or payable in respect of such reports or reserves relating to such reports; provided, however, that if Buyer is unable to collect Medicare deductibles and coinsurance for amounts included in the Purchased Assets, Seller agrees to include, if necessary, such Medicare bad debts in the cost reports for periods ended on or prior to the Closing Date and to remit to Buyer an amount equal to any reimbursement received for such bad debts. Such rights shall include the right to appeal any Medicare determinations relating to Agency Receivables and Seller Cost Reports. Seller shall retain the originals of Seller Cost Reports, correspondence, work papers and other documents relating to Seller Cost Reports and the Agency Receivables. Buyer acknowledges and agrees that any accounts receivable relating to Medicare deductibles and coinsurance pertaining to services rendered prior to Closing that, based on the balance sheets of Seller in respect of the Facility for the period ending as of Closing, have been (i) recorded to Account 110-156 and/or Account 110-058 (which are general ledger accounts for patient accounts receivable that have been written off and sent to a secondary collection agency for collection) in respect of the Facility and (ii) sent to a secondary

collection agency for collection (the "Written-Off Receivables") are not included within the Assets that Buyer is acquiring hereunder. Buyer further agrees, that upon request by Seller, which request may be made at any time and from time to time after Closing, but will only be made after a secondary collection agency has determined that all or any portion of the Written-Off Receivables are not collectible and has returned same to Seller (any of such returned Written-Off Receivables are referred to herein as the "Uncollectible Receivables"), Buyer shall include the Uncollectible Receivables in the applicable cost report covering periods after the Closing Date. Buyer shall remit to Seller an amount equal to any reimbursement received for such Uncollectible Receivables within ten (10) days after receipt of any such reimbursement. Seller shall have the sole right to appeal any Medicare determinations relating to the Uncollectible Receivables. Buyer herby appoints Seller as the authorized representative to pursue appeal rights on the Uncollectible Receivables.

9.3 Employees.

- As of the Closing Date, Seller shall terminate or cause its Affiliates to terminate all employees of Seller or its Affiliates at the Facilities (other than any member of Senior Management that has elected to remain with Seller), and PNI shall offer employment to all of the employees of Seller and its Affiliates at the Facilities (other than members of Senior Management) as of the Closing Date as employees-at-will (except as provided in Section 9.3(b)) at compensation and benefit levels comparable to those paid by Affiliates of PNI to their employees who work in similar positions and who have similar seniority as the applicable employee of Seller, provided that such employees (i) meet PNI's employment qualifications, which shall not differ from those qualifications in effect at those facilities operated by Buyer, PPHS or their respective Affiliates, (ii) meet all federal and State of Georgia laws governing the hiring of employees, and (iii) have not been previously discharged by PNI or any of its Affiliates. Not later than January 4, 2011, Seller shall provide PNI the names of Seller's (and its Affiliates') employees employed at the Hospital or the other Facilities. Not later than January 14, 2011, PNI shall provide Seller the names of any employee to whom PNI will not offer employment. PNI shall not be required to hire any member of Senior Management. In the event that PNI does desire to hire any member of Senior Management, an offer of employment shall be extended to such individual not later than January 14, 2011. Except with respect to Senior Management, until January 17, 2011, Seller shall not, directly or indirectly, (A) make an offer to any such employee to remain employed, or become re-employed, by Seller after the Closing Date or (B) transfer any such employee to a position at any facility owned or operated by Seller or any of its Affiliates. Except for any employee of Seller who fails to meet the documentation and verification obligations required by Law, PNI shall be responsible for, and hereby assumes, any and all liabilities and obligations resulting from (x) the termination of any such employee as set forth in this Section 9.3(a) (other than a member of Senior Management) and (y) any claims by any employee of Seller who was not offered employment with PNI based on allegations that the failure to be offered employment violated applicable Law, including labor Laws.
- any of its Affiliates who accepts employment with PNI as of the Closing Date. All Employees will be retained as employees-at-will (except to the extent that such Employees are parties to contracts providing for other employment terms, in which case such Employees shall be retained in accordance with the terms of such contracts) and PNI shall provide such Employees with the same customary employee benefits as PNI or its Affiliates provide to their employees who work in similar positions and who have similar seniority as the applicable Employee under benefit plans and arrangements that are maintained by PNI or its Affiliates ("Palmyra Medical Center Plans"). The terms of all such Employees' employment with PNI shall (i) be in accordance with PNI's usual and customary practices for its own employees, which shall not differ from those practices in effect at those facilities operated by Buyer, PPHS or their Affiliates, and (ii) include recognition of the existing seniority of all such Employees in determining eligibility for benefits under the Palmyra Medical Center Plans, except as provided below. PNI shall provide credit for

eligibility, benefit accrual and vesting purposes for all such Employees' periods of service with Seller (or any Affiliate of Seller) as provided under the ERISA and non-ERISA plans of Seller and its Affiliates for purposes of any Palmyra Hospital Plan, provided that, with respect to Palmyra Medical Center Plans that are active defined benefit pension plans, the existing seniority of such Employees shall only be recognized for eligibility and vesting purposes and not for benefit accrual purposes; and provided further that such Employees shall not participate in any Palmyra Medical Center Plan that is an inactive defined benefit plan. Any future Palmyra Medical Center Plans that provide for benefit and vesting service to employees of PNI or its Affiliates from their original date of hire shall (to the extent permitted by applicable Law) include all vesting and benefit service credit as would be included by recognizing such Employees' original date of hire as recognized by Seller or one of its Affiliates. The service credited under Palmyra Medical Center Plans that are welfare and other benefit plans will include all service credited under the welfare and other benefit plans of Seller and its Affiliates, respectively. Deductibles and out-of-pocket limits met and applied under Seller's medical plans during the year in which the Closing takes place shall be applied to any Palmyra Medical Center Plans that are medical plans following the Closing Date. Participation shall begin as of the Closing Date for participating Employees (and eligible dependents) and for all other Employees who, given their Seller service, have met the age and service requirements for participation under the respective Palmyra Medical Center Plans. Additionally, all Accrued PTO Banks for employees (to the extent the liability for such Accrued PTO is reflected in the Accrued PTO Amount) and all Accrued EIB Banks for employees will be assumed by PNI upon the Closing Date. PNI will, to the extent lawful, waive all pre-existing condition limitations in the applicable Palmyra Medical Center Plans that are employee welfare benefit plans for any Employees and their eligible dependents. PNI shall be liable and responsible for any notification required under the Worker Adjustment and Retraining Notification Act, 29 U.S.C. §2101 et. seq. (or under any similar state or local Law) (collectively, the "WARN Act"). In the event that the chief executive officer, chief financial officer, chief operating officer or chief nursing officer of any Facility is employed by PNI and terminated by PNI within twelve months following the Closing Date, PNI agrees to pay such individual severance benefits not less than the severance benefits to which such individual would have been entitled under the severance policies, practices, programs or plan of Seller in effect as of the date hereof, as previously disclosed to PNI.

- PNI shall through the applicable Palmyra Medical Center Plans assume responsibility for COBRA coverage for any applicable employee of Seller and their Affiliates identified as a current participant (and any eligible dependent thereof) who is eligible to receive continuation coverage (within the meaning of Code Section 4980B and Part 6 of Subtitle B of Title I of ERISA) and who meets each of the following criteria: (i) such employee is employed at the Facilities on the date hereof (or is hired at the Facilities after the date hereof); (ii) such employee's employment is terminated at the Closing as contemplated by Section 9.3(a); and (iii) such employee either is offered employment as of the Closing Date by PNI or is identified by PNI to Seller not later than January 14, 2011, as an employee to whom PNI will not offer employment, provided, however, that PNI shall not assume responsibility for COBRA coverage for any employee of Seller who PNI does not employ because such employee fails to meet the documentation and verification obligations required by Law. Immediately following the Closing Date and as a result of the transactions contemplated by this Agreement, Seller shall cease to offer COBRA benefits for any applicable group health plan to former employees (and their dependents) who are employed by PNI at any Facility following the Closing Date or to any of the other employees (and their dependants) who meet each of the criteria set forth in clauses (i), (ii) and (iii) of this Section 9.3(c). Seller will thereby be released of COBRA responsibility and liability for such employees (and their dependents).
- (d) As of the Closing Date, Seller will, at its expense or at the expense of the applicable Plan, (i) terminate all Plans, if any, relating solely to employees at any Facility, (ii) terminate the participation of all employees employed at any Facility from all other Plans, (iii) take such actions as

are necessary to make, or cause such Plans to make, timely appropriate distributions to such employees to the extent required or permitted by, and in accordance with, such Plans and applicable Law, as determined by Seller and/or its counsel, and (iv) comply with all applicable Laws in connection with the foregoing. Seller shall indemnify and hold harmless Buyer and PNI from and against any and all liabilities and obligations whatsoever with respect to the Plans or the acts or omissions of Seller under this Section 9.3.

10. <u>ADDITIONAL AGREEMENTS</u>

10.1 Break-Up Fee; Rescission Fee.

(a) <u>Break-Up Fee.</u>

- (i) PPHS shall pay Seller thirty-five million dollars (\$35,000,000.00) (the "Break-Up Fee") in the event that either Buyer or Seller terminates this Agreement prior to the Closing in accordance with Section 10.3(a)(ii), (iii), (iv) or (v); provided, however, that if Buyer terminates this Agreement in accordance with Section 10.3(a)(ii) or (iii) because one or more of the conditions set forth in Section 7.1, 7.2, 7.5 or 7.6 are not satisfied as of June 1, 2011 (or, if the optional termination date provided for in Section 10.3(a)(ii) shall have been extended to October 1, 2011 in accordance with Section 10.3(a)(ii), as of October 1, 2011), PPHS shall not be required to pay Seller the Break-Up Fee pursuant to this Section 10.1(a).
- (ii) PPHS shall make any payment required to be paid pursuant to this Section 10.1(a) by wire transfer of immediately available funds no later than two business days following the termination of this Agreement that results in PPHS' obligation to make such payment.
- (b) <u>Rescission Fee.</u> If the purchase and sale of the Purchased Assets shall have been consummated and any court or other Governmental Authority of competent jurisdiction (other than Buyer or its Affiliates) shall have issued an order, decree or ruling or taken any other action rescinding such transaction and requiring the parties to cause all or substantially all of the Facilities to be re-conveyed to Seller and such order, decree, ruling or other action shall have become final and non-appealable, PPHS shall pay Buyer thirty-five million dollars (\$35,000,000.00) (the "Rescission Fee") at the closing of such rescission transaction (or, if PPHS fails or refuses to pay such Rescission Fee, Seller shall be entitled to retain thirty-five million dollars (\$35,000,000.00) of the amount that Seller otherwise would be required to pay to Buyer at the closing of such rescission transaction).

(c) Other Provisions.

- (i) The parties expressly acknowledge and mutually agree that the Break-Up Fee and the Rescission Fee are based on a reasonable determination of damages that would be incurred by Seller in the event that this Agreement is terminated under the circumstances contemplated by Section 10.1(a)(i) or the parties are required to rescind the transactions contemplated hereby as contemplated by Section 10.1(b). The parties also expressly acknowledge that this Section 10.1 is a specifically bargained for provision of this Agreement and Buyer acknowledges that Seller would not have entered into this Agreement without the provisions contained in this Section 10.1.
- (ii) In the event that PPHS pays to Seller the Break-Up Fee or the Rescission Fee (or Seller retains the Rescission Fee pursuant to Section 10.1(b)) such payment of the Break-Up Fee or the Rescission Fee (or such retention of the Rescission Fee, as applicable) shall be the exclusive remedy available to Seller against Buyer, PPHS, PNI or any of their Affiliates under the applicable circumstances described in Section 10.1(a) or Section 10.1(b). Notwithstanding the foregoing, nothing in

4016376.6

this <u>Section 10.1</u> is intended to limit Buyer's obligations under (A) <u>Section 12.7</u> to pay all fees and expenses incurred by Seller in relation to any investigation or challenge of the transactions contemplated hereby described in such <u>Section 12.7</u> (including any such investigation that results in PPHS' obligation to pay the Break-Up Fee or the Reseission Fee or Seller's right to retain the Reseission Fee), (B) <u>Section 11.2(f)</u> with respect to indemnification for Indemnifiable Losses incurred by any Seller Indemnified Party consisting of fees and expenses relating to any investigation, request for information, subpoena, claim, demand, action, suit or other action as contemplated by <u>Section 11.2(f)</u> (whether initiated by the FTC, the Justice Department, the Attorney General of the State of Georgia or any other Person) or consisting of fines, penalties, judgments or other amounts that a Seller Indemnified Party is obligated to pay as a result of any such matter, or (C) under <u>Section 11.2(b)</u> with respect to a breach by Buyer of its obligations under <u>Section 6.6(a)(x)</u> to defend and oppose any claim, action or proceeding which results in Buyer's obligation to pay the Break-Up Fee or the Rescission Fee.

(iii) In the event of any legal action brought by the Justice Department, the FTC, the Attorney General of the State of Georgia or any other Person seeking to restrain or prohibit the consummation of the transactions contemplated hereby, impose material damages or penalties in connection therewith or rescind the transaction following the Closing under (in the case of any such proceedings) state or federal antitrust Laws, PPHS shall have the sole right to control said litigation in the manner set forth in Section 11.4 of this Agreement.

10.2 Allocation of Purchase Price.

- Assets in the manner required by Section 1060 of the Code and as set forth in Schedule 10.2. In the event the parties are unable to agree upon the form and substance of Schedule 10.2 within 60 days following the Closing, then any disputed matters will be finally and conclusively determined by an independent certified accounting firm or independent certified appraisal firm (the "Allocation Arbiter"), which firm shall be mutually agreed by Buyer and Seller. Promptly, but not later than 15 days after its acceptance of appointment hereunder, the Allocation Arbiter will determine (based solely on presentations by Seller and Buyer and not by independent review) only those matters in dispute and will render a written report as to the disputed matters and the resulting allocation of the Purchase Price, which report shall be conclusive and binding upon the parties. Such Allocation Arbiter's fees and expenses shall be borne equally by the parties.
- (b) Seller will provide to Buyer copies of Internal Revenue Service Form 8594 and any required exhibits thereto, consistent with the allocations in <u>Section 10.2(a)</u>. In this regard, the parties agree that, to the extent required, all tax returns or other tax information they may file or cause to be filed with any Governmental Entity shall be prepared and filed consistently with such allocation.

10.3 Termination Prior to Closing.

(a) Notwithstanding anything in this Agreement to the contrary, this Agreement and the transactions contemplated by this Agreement may not be terminated, except prior to the Closing as follows: (i) by mutual consent in writing of Buyer and Seller; (ii) by Seller or Buyer at any time after June 1, 2011 if the Closing shall not have occurred by such date; provided, that the right to terminate this Agreement under this Section 10.3(a)(ii) shall not be available to any party whose failure to fulfill any obligation under this Agreement or whose failure to use commercially reasonable efforts to cause the satisfaction of the conditions under Sections 7 and 8 prior to the date of such intended termination has been the cause of, or resulted in, the failure of the Closing to occur by such date; and provided further, that if a court or other Governmental Entity of competent jurisdiction (other than Buyer or its Affiliates) shall have issued an order, decree or ruling or taken any other action enjoining, restraining or otherwise

prohibiting the consummation of the transactions contemplated by this Agreement that remains outstanding and effective as June 1, 2011, the date provided for in this Section 10.3(a)(ii) shall be extended to October 1, 2011; (iii) by Buyer by written notice to Seller if any event occurs or condition exists which causes Seller to be unable to satisfy one or more obligations of Seller that is a condition to the obligation of Buyer to consummate the transactions contemplated by this Agreement as set forth in Section 7; (iv) by Seller by written notice to Buyer if any event occurs or condition exists which causes Buyer to be unable to satisfy one or more obligations of Buyer that is a condition to the obligation of Seller to consummate the transactions contemplated by this Agreement as set forth in Section 8; or (v) by Buyer or Seller in the event that any court or other Governmental Authority of competent jurisdiction (other than Buyer or its Affiliates) shall have issued an order, decree or ruling or taken any other action restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such order, decree, ruling or other action shall have become final and non-appealable.

- (b) In the event that this Agreement shall be terminated pursuant to Section 10.3(a) and except as otherwise provided in this Section 10.3(b), all further obligations of the parties under this Agreement shall terminate without further liability of any party to another; provided that the obligations of PPHS contained in Section 10.1, and the parties in this Section 10.3(b), Section 6.9, Section 11.2(f) and Article 12 shall survive any such termination. Except as set forth in Section 10.1 above, a termination under Section 10.3(a) shall not relieve any party of any liability for a breach of, or for any misrepresentation under this Agreement, or be deemed to constitute a waiver of any available remedy (including specific performance if available) for any such breach or misrepresentation.
- Post-Closing Access to Information. Buyer and Seller acknowledge that, subsequent to the Closing, Buyer and Seller may each need access to information, documents or computer data in the control or possession of the other, and Seller may need access to the Purchased Assets or the Facilities for purposes of concluding the transactions contemplated herein and for audits, investigations, compliance with governmental requirements, regulations and requests, and the prosecution or defense of third party claims. Accordingly, Buyer agrees that, at the sole cost and expense of Seller, it will make available to Seller and its agents, independent auditors and/or Governmental Entities such documents and information as may be available relating to the Purchased Assets and Facilities in respect of periods prior to Closing and will permit Seller to make copies of such documents and information. Seller agrees that, at the sole cost and expense of Buyer, Seller will make available to Buyer and its agents, independent auditors and/or Governmental Entities such documents and information as may be in the possession of Seller relating to the Purchased Assets and Facilities in respect of periods prior to the Closing and will permit Buyer to make copies of such documents and information. The parties agree that Buyer and its agents and independent auditors may review Seller's cost report work papers with respect to the pre-Closing operations of the Facilities, but that neither Buyer nor such agents or auditors may remove copies of such work papers.
- 10.5 Preservation and Access to Records After the Closing. After the Closing, Buyer shall for that period of time required by Law keep and preserve in their original form or in an electronic format all medical and other records of the Facilities existing as of the Closing. For purposes of this Agreement, the term "records" includes all documents, electronic data and other compilations of information in any form. Buyer acknowledges that as a result of entering into this Agreement and managing the Facilities it and its Affiliates will gain access to patient and other information which is subject to rules and regulations regarding confidentiality. Buyer shall abide by any such rules and regulations relating to the confidential information that it acquires. Buyer shall maintain the patient records held at each Facility or delivered to Buyer at Closing at the Facilities after Closing in accordance with applicable Law (including, if applicable, Section 1861(v)(i)(I) of the Social Security Act (42 U.S.C. § 1395(V)(1)(i)), and requirements of relevant insurance carriers, all in a manner consistent with the maintenance of patient records generated at the Facilities after Closing. Upon reasonable notice, during normal business hours and upon Buyer's

receipt of appropriate consents and authorizations, Buyer shall afford to representatives of Seller, including its counsel and accountants, full and complete access to, and the right to make copies of, the records transferred to Buyer at the Closing (including, without limitation, access to patient records in respect of patients treated by Affiliates of Seller at the Facilities). In addition, Seller shall be entitled to remove from the Facilities any such patient records, but only for purposes of pending litigation involving a patient to whom such records refer, as certified in writing prior to removal by counsel retained by Seller in connection with such litigation. Any patient records so removed from the Facilities shall be promptly returned to Buyer following its use by Seller. Pending return, Seller will provide a copy of the removed records to Buyer.

- 10.6 <u>Capital Expenditures</u>. This Agreement shall not be deemed to be an acquisition or obligation of a capital expenditure or of funds within the meaning of the certificate of need statute of any state, until the appropriate Governmental Entities shall have granted a certificate of need or the appropriate approval or ruled that no certificate of need or other approval is required.
- Reproduction of Documents. This Agreement and all documents relating hereto, including (a) consents, waivers and modifications which may hereafter be executed, (b) the documents delivered at the Closing, and (c) financial statements, certificates and other information previously or hereafter furnished to Seller or to Buyer, may, subject to the provisions of Section 12.8, be reproduced by Seller and by Buyer by any photographic, photostatic, microfilm, micro-card, miniature photographic or other similar process and, unless required by Law or this Agreement otherwise, Seller and Buyer may destroy any original documents so reproduced. Seller and Buyer agree and stipulate that any such reproduction shall be admissible in evidence as the original itself in any judicial, arbitral or administrative proceeding (whether or not the original is in existence and whether or not such reproduction was made by Seller or Buyer in the regular course of business) and that any enlargement, facsimile or further reproduction of such reproduction shall likewise be admissible in evidence.
- 10.8 Cooperation on Tax Matters. Following the Closing, the parties shall cooperate fully with each other and shall make available to the other, as reasonably requested and at the expense of the requesting party, and to any taxing authority, all information, records or documents relating to tax liabilities or potential tax liabilities of Seller for all periods on or prior to the Closing and any information which may be relevant to determining the amount payable under this Agreement, and shall preserve all such information, records and documents (to the extent a part of the Purchased Assets delivered to Buyer at Closing) at least until the expiration of any applicable statute of limitations or extensions thereof.
- 10.9 <u>Litigation Support</u>. Following the Closing, Buyer shall make available to Seller, at no charge to Seller, Buyer's personnel, to the extent reasonably required by Seller in connection with any litigation, investigation or other judicial or administrative proceedings attributable to the ownership or operation of the Facility on or prior to Closing.
- 10.10 HCA Post-Closing Services to be Provided Buyer. If requested by Buyer, the parties shall negotiate in good faith the terms of applicable transition services agreements covering any of the following categories of services as may be reasonably required by Buyer to operate the Purchased Assets after the Closing: computer and data processing, supply chain services, revenue cycle services, and group purchasing. The parties agree that (i) the terms of such transition services agreements shall be only for such length of time as may be reasonably required to transfer such functions to Buyer and will include optional termination provisions that permit Buyer to terminate such agreements as and when Buyer no longer needs the applicable services (upon not less than 90 days prior written notice to Seller), (ii) the charges for services rendered by Seller or its Affiliates under such transition services agreements will be substantially similar to those historically charged by Seller (or its applicable Affiliate) in connection with the provision of such services to Seller Affiliates or to third parties in divestiture situations, (iii) the costs

of establishing the systems, adopting the systems for Buyer's purposes and unwinding the systems upon termination (including one-time set up charges substantially similar to those historically charged by Seller (or its applicable Affiliate) to third parties in divestiture situations) will be borne by Buyer, (iv) except as contemplated by this Section 10.10 (including as contemplated in this Section 10.10 regarding the term and termination provisions of such agreements), the form and substance of such transition services agreements (including the provisions limiting the liability of Seller or its applicable Affiliate under such agreements) will be substantially similar to the forms of the applicable transition services agreements that Seller (or its applicable Affiliates) are currently utilizing with third parties in divestiture situations, copies of which have been made available to Buyer prior to the execution of this Agreement. The parties shall endeavor to finalize any such transition services agreements by January 15, 2011.

- 10.11 No-Shop. Except for the sale of Inventory and other assets in the ordinary course, Seller agrees that, from the date hereof through the date of consummation or earlier termination of this Agreement, it shall not, and shall not permit any of its subsidiaries or employees or any investment banker, attorney, adviser or other representative of Seller to, directly or indirectly (i) offer to sell the Purchased Assets (or any material portion thereof) or any ownership interest in any entity owning any of the Purchased Assets to any third party other than Buyer, (ii) solicit offers to buy all or any material portion of the Purchased Assets or any ownership interest in any entity owning any of the Purchased Assets, or (iii) enter into any agreement with any party other than Buyer with respect to the sale or other disposition of the Purchased Assets (or any material portion thereof) or the sale, merger, consolidation or similar transaction with respect to any ownership interest in any entity owning any of the Purchased Assets.
- 10.12 <u>Post-Closing Operations</u>. Buyer acknowledges that it is an experienced and knowledgeable owner and operator of facilities and assets similar to the Purchased Assets and will rely on its own expertise and the expertise of PNI in operating the Purchased Assets from and after the Closing. Buyer covenants for the benefit of Seller to operate the Purchased Assets in material compliance with all applicable Laws, including Laws relating to the regulation of the Facilities and operation of the Purchased Assets and all Environmental Laws, from and after the Closing.
- 10.13 Maintenance of Programs. Buyer represents and agrees that, as of Closing, it has or will implement and maintain an effective program to prevent and detect violations of legal requirements applicable to the delivery of goods and services in connection with any health care benefits and that such a program will comply with the provisions of the U.S. Sentencing Guidelines relating to corporate compliance programs and will be mindful of any applicable guidance issued by the U.S. Department of Health and Human Services. Buyer agrees that it will maintain such program for no less than five years following the date of Closing.
- 10.14 <u>Provision of Certain Employees</u>. After the Closing Date, Buyer agrees to make available to Seller such of Buyer's employees who were Seller's employees prior to the Closing Date as Seller shall reasonably request for the purpose of closing Seller's books, preparing the Closing Balance Sheet and otherwise complying with <u>Article 2</u> hereof as well as preparing and providing information in relation to Seller Cost Reports pursuant to <u>Section 9.2</u> and other similar transitional matters. Seller agrees to reimburse Buyer for direct costs associated with such provision of Buyer's employees to Seller.

10.15 Non-Competition.

(a) This <u>Section 10.15</u> is a specifically bargained for provision of this Agreement and Seller acknowledges that Buyer would not have entered into this Agreement without the provisions contained in this <u>Section 10.15</u>.

- (b) Except as permitted in this <u>Section 10.15</u>, during the period commencing on the Closing Date and ending on the third anniversary of the Closing Date, Seller agrees that it shall not, and shall cause each of its Affiliates not to, acquire, construct, lease, own or be a controlling shareholder, controlling partner, controlling member or controlling equity holder of, exercise management control over, provide consulting services for, or acquire or maintain a controlling interest in, a Competing Business located in the Territory.
- (c) If HCA is acquired by another Person that owns or operates five or more general acute care hospitals (whether such acquisition is by stock purchase, merger, purchase of all or substantially all of its assets or otherwise), the prohibition included in Section 10.15(b) shall continue to apply to Seller and its Affiliates following such acquisition, but shall not apply to any Competing Business located within the Territory owned, leased, operated, managed or being built by such Person at the time it acquires HCA (or to the Person's ownership, leasing, operation, management or continued construction of such Competing Business) or to any expansion, addition, alteration or improvement of or to such Competing Business. If HCA acquires, directly or indirectly, another Person (by stock purchase, merger, purchase of all or substantially all of its assets or otherwise) that owns or operates five or more general acute care hospitals, and such Person owns, leases, operates, manages or is building a Competing Business within the Territory at the time of such acquisition, this Section 10.15 shall not prohibit HCA's acquisition of such Competing Business or its subsequent ownership, leasing, operation or management or such Competing Business of its completion of the construction of such Competing Business, or to any expansion, addition, alteration or improvement of or to such Competing Business.
- (d) The covenants in Section 10.15(b) will not apply to (i) any Persons (including Seller) or health care assets, facilities or operations that cease to be owned or controlled, directly or indirectly, by HCA, including any Person, health care assets, facilities or operations that may be divested or spun-off by HCA or its Affiliates, or (ii) any services (including consulting services; information systems services; billing, collection or other revenue cycles services; purchasing or distribution services (including participation in group purchasing organizations), but excluding management services) provided by or on behalf of any Affiliate of HCA to (A) any such Person or assets or facilities referred to in clause (i) above or (B) any Persons in which HCA or its Affiliates have, directly or indirectly, any minority ownership interest.
- (e) For purposes of this <u>Section 10.15</u>, (i) the term "<u>Competing Business</u>" means the business of owning, operating, leasing or managing general or tertiary acute care hospitals, including any general or tertiary acute care hospital currently within the Territory or hereafter constructed within the Territory; (ii) the term "<u>control</u>" has the meaning set forth in the definition of "Affiliate;" (iii) the term "<u>controlling</u>" means having "<u>control</u>" over the applicable Person; and (iv) the term "<u>Territory</u>" means the territory within 20 miles of the Hospital.

10.16 Settlement and Release.

(a) Effective upon the Closing, Seller, on behalf of itself and its Affiliates, HCA shareholders, directors, officers, agents, representatives, successors, assigns, heirs, administrators, employees and attorneys, hereby accepts the terms and conditions of this Agreement in full settlement, satisfaction and accord of any and all claims, rights, liens or causes of action it has or could have, whether directly, derivatively, or otherwise known or unknown, against Buyer, PPHS, or their Affiliates, successors, assigns or any officer, director, employee, agent, attorney or representative of Buyer, PPHS or their Affiliates which arise out of, relate to, or otherwise concern the claims in (i) Palmyra Park Hospital, Inc., d/b/a Palmyra Medical Center v. Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Health System, Inc., Civil Action File No. 1:08-CV-00102-WLS, pending in the United States District Court for the Middle District of Georgia, Albany Division (the "Antitrust Claim"); (ii) the decision rendered by the

Superior Court of Sumter County, Georgia, in <u>Phoebe Sumter Medical Center v. Ga. Department of Community Health and Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center, Case Number 10CV-30-P, on appeal in the Court of Appeals for the State of Georgia, regarding CON Application # 2008-081; and (iii) the decision rendered by the Superior Court of Dougherty County, Georgia, in <u>Phoebe Putney Memorial Hospital v. Ga. Department of Community Health and Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center, Case Number 10CV-097-1, on appeal in the Court of Appeals for the State of Georgia, regarding CON Application # 2008-081.</u></u>

- (b) Seller and its Affiliates, HCA shareholders, directors, officers, agents, representatives, successors, assigns, heirs, administrators, employees and attorneys, shall not reassert the Antitrust Claim in any adjudicative body unless the conditions precedent to PPHS' payment of the Rescission Fee described at Section 10.1(b) above have occurred.
- (c) Upon the occurrence of the conditions precedent to PPHS' payment of the Rescission Fee described at Section 10.(b) above, Seller and its Affiliates, HCA shareholders, directors, officers, agents, representatives, successors, assigns, heirs, administrators, employees and attorneys shall be entitled to reassert the Antitrust Claim in any manner permitted by the courts in which the Antitrust Claim was originally filed. Buyer, PPHS, and their Affiliates, successors, assigns or any officer, director, employee, agent, attorney or representative of Buyer, PPHS or their Affiliates shall not oppose such reassertion of the Antitrust Claim based on the Antitrust Claim's initial dismissal without prejudice or reassertion. By this Section 10.16(c), Buyer, PPHS, and their Affiliates, successors, assigns or any officer, director, employee, agent, attorney and representative of Buyer, PPHS or their Affiliates waive no objection, defense or immunity to the Antitrust Claim they may have existing as of any time prior to Closing and may reassert same should the Antitrust Claim be reasserted.
- (d) The parties agree that they shall preserve all evidence relevant to the Antitrust Claim until the later of: (i) five years from the Closing or (ii) the entry of a final, non-appealable judgment in any action taken by the FTC, the Justice Department, the Attorney General of the State of Georgia, or any other Person in connection with the transactions contemplated hereby under state or federal antitrust Law that is pending on the fifth anniversary of the Closing Date.

11. <u>INDEMNIFICATION</u>

- 11.1 <u>Indemnification by Seller</u>. From and after the Closing Date, Seller shall indemnify and hold harmless Buyer, its subsidiaries and Affiliates, and its and their respective officers, directors, principals, attorneys, agents, employees or other representatives (collectively, "<u>Buyer Indemnified Parties</u>") from and against any and all Indemnifiable Losses that such Buyer Indemnified Party incurs as a result of, or with respect to:
- (a) any inaccuracy in any of the representations or warranties made by Seller in this Agreement;
- (b) any material breach or nonfulfillment of any covenants or other agreements made by Seller in this Agreement;
 - (c) any of the Excluded Liabilities; and
- (d) except as contemplated in clause (d) of the definition of the term "ADA Liabilities," any of the claims, actions, suits, audits, compliance reports or information requests, proceedings, investigations, judgments, orders or decrees listed in <u>Schedule 4.17(a)</u> or <u>Schedule 4.17(c)</u>, or any breach of the representations provided in <u>Section 4.17(a)</u> or <u>Section 4.17(c)</u>.

Buyer acknowledges that any representations and warranties made by Seller as of the Closing Date that address the operation of the Purchased Assets by Seller shall be deemed to be made immediately prior to the sale of the Purchased Assets (except to the extent any of such representations and warranties speaks as of a specific date, in which case such representation and warranty shall be made as of the date specified).

- Indemnification by Buyer. From and after the Closing Date, Buyer shall indemnify and hold harmless Seller, its subsidiaries and Affiliates, and its and their respective officers, directors, principals, attorneys, agents, employees or other representatives (collectively, "Seller Indemnified Parties") from and against any and all Indemnifiable Losses that such Seller Indemnified Party incurs as a result of, or with respect to:
- (a) any inaccuracy in any of the representations or warranties made by Buyer in this Agreement;
- (b) any material breach or non-fulfillment of any of the covenants or other agreements made by Buyer in this Agreement;
 - (c) any of the Assumed Liabilities;
 - (d) the use, operation or ownership of any of the Purchased Assets after the Closing;
- (e) any notification required under the WARN Act in connection with the transactions contemplated hereby, including any such losses, liabilities, damages, judgments, claims, costs and expenses that such Seller Indemnified Party incurs as a result of PNI's failure to employ a sufficient number of employees at each Facility as of the Closing Date and to continue to employ such persons for a sufficient period following the Closing Date so as not to constitute a "plant closing" or "mass lay-off" under the WARN Act at any of the Facilities in the absence of pre-Closing notice to the employees of the Facilities under the WARN Act; and
- (f) any pre- or post-Closing investigation, request for information, subpoena or other action taken by the FTC, the Justice Department, the Attorney General of the State of Georgia, or any other state or federal agency in connection with the transactions contemplated hereby under state or federal antitrust Laws; or any claim, demand, action or suit by any Person claiming injury in its trade or business under state or federal antitrust Laws in connection with the transaction contemplated by this Agreement.

11.3 Limitation.

- (a) The liability of Seller and Buyer for indemnification under <u>Section 11.1(a)</u> or <u>11.2(a)</u>, respectively, shall be limited to an amount equal to the Purchase Price.
- (b) Neither Seller nor Buyer shall be required to make any indemnification payment pursuant to Section 11.1(a) or 11.2(a), respectively, unless the aggregate of all amounts for which indemnity would be payable by such party exceeds \$3,000,000.00 (the "Basket Amount"), and in such event, such party shall be responsible for only the amount in excess of the Basket Amount, provided, however, that the limitation on the liability of (i) Seller provided for in this paragraph shall not apply to any Excluded Liability and (ii) Buyer provided for in this paragraph shall not apply to any Assumed Liability.

- (c) The amount of any Indemnifiable Losses that may be imposed on or otherwise incurred or suffered by Buyer ("Buyer Losses") shall be reduced or reimbursed, as the case may be, by any amount received by Buyer Indemnified Parties with respect thereto under any insurance coverage or for any other party alleged to be responsible therefor. Buyer Indemnified Parties shall use reasonable efforts to collect any amounts available under such insurance coverage and from such other party alleged to have responsibility. If a Buyer Indemnified Party receives an amount under insurance coverage or from such other party with respect to Buyer Losses at any time subsequent to any indemnification provided by Seller pursuant to Section 11.1, then such Buyer Indemnified Party shall promptly reimburse Seller for any payment made or expense incurred by Seller in connection with providing such indemnification up to such amount received by Buyer Indemnified Party.
- (d) Any indemnification payments required to be made hereunder with respect to any matter shall be reduced by the amount of any benefits that can be proven to have been received by the indemnified party as a result of such matter.

11.4 Notice and Control of Litigation.

- (a) If any claim or liability is asserted in writing against a Person entitled to indemnification under this Section 11 (the "Indemnified Party") which would give rise to a claim under this Section 11 or Section 10.1, the Indemnified Party shall notify the Person giving the indemnity ("Indemnifying Party") in writing of the same within ten (10) business days of receipt of such written assertion of a claim or liability, provided, however, that the failure to provide such notice as so indicated shall not affect the Indemnifying Party's obligation to indemnify and the Indemnifying Party shall have no remedy by reason of such failure except to the extent of any actual prejudice resulting from such delay. The Indemnifying Party shall have the right to defend any such claim, select the counsel and control the defense, settlement and prosecution of any litigation. If the Indemnifying Party, within ten (10) business days after notice of such claim, fails to provide the Indemnified Party with notice of its intent to defend such claim, or thereafter fails to take action to defend such claim, the Indemnified Party will (upon further notice to the Indemnifying Party) have the right to undertake the defense, compromise or settlement of such claim on behalf of and for the account and risk of the Indemnifying Party; provided however that such claim shall not be compromised or settled without the consent of the Indemnifying Party, which consent shall not be unreasonably withheld.
- (b) The Indemnified Party shall cooperate in all reasonable respects with the Indemnifying Party in the investigation, trial and defense of any lawsuit or action that may be subject to this <u>Section 11</u> and any appeal arising therefrom; provided however that the Indemnified Party may, at its own cost, participate in the investigation, trial and defense of such lawsuit or action any appeal arising therefrom. The parties shall cooperate with each other in any notifications to insurers.
- Agreement (other than those set forth in Sections 4.1 and 5.1, which shall survive indefinitely, and those set forth in Sections 4.17 and 4.18, which shall survive until the expiration of the statute of limitations) shall survive the Closing until twenty-one (21) months after the Closing (the "Survival Period") but shall not survive any termination of this Agreement. The parties intend to shorten the statute of limitations and agree that no claims or causes of action may be brought against Seller or Buyer based upon, directly or indirectly, any of the representations or warranties contained in this Agreement or any agreements contained in Section 6 after the Survival Period or any termination of this Agreement (except as provided in Section 10.3(b)). For the avoidance of doubt, this Section 11.5 shall not affect any rights to bring claims after the Survival Period based on (i) any covenant or agreement of the parties which contemplates performance after the Survival Period, (ii) the obligations of Seller under Section 11.1(c) or 11.1(d), or (iii) the obligations of Buyer under Section 11.2(c), 11.2(d), 11.2(e) or 11.2(f).

- 11.6 <u>Materiality</u>; <u>De Minimis Claims</u>. For purposes of the parties' indemnification obligations under this <u>Article 11</u>, (a) the words "material," "materiality," and "material adverse effect" (and similar words or phrases) which modify any representation, warranty, covenant or agreement shall be disregarded for purposes of determining the amount of indemnifiable losses, liabilities, damages, costs and expenses resulting from a breach of the applicable representation, warranty, covenant or agreement, but will not be disregarded for purposes of determining whether the applicable representation, warranty, covenant or agreement has been breached, and (b) any individual claim or matter involving losses, liabilities, damages, judgments, claims, costs and expenses of less than \$20,000 based on <u>Section 11.1(a)</u> and <u>Section 11.2(a)</u>, respectively, shall be disregarded, except in the determination of the Basket Amount.
- this Agreement shall be terminated and extinguished upon the earlier of the end of the Survival Period or any Section 10.3 termination of this Agreement. Thereafter, none of Seller, Buyer or any shareholder, partner, officer, director, principal or Affiliate of any of the preceding shall be subject to any liability of any nature whatsoever with respect to any such representation or warranty. Moreover, each party's sole and exclusive remedy for any claim by it for Indemnifiable Losses arising under this Agreement or in connection with or as a result of the transactions contemplated by this Agreement shall be the remedies provided by this Section 11.
- Mitigation. The Indemnified Party shall take all reasonable steps to mitigate all liabilities and claims, including availing itself as reasonably directed by the Indemnifying Party of any defenses, limitations, rights of contribution, claims against third parties and other rights at law, and shall provide such evidence and documentation of the nature and extent of any liability as may be reasonably requested by the Indemnifying Party. Each party shall act in a commercially reasonable manner in addressing any liabilities that may provide the basis for an indemnifiable claim (that is, each party shall respond to such liability in the same manner that it would respond to such liability in the absence of the indemnification provided for in this Agreement). Any request for indemnification of specific costs shall include invoices and supporting documents containing reasonably detailed information about the costs and/or damages for which indemnification is being sought.
- Limitation on Damages. NOTWITHSTANDING ANYTHING TO THE CONTRARY ELSEWHERE IN THIS AGREEMENT, NO PARTY TO THIS AGREEMENT (OR ANY OF ITS AFFILIATES) SHALL, IN ANY EVENT, BE LIABLE TO ANY OTHER PARTY (OR ANY OF ITS AFFILIATES) FOR SPECIAL, CONSEQUENTIAL, PUNITIVE, EXEMPLARY, INCIDENTAL OR INDIRECT DAMAGES, COSTS, EXPENSES, CHARGES OR CLAIMS, PURSUANT TO THIS SECTION 11 except to the extent that Indemnifiable Losses resulting from a third party claim include special, consequential, punitive, exemplary, incidental or indirect damages, costs, expenses, charges or claims of the third party and then, only to the extent of such Indemnifiable Losses, subject however, to all of the limitations set forth in this Agreement.
- 11.10 <u>Treatment of Payments</u>. All payments made pursuant to this <u>Section 11</u> shall be treated as adjustments to the purchase price for the Purchased Assets.

12. <u>GENERAL</u>

12.1 <u>Consents, Approvals and Discretion</u>. Except as herein expressly provided to the contrary, whenever this Agreement requires any consent or approval to be given by either party or either party must or may exercise discretion, the parties agree that such consent or approval shall not be unreasonably withheld or delayed and such discretion shall be reasonably exercised.

12.2 <u>Legal Fees and Costs</u>. In the event either party elects to incur legal expenses to enforce or interpret any provision of this Agreement, the prevailing party will be entitled to recover such legal expenses, including reasonable attorney's fees actually incurred, costs and necessary disbursements, in addition to any other relief to which such party shall be entitled.

12.3 Governing Law; Arbitration.

- (a) The parties agree that this Agreement shall be governed by and construed in accordance with the laws of the State of Georgia, without giving effect to any choice or conflict of law provision or rule thereof that would cause the application of the laws of any other jurisdiction.
- (b) All disputes between the parties arising out of or in any way connected with the execution, interpretation and performance of this Agreement (including the validity, scope and enforceability of this arbitration provision) or the relationship created thereby shall, except as provided herein, be solely and finally settled by a single arbitrator; provided, however, in the event the amount in controversy, whether to an individual claim or dispute or in the aggregate as to multiple claims or disputes between the parties, is five hundred thousand dollars (\$500,000.00) or more, the parties agree to submit such claims or disputes to a board of arbitrators consisting of three arbitrators, as set forth below (the term "Arbitrators" shall refer to the board of arbitrators or the single arbitrator, as applicable). The arbitration proceedings shall be held in Atlanta, Georgia, and except as otherwise may be provided in this Section 12.3(b), the arbitration proceedings shall be conducted in accordance with the Commercial Arbitration Rules (the "AAA Rules") of the American Arbitration Association (the "AAA"). If any Person determines to submit a dispute for arbitration pursuant to this Section 12.3(b), such Person shall furnish the other parties to the dispute with a dated, written statement (the "Arbitration Notice") indicating (i) such Person's intent to commence arbitration proceedings, (ii) the nature, with reasonable detail, of the dispute and (iii) the remedy or remedies such Person will seek. Where the parties use a single arbitrator, within twenty (20) days of the Arbitration Notice, the parties shall select a single arbitrator from a list of members of the AAA's National Panel of Commercial Arbitrators. If the parties use a single arbitrator, that arbitrator must be "neutral." A "neutral" arbitrator shall be a Person who would not be subject to disqualification under rule No. 17 of the AAA Rules. If the parties do not reach agreement on the selection of a single arbitrator within the twenty (20) day period, the AAA shall have the right to make such selection upon the request of any party to the arbitration proceedings. Where the parties use a board of arbitrators, within twenty (20) days of the date of the Arbitration Notice, the Person commencing the arbitration (collectively, the "Petitioner") and the party with whom the Petitioner has its dispute (collectively, the "Respondent") shall each select one qualifying arbitrator (and provide written notice of such selection to the Respondent and Petitioner). A "qualifying" arbitrator is a Person who is not (i) an Affiliate of either the Petitioner or Respondent or (ii) counsel to any such Person at such time. If either the Petitioner or Respondent fails to select a qualifying arbitrator or provide such notice within the twenty (20) day period, the AAA shall have the right to make such selection upon the request of any party to the arbitration proceedings. (Such qualifying arbitrators hereafter may be referred to, respectively, as the "First Arbitrator" and the "Second Arbitrator"). Within ten (10) days following their selection, the First and Second Arbitrator shall select (and provide written notice to the Respondent and the Petitioner of such selection) a third arbitrator (the "Third Arbitrator") from a list of members of the AAA's National Panel of Commercial Arbitrators. The Third Arbitrator must be "neutral." A "neutral" arbitrator shall be a Person who would not be subject to disqualification under rule No. 17 of the AAA Rules. At any time within fifty (50) days after the date of the Arbitration Notice, the Petitioner and Respondent can make discovery requests of the other (including, but not limited to, requests for delivery of documents, production of witnesses for deposition testimony and delivery of interrogatory responses). The recipient of a discovery request shall have twenty (20) days after the receipt of such request to object to any or all portions of such request and make an application to the Arbitrators to limit the scope of such discovery request, and shall respond to any portions of such request not so objected to within twenty (20) days of

the receipt of such request. All objections shall be in writing and shall indicate the reasons for such objections. Within five (5) business days after the end of the period for the submission by the requested party of an application to limit the discovery request, the Arbitrators shall grant or deny such discovery request, in whole or in part, to the extent the Arbitrators determine such discovery is or is not, as the case may be, reasonably necessary to enable the requesting party to obtain information relevant to the dispute without unreasonably burdening the requested party. The requested party shall comply with a discovery request granted by the Arbitrators within twelve (12) business days after such discovery request is granted, or within such longer period as the Arbitrators may determine upon application of the requested party for extension thereof for reasonable cause. Neither party shall be permitted to make more than one application for discovery to the Arbitrators. All depositions shall be taken in the city in which the Person being deposed resides or has its principal place of business, unless otherwise agreed by the parties. The Arbitrators are not authorized to subpoena documents or perform independent investigations. Hearings must commence no later than ninety (90) days following the date of the Arbitration Notice and such hearings shall be conducted for no more than five (5) business days. Each of the Petitioner and the Respondent shall submit a brief, outlining such party's claim for relief or defense to any claim, to the other and to the Arbitrators on or before the tenth (10th) day before the date of the first hearing. Reply briefs must be exchanged and submitted to the Arbitrators on or before the third (3rd) day before the date of the first hearing. The Respondent and Petitioner shall each submit to the Arbitrators within twenty (20) days following the hearing a proposed order for disposing of the claims(s) presented. The final decision of the Arbitrators is due on or before the thirtieth (30th) day following the date of the last hearing. The Arbitrators shall issue a final decision that, in their judgment, is consistent with the terms of this Agreement, applicable Law, and the intent of the parties, as supported by evidence presented by the Petitioner and Respondent in the arbitration proceeding or, if the subject matter of the dispute is not clearly addressed in or determinable under this Agreement, that, in their opinion, would be most fair to the Petitioner and Respondent under the arbitration. The Arbitrators shall have no authority to award any of the types of damages excluded by Section 11.9. The Arbitrators shall be required to provide reasons for their decision. The foregoing time periods and procedural steps may be modified or extended by the Arbitrators in their discretion to the extent they deem necessary to prevent fundamental unfairness; provided that at all times the Arbitrators shall be mindful of the parties' desire for the most expeditious possible resolution of disputes; and provided, further, that a final decision of the Arbitrators shall be rendered within 120 days of the Arbitration Notice. To the extent permissible under applicable Law, the parties agree that the award of the Arbitrators shall be final and shall not be subject to judicial review. Judgment on the arbitration award may be entered and enforced in any court having jurisdiction over the parties or their assets. It is the intent of the parties that the arbitration provisions hereof be enforced to the fullest extent permitted by applicable law, including the Federal Arbitration Act, 9 U.S.C. § 2. Nothing contained in this Exhibit shall prevent the parties from seeking injunctive relief or require arbitration of any issue for which injunctive relief is sought by either party hereto.

- (c) In the event of a breach of any party's obligation to consummate this Agreement or breach of any covenant by any party to this Agreement, the non-breaching party shall be entitled to enforce this Agreement as to such matters by injunctive relief and by specific performance, such relief to be without the necessity of posting a bond, cash or otherwise (unless required by applicable Law).
- 12.4 <u>Benefit</u>: Assignment. Subject to provisions herein to the contrary, this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and assigns. No party may assign this Agreement without the prior written consent of the other party, which consent shall not be unreasonably withheld; provided, however, that a party hereto may assign its interest in this Agreement to an Affiliate, but in such event, the assignor shall be required to remain obligated hereunder in the same manner as if such assignment had not been effected.

4016376.6

12.5 Accounting Date. The Closing of the transactions contemplated hereby shall be deemed to be effective for accounting purposes as of 12:01 a.m. on the calendar day immediately following the Closing Date, unless otherwise agreed in writing by Seller and Buyer. The parties will use commercially reasonable efforts to cause the Closing to be effective as of a month end, with equitable adjustments made to the Purchase Price necessary to give effect to the foregoing. For purposes of allocating and prorating the liabilities, revenues, benefits and burdens associated with the Purchased Assets hereunder (including, by way of illustration and not limitation, the determination of the Price Adjustment Amounts, the preparation of the Closing Balance Sheet, the allocation of PIP payments, Transition Patient Receivables and other revenues under Section 9.1 and the determination of responsibility for a potentially indemnifiable act), all references to the "Closing" and the "Closing Date" shall be deemed to refer to the effective time of the Closing and not the time at which the Closing takes place or the day on which the Closing takes place.

12.6 No Brokerage.

- (a) Seller represents to Buyer that neither Seller nor any of its Affiliates have contracted in any way with any broker in connection with the transactions contemplated hereby. Seller agrees to indemnify Buyer and its Affiliates from and against all loss, cost, damage or expense arising out of claims for fees or commissions of brokers employed or alleged to have been employed by Seller or its Affiliates.
- (b) Buyer represents to Seller that neither Buyer, PPHS, PNI nor their respective Affiliates have contracted in any way with any broker in connection with the transactions contemplated hereby. Buyer agrees to indemnify Seller and its Affiliates from and against all loss, cost, damage or expense arising out of claims for fees or commissions of brokers employed or alleged to have been employed by Buyer, PPHS, PNI and their respective Affiliates.
- 12.7 <u>Cost of Transaction</u>. Whether or not the transactions contemplated hereby shall be consummated and except as otherwise provided herein, the parties agree as follows:
- (a) Except as provided otherwise elsewhere herein, Seller will pay the fees, expenses and disbursements of Seller and its agents, representatives, accountants, and counsel incurred in connection with the subject matter hereof and any amendments hereto;
- (b) Except as provided otherwise elsewhere herein, Buyer shall pay the fees, expenses and disbursements of Buyer and its agents, representatives, accountants, and counsel incurred in connection with the subject matter hereof and any amendments hereto, including the fees and expenses of Sovereign Group, LLC;
- (c) Buyer shall pay all fees and expenses incurred by Seller in relation to any investigation or challenge of the transactions contemplated hereby initiated by the FTC, the Justice Department or the Attorney General of any state on, prior to or after the Closing under antitrust or similar Laws, including all fees and expenses of all parties with respect to filings under the HSR Act (or any other applicable antitrust Law) and all costs and expenses resulting from or relating to any "second request" (or other request or demand for information) issued in connection with any such HSR Act filings (or other filings under any applicable antitrust Law); and
- (d) Buyer will pay the cost of surveys, title insurance, environmental and other inspections, documentary stamps, transfer taxes, recording fees and regulatory approvals.

12.8 Public Announcements; Confidentiality.

- It is understood by the parties hereto that the information, documents and (a) instruments delivered to Buyer by Seller or Seller's agents and the information, documents and instruments delivered to Seller by Buyer or Buyer's agents are of a confidential and proprietary nature. Each of the parties hereto agrees that both prior and subsequent to Closing they will maintain the confidentiality of all such confidential information, documents or instruments delivered to them by each of the other parties hereto or their agents in connection with the negotiation of this Agreement or in compliance with the terms, conditions and covenants hereof and only disclose such information, documents and instruments to their duly authorized officers, directors, representatives and agents. Each of the parties hereto further agrees that if the transactions contemplated hereby are not consummated, they will return all such documents and instruments and all copies thereof in their possession to the other party to this Agreement. Each of the parties hereto recognizes that any breach of this Section would result in irreparable harm to the other parties to this Agreement and their Affiliates and that therefore either Seller or Buyer shall be entitled to an injunction to prohibit any such breach or anticipated breach, without the necessity of posting a bond, cash or otherwise, in addition to all of their other legal and equitable remedies. Nothing in this Section, however, shall prohibit the use of such confidential information, documents or information for the purpose of: (i) Buyer complying with the Hospital Authorities Law; (ii) complying with Georgia open records laws under Georgia Code Annotated §§ 50-18-70 et seq. and the Georgia Open and Public Meetings Law, Georgia Code Annotated §§ 50-14-1, et seq.; or (iii) such governmental filings as in the mutual opinion of Buyer's counsel and Seller's counsel are (A) required by Law or (B) otherwise appropriate. Also, this Section shall not prohibit the disclosure by Buyer or Seller of any information, instruments or documents that are required to be filed with Governmental Entities by Seller or Buyer under applicable securities Laws.
- (b) The parties shall mutually agree upon the date and content of the public announcement of the transactions contemplated by this Agreement. Seller shall participate with Buyer in the initial public announcement of the transactions contemplated by this Agreement.
- 12.9 <u>Waiver of Breach</u>. The waiver by either party of breach or violation of any provision of this Agreement shall not operate as, or be construed to constitute, a waiver of any subsequent breach of the same or other provision hereof.
- 12.10 Notice. Any notice, demand or communication required, permitted, or desired to be given under this Agreement shall be in writing and sent by United States mail, with postage prepaid thereon, certified or registered mail, return receipt requested, or delivered by reputable express delivery service, or personally delivered by private courier and followed by such mailing or express delivery, addressed as follows:

Seller:

Palmyra Park Hospital, Inc.

c/o HCA Inc. One Park Plaza Nashville, TN 37203

Attention: Senior Vice President and Chief Development Officer

with copies to:

HCA Inc.

One Park Plaza Nashville, TN 37203

Attention: General Counsel

and

Waller, Lansden, Dortch & Davis, LLP

511 Union Street, Suite 2700 Nashville, TN 37219-8966 Attn: G. Scott Rayson, Esq.

Buyer:

Hospital Authority of Albany-Dougherty County

417 Third Avenue P.O. Box 1828 Albany, GA 31702

with copies to:

Perry & Walters, LLP 212 North Westover Blvd Post Office Box 71209 Albany, GA 31708-1209

Attn: James E. Reynolds, Jr., Esq.

and

Phoebe Putney Health System, Inc.

P.O. Box 1828 417 Third Avenue Albany, GA 31702-1828

Attention: Thomas S. Chambless, Esq., General Counsel

and

Baudino Law Group, PLC 2600 Grand Avenue, Suite 300

Des Moines, IA 50312

Attention: Robert J. Baudino, Jr., Esq.

or to such other address, and to the attention of such other person or officer, as any party may designate. Any party may change the person and address to which notices or other communications are to be sent to it by giving written notice of any such change in the manner provided herein. Any notice given in accordance with this Section 12.10 by any party to another party shall be deemed to have been given on the earlier of (a) the date on which it is personally received, (b) four (4) days after it is deposited in the United States mail, or (c) the date it is delivered by reputable express delivery service.

- 12.11 <u>Severability</u>. In the event any provision of this Agreement is held to be invalid, illegal or unenforceable for any reason and in any respect, such invalidity, illegality, or unenforceability shall in no event affect, prejudice or disturb the validity of the remainder of this Agreement, unless doing so would result in an interpretation of this Agreement that is manifestly unjust.
- 12.12 No Inferences. Inasmuch as this Agreement is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Agreement has been drafted by or on behalf of such party.

- 12.13 <u>Divisions and Headings</u>. The divisions of this Agreement into sections and subsections and the use of captions and headings in connection therewith are solely for convenience and shall have no legal effect in construing the provisions of this Agreement.
- 12.14 <u>No Third-Party Beneficiaries</u>. The terms and provisions of this Agreement are intended solely for the benefit of Buyer, PPHS, PNI, Seller and their respective permitted successors or assigns, and it is not the intention of the parties to confer, and this Agreement shall not confer, third-party beneficiary rights upon any other Person.
- Agreement, none of the parties (nor any of the parties' respective counsel, accountants or other representatives) has made or is making any representations to any other party (or to any other party's counsel, accountants or other representatives) concerning the consequences of the transactions contemplated hereby under applicable tax Laws or under the Laws governing the Medicare program. Each party has relied solely upon the tax and Medicare advice of its own employees or of representatives engaged by such party and not on any such advice provided by any other party.
- 12.16 Entire Agreement; Amendment. This Agreement supersedes all previous contracts and constitutes the entire agreement of whatsoever kind or nature existing between or among the parties representing the within subject matter and no party shall be entitled to benefits other than those specified herein. As between or among the parties, no oral statement or prior written material not specifically incorporated herein shall be of any force and effect. The parties specifically acknowledge that in entering into and executing this Agreement, the parties rely solely upon the representations and agreements contained in this Agreement and no others. All prior representations or agreements, whether written or verbal, not expressly incorporated herein are superseded unless and until made in writing and signed by all parties hereto. This Agreement may be executed in two or more counterparts each and all of which shall be deemed an original and all of which together shall constitute but one and the same instrument.
- 12.17 Knowledge. Whenever any statement herein or in any schedule, exhibit, certificate or other documents delivered to any party pursuant to this Agreement is made "to its knowledge" or words of similar intent or effect of any party or its representative, such person shall make such statement only if such facts and other information which, as of the date the representation is given, are actually known to the party making such statement, which, with respect to Persons that are corporations, means the knowledge of its executive officers.
- 12.18 <u>FIRPTA Withholding</u>. Seller shall provide Buyer on or prior to the Closing Date a certification pursuant to United States Treasury Department regulation section 1.1445-2 to the effect that Seller is not a foreign person.
- Assets may be owned by Affiliates of Seller and not Seller. Notwithstanding the foregoing, and for purposes of all representations, warranties, covenants and agreements contained herein, Seller agrees that (i) its obligations with respect to any Purchased Assets shall be joint and several with any Affiliate which owns or controls such Purchased Assets, (ii) the representations and warranties herein, to the extent applicable, shall be deemed to have been made by, on behalf of and with respect to, such Affiliates in their ownership capacity, and (iii) Seller has the legal capacity to cause, and it shall cause, any Affiliate which owns or controls any Purchased Assets to meet all of Seller's obligations under this Agreement with respect to such Purchased Assets. Seller hereby waives any defense to a claim made by Buyer under this Agreement based on the failure of any Person who owns or controls the Purchased Assets to be a party to this Agreement.

Property, and WITH NO WARRANTIES, INCLUDING, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, with respect to any personal property which is among the Purchased Assets, any and all of which warranties (both express and implied) Seller hereby disclaim. All of the real and personal property included as Purchased Assets shall be further subject to normal wear and tear on the land, improvements and equipment in the ordinary course of business up to the Closing Date.

12.21 Guarantee.

- absolutely guarantees the prompt payment by Buyer and PNI (and by any assignee by, through or under Buyer or PNI) of each and every financial obligation, Assumed Liability and agreement requiring financial performance by Buyer or PNI (or any such assignee) arising out of, connected with, or related to, this Agreement or any ancillary documents hereto and any assignment, extension, renewal and/or modification thereof (the "Obligations"). The obligation of PPHS under this Section 12.21 is a continuing financial guaranty and shall remain in effect, and the obligations of PPHS shall not be affected, modified or impaired upon the happening from time to time of any of the following events, whether or not with notice or consent of PPHS:
 - (i) The compromise, settlement, release, change, modification or amendment (except to the extent of such compromise, settlement, release, change, modification or amendment) of any or all of the obligations, duties, covenants, or agreements of any party under this Agreement or any ancillary documents hereto;
 - (ii) The extension of the time for performance or payment of any of the Obligations or the extension or the renewal thereof, or
 - (iii) The assignment of any of the Obligations.
- (b) <u>PPHS Obligations</u>. Without limiting the foregoing, the validity of this payment guarantee and PPHS's obligations hereunder shall not be impaired by any event whatsoever, including without limitation any of the following, whether or not with notice to or consent of PPHS:
 - (i) any claim of immunity, defense, set-off or counterclaim (other than full and final payment of the Obligations) on behalf of Buyer or PNI, any other guarantor or any other Person liable for the Obligations;
 - (ii) any change in the time, place or manner of payment or performance, or any release, waiver, indulgence, compromise, settlement, increase, decrease, extension, renewal, acceleration, impairment or termination (voluntary or otherwise) with respect to any or all of the Obligations;
 - (iii) any release, addition, exchange, waiver, indulgence, compromise, or settlement with respect to Buyer or PNI, any other guarantor, any other Person liable for the Obligations, or any of the properties, rights and assets securing the Obligations at any time, if any

(the "Collateral"), or any failure to take, perfect or protect any lien or interest intended as Collateral, if any;

- (iv) any modification, amendment, restatement or replacement (in whole or in part) of any documents, agreements or instruments evidencing, comprising, securing, guaranteeing or otherwise relating to, or executed or delivered in connection with, the Obligations or of any covenants, obligations or terms set forth in such documents, agreements or instruments;
- (v) the voluntary or involuntary liquidation of, sale or other disposition of all or substantially all the assets of, cessation of business of, marshalling of assets and liabilities of, receivership of, financial decline of, insolvency of, bankruptcy of, assignment for the benefit of creditors of, reorganization of, arrangement of, composition with creditors or readjustment of, or other similar proceedings affecting, Buyer, PNI or any of their respective Affiliates or their assets or any allegation or contest of the validity of the Obligations or this guarantee, or the disaffirmance or attempted disaffirmance of the Obligations or this guarantee, in any such proceedings; and/or
- (vi) Seller's failure to notify PPHS of the breach of any provisions under this guarantee, the occurrence of any event of default or breach of any of the Obligations, acceleration of any of the Obligations, the foreclosure or sale of any Collateral, any other action by or on behalf of Seller with respect to the Obligations, or any of the events or circumstances listed in this Section 12.21.
- and not of collection. Accordingly, Seller may enforce this guarantee against PPHS without first making demand on Buyer, PNI or any other Person, or taking action against any Collateral, or instituting collection proceedings upon the Obligations. PPHS's liability for the Obligations is hereby declared to be primary, and not secondary, and each document presently or hereafter executed by Buyer or PNI to evidence or secure the Obligations to Seller is incorporated herein by reference and shall be fully enforceable against PPHS.
- (d) <u>Waivers by PPHS</u>. PPHS hereby knowingly, willingly, and irrevocably waives the following rights, defenses and benefits of law or equity with respect to this guarantee and the Obligations:
 - (i) the issuance of instruments evidencing the Obligations, acceptance of this guarantee, presentment, protest, demand, notice and proof of reliance on this guarantee, and the filing of claims with a court in the event of bankruptcy of Buyer or PNI, any other guarantor or any other Person liable for the Obligations; and
 - (ii) any right to require Seller to marshal assets or proceed first against Buyer or PNI, the Collateral, any other guarantor or any other Person liable for the Obligations.
- hereby is incorporated by reference into this Agreement and will be considered a part hereof as if set forth herein in full. Each Schedule annexed hereto on the date hereof or to be annexed hereto after the date hereof may be updated as necessary or, if Buyer consents (not to be unreasonably withheld), amended on or before the Closing Date. The disclosures in the Schedules of any information shall not be deemed to constitute an acknowledgment that such information is required to be disclosed in connection with the representations and warranties made by Seller in this Agreement or that it is material, nor shall such information be deemed to establish a level or standard of materiality for purposes of this Agreement. The disclosures in the Schedules and in this Agreement set for exceptions to the representations, warranties

and other agreements made by Seller in this Agreement and are intended to qualify such representations, warranties and other agreements. The information set forth in the Schedules with respect to any Section of this Agreement shall also be deemed to qualify each other Section of this Agreement to which such information is applicable (regardless of whether or not such other Section is qualified by reference to a Schedule).

67

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed in multiple originals by their authorized officers, all as of the date and year first above written.

BUYER:

HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY

By:

Ralph Rosenberg Chairman

PPHS:

PHOEBE PUTNEY HEALTH SYSTEM, INC.

Lermel V Griffin

Chairman

PHOEBE NORTH, INC.:

PHOEBE NORTH, INC

Joel Wernick

President and CEO

SELLER:

PALMYRA PARK HOSPITAL, INC.

Grego Gerl

Vice President

CERTIFICATE OF SERVICE

I hereby certify that on the ____ day of January, 2011, the foregoing PLAINTIFF'S MOTION FOR ORDER OF DISMISSAL WITHOUT PREJUDICE, was filed using the CM/ECF system, which will automatically send an electronic notice of such filing to the following counsel of record:

Karin A. Middleton BAUDINO LAW GROUP, PLC 2409 Westgate Drive Albany, Georgia 31707 middleton@baudino.com

Denise Cleveland-Leggett
Amy McCullough
BAUDINO LAW GROUP, PLC
1201 Peachtree Street, NE, Ste. 2020
Atlanta, Georgia 30361
leggett@baudino.com
mccullough@baudino.com

David J. Darrell
Ralph Walker
Jan D. Gibson
BAUDINO LAW GROUP, PLC
2600 Grand Avenue, Suite 300
Des Moines, Iowa 50312
darrell@baudino.com
walker@baudino.com
gibson@baudino.com

Attorneys for Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Hospital System, Inc. Edgar Baughn Wilkin, Jr.
PERRY & WALTERS, L.L.P.
P. O. Box 71209 / 212 N. Westover Blvd.
Albany, Georgia 31708-1209
ewilkin@perrywalters.com

James Sulentic
John Passarelli
KUTAK ROCK LLP
1650 Farnam Street
Omaha, Nebraska 68102
james.sulentic@kutakrock.com
john.passarelli@kutakrock.com

Attorneys for the Hospital Authority of Albany/Dougherty County

Christine A. Hopkinson Georgia Bar No. 142106

EXHIBIT H

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF)
COMMUNITY HEALTH AND	ý
PALMYRA PARK HOSPITAL, INC.)
D/B/A PALMYRA MEDICAL CENTER,	ý
APPELLANTS,	
vs.	<u> </u>
PHOEBE SUMTER MEDICAL CENTER, INC.	APPEALS CASE NO.: [A10D0461] (Appeals case number to be confirmed]
APPELLEE.)
MOTION FOR PERMISSI	A PARK HOSPITAL, INC.'S ION TO WITHDRAW APPEAL k Hospital, Inc. d/b/a Palmyra Medical Center
("Palmyra") and pursuant to Rule 41 (g)(1) of	the Rules of the Court of Appeals for the State of
Georgia and files this Motion for Permission to	Withdraw Appeal pending in the above
captioned matter before this Court. Appellant	has decided not to pursue the pending appeal in
the above captioned matter and therefore respe	ctfully requests that this Court grant its Motion fo
Permission to Withdraw Appeal. For the Cour	t's convenience, an Order granting this Motion is
attached hereto as Exhibit A.	
Respectfully submitted this day of	of, 2011.
	ATTORNEYS FOR APPELLANT PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA
	MEDICAL CENTER

EXHIBIT A

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER, APPELLANTS,))))
vs. PHOEBE SUMTER MEDICAL CENTER, INC. APPELLEE.	APPEALS CASE NO.: [A10D0461] [Appeals case number to be confirmed]
ORDER OF WITHI	DRAWAL OF APPEAL
HAVING READ AND CONSIDERED	Appellant Palmyra Park Hospital, Inc. d/b/a
Palmyra Medical Center's Motion for Permissi	on to Withdraw Appeal and for good cause
shown,	
It is hereby ORDERED that Appellant l	Palmyra Park Hospital, Inc. d/b/a Palmyra
Medical Center's appeal be withdrawn and disr	nissed.
SO ORDERED this day of	, 20
	Judge Court of Appeals for the State of Georgia
	Judge Court of Appeals for the State of Geor

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER))) },
APPELLANTS,))
vs.) APPEALS CASE NO.: [A10D0461]
PHOEBE SUMTER MEDICAL CENTER, INC.	[Appeals case number to be confirmed]
APPELLEE.	3
CERTI	FICATE OF SERVICE
I hereby certify that on the	day of, 20 the foregoing pleading
was sent to the following attorneys of r	ecord via regular first class U.S. Mail at the addresses
listed below:	

EXHIBIT I

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER,))))
APPELLANTS,))
vs. PHOEBE PUTNEY MEMORIAL HOSPITAL, INC. APPELLEE.	APPEALS CASE NO.: [A11A0733] [Appeals case number to be confirmed]
APPELLANT PALMYRA PARK PERMISSION TO W	HOSPITAL, INC.'S MOTION FOR /ITHDRAW APPEAL
COMES NOW Appellant Palmyra Park	Hospital, Inc. d/b/a Palmyra Medical Center and
	ourt of Appeals for the State of Georgia and files
this Motion for Permission to Withdraw Appeal	
this Court. Appellant has decided not to pursue	the pending appeal in the above captioned matter
and therefore respectfully requests that this Cou	rt grant its Motion for Permission to Withdraw
Appeal. For the Court's convenience, an Order	granting this Motion is attached hereto as
Exhibit A.	
Respectfully submitted this day or	f, 2011.
	ATTORNEYS FOR APPELLANT PALMYRA PARK HOSPITAL, INC.

D/B/A PALMYRA MEDICAL CENTER

EXHIBIT A

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER,)))
APPELLANTS,)))
VS.) APPEALS CASE NO.: [A11A0733]
PHOEBE PUTNEY MEMORIAL HOSPITAL, INC.) [Appeals case number to be confirmed]
APPELLEE.))
OR	DER_
HAVING READ AND CONSIDERED	Appellant Palmyra Park Hospital, Inc. d/b/a
Palmyra Medical Center's Motion for Permissio	n to Withdraw Appeal and for good cause
shown,	
It is hereby ORDERED that Appellant Pa	almyra Park Hospital, Inc. d/b/a Palmyra
Medical Center's appeal be withdrawn and dism	issed.
SO ORDERED this day of	, 2011.
	Judge Court of Appeals for the State of Georgia

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER))) L,)
APPELLANTS,))
vs.))) APPEALS CASE NO.: [A11A0733]
PHOEBE PUTNEY MEMORIAL HOSPITAL, INC.) [Appeals case number to be confirmed]
APPELLEE.	<u> </u>
CERTIF	CICATE OF SERVICE
I hereby certify that on thed	lay of, 20 the foregoing pleading
was sent to the following attorneys of re	ecord via regular first class U.S. Mail at the addresses
listed below:	

EXHIBIT J

Date

Marsha Hopkins
Georgia Department of Community Health
Division of Health Planning
2 Peachtree Street
5th Floor
Atlanta, Georgia 30303-3159

RE: Withdrawal of Certificate of Need Application by Palmyra Medical Center, CON # 2008-081

Dear Ms. Hopkins:

This letter serves as Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center's formal request to withdraw its Certificate of Need application, **CON # 2008-081**, which was approved by the Department of Community Health on January 9, 2009, and subsequently appealed by Phoebe Putney Memorial Hospital and Phoebe Sumter Medical Center on February 6, 2009.

Palmyra Medical Center has decided not to pursue its discretionary appeals granted by the Georgia Court of Appeals on August 2, 2010 of the decision rendered by the Superior Court of Sumter County on June 9, 2010 and the decision rendered by the Superior Court of Dougherty County on June 22, 2010. Accordingly, Palmyra Medical Center has filed Motions for Permission to Withdraw its appeal of case numbers [A11A0733] and [A10D0461] [Appeals case numbers to be confirmed] pending before the Georgia Court of Appeals.

Since	rely,		
	•		
cc.	Karin ∆	Middleton	Fea

Schedule 5.2

List of Material Permits and Approvals Needed Prior to Closing

Page 1 of 2

Permits or Approvals Required At or Before Closing:

Food Services Permit Georgia Department of Community Health Environmental Health Division

Permits or Approvals Which Require Notice and/or an Application At or Before Closing:

General Hospital Permit
Georgia Department of Community Health
Healthcare Facilities Regulation Division
(or Georgia Department of Human Resources Office of Regulatory Services)

Radioactive Materials License Georgia Department of Natural Resources Environmental Protection Division Radioactive Materials Program

Pharmacy License(s)
Georgia State Board of Pharmacy

Controlled Substances Registration Certificate(s)
Drug Enforcement and Administration
Diversion Control Program

Clinical Laboratory License(s)
Georgia Department of Community Health
Healthcare Facilities Regulation Division
(or Georgia Department of Human Resources Office of Regulatory Services)

CLIA Certificate of Accreditation/Certificate of Waiver Centers for Medicare and Medicaid Services

X-Ray Registration Georgia Department of Community Health Healthcare Facilities Regulation Division

Certificate of Occupancy
Office of Georgia Safety Fire Commissioner

Schedule 5.2 - page 2 of 2

Elevator Operating Permits Georgia Department of Labor Safety Engineering Division

Boiler Operating Permits Georgia Department of Labor Safety Engineering Division

Accreditation Certificate
The College of American Pathologists

Mammography Accreditation
American College of Radiology

Uniform Licensing Services – Wireless Services Federal Communications Commission

·				
			·	
	,			

Schedule 5.2

List of Material Permits and Approvals Needed Prior to Closing

Page 1 of 2

Permits or Approvals Required At or Before Closing:

Food Services Permit Georgia Department of Community Health Environmental Health Division

Permits or Approvals Which Require Notice and/or an Application At or Before Closing:

General Hospital Permit
Georgia Department of Community Health
Healthcare Facilities Regulation Division
(or Georgia Department of Human Resources Office of Regulatory Services)

Radioactive Materials License Georgia Department of Natural Resources Environmental Protection Division Radioactive Materials Program

Pharmacy License(s) Georgia State Board of Pharmacy

Controlled Substances Registration Certificate(s) Drug Enforcement and Administration Diversion Control Program

Clinical Laboratory License(s)
Georgia Department of Community Health
Healthcare Facilities Regulation Division
(or Georgia Department of Human Resources Office of Regulatory Services)

CLIA Certificate of Accreditation/Certificate of Waiver Centers for Medicare and Medicaid Services

X-Ray Registration Georgia Department of Community Health Healthcare Facilities Regulation Division

Certificate of Occupancy
Office of Georgia Safety Fire Commissioner

Schedule 5.2 - page 2 of 2

Elevator Operating Permits Georgia Department of Labor Safety Engineering Division

Boiler Operating Permits Georgia Department of Labor Safety Engineering Division

Accreditation Certificate
The College of American Pathologists

Mammography Accreditation American College of Radiology

Uniform Licensing Services – Wireless Services Federal Communications Commission

EXHIBIT A

ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AN	D ASSUMPTION	ON AGREEN	MENT ("Assignn	<u>ient</u> ") is made a	ind entered
into as of	, 20 by	and between	n Palmyra Park	Hospital, Inc.,	a Georgia
corporation ("Assignor"), and the	Hospital Author	ority of Albai	ny-Dougherty Co	ounty, a hospita	l authority
organized and existing pursuant	to O.C.G.A. §§	31-7-70 et .	seq. ("Assignee"	'). Capitalized	terms not
otherwise defined herein shall have	e the meanings	s given to suc	ch terms in the A	Asset Purchase	Agreement
(as defined below).					

WITNESSETH:

WHEREAS, Assignor, Assignee, Phoebe Putney Health System, Inc., a Georgia nonprofit corporation, and Phoebe North, Inc., a Georgia nonprofit corporation, have entered into an Asset Purchase Agreement, dated December _____, 2010 (the "Asset Purchase Agreement"), pursuant to which, among other things, Assignee will purchase from Assignor the Purchased Assets; and

WHEREAS, pursuant to the Asset Purchase Agreement, Assignor agreed to transfer to Assignee and Assignee agreed to accept from Assignor all of Assignor's right, title and interest in and to the Assumed Contracts and Assignee agreed to assume the future payment and performance of the Assumed Liabilities.

NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows:

- 1. <u>Assignment</u>. Assignor does hereby assign, grant, convey and transfer to Assignee, and its successors and assigns, all of Assignor's right, title and interest in and to the Assumed Contracts.
- 2. <u>Assumption</u>. Assignee hereby assumes the future payment and performance of the Assumed Liabilities.
- 3. <u>Irrevocable Assignment</u>. Assignor's assignment and Assignee's assumption pursuant to this Assignment are made for good and valuable consideration, the receipt and sufficiency of which are acknowledged, and are coupled with an interest; therefore, the assignment and assumption contained herein are irrevocable.
- 4. <u>Further Assurances</u>. Assignor and Assignee each agrees to take any and all additional actions and to execute, acknowledge and deliver any and all documents which the other party hereto may reasonably request in order to effect the intent and purposes of this Assignment and the transactions contemplated hereby and/or by the Asset Purchase Agreement.
- 5. <u>Remedies.</u> Assignor's and Assignee's remedies with respect to any claim arising from a breach of this Assignment shall be as set forth in the Asset Purchase Agreement.
- 6. Waiver. Any term or condition of this Assignment may be waived at any time by the party which is entitled to the benefit thereof. Any such waiver must be in writing and must be duly

executed by such party. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.

- 7. Amendment. This Assignment may be modified or amended only by a written instrument duly executed by each of the parties hereto.
- 8. No Third-Party Beneficiaries. The terms and provisions of this Assignment are intended solely for the benefit of the parties hereto and their respective successors or assigns, and it is not the intention of the parties to confer third party beneficiary rights upon any other Person.
- 9. <u>Governing Law.</u> THIS ASSIGNMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF GEORGIA WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.
- 10. <u>Disputes</u>. Any dispute under this Assignment shall be settled in accordance with the provisions of Section 12.3 of the Asset Purchase Agreement.
- 11. <u>No Inferences</u>. Inasmuch as this Assignment is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Assignment has been drafted by or on behalf of such party.
- 12. <u>Binding Effect</u>. This Assignment shall be binding upon and will inure to the benefit of the parties and their respective successors and permitted assigns.
- 13. <u>Headings</u>. The headings of this Assignment are for purposes of reference only and shall not limit or otherwise affect the meaning hereof.
- 14. Severability: Invalid Provisions. If any provision of this Assignment is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable, (b) this Assignment will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Assignment will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Assignment a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.
- 15. <u>Inconsistencies with Asset Purchase Agreement</u>. Notwithstanding anything to the contrary contained herein, the terms of this Assignment are subject to the terms, provisions, conditions and limitations set forth in the Asset Purchase Agreement, and this Assignment is not intended to alter the obligations of the parties to the Asset Purchase Agreement. In the event of any inconsistencies between the terms of this Assignment and the terms of the Asset Purchase Agreement, the parties hereto agree that the terms of the Asset Purchase Agreement shall control.
- 16. <u>Counterparts and Facsimile Signatures</u>. This Assignment may be executed simultaneously in any number of counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument. Facsimile signatures on this Assignment shall be deemed to be original signatures for all purposes.

17. No Ratification, Extension or Renewal. This Assignment is not intended to, and does not, in anyway, ratify, extend, or renew any of the Assumed Liabilities or other liabilities that have terminated or expired pursuant to their respective terms or otherwise.

first a	IN WITNESS WHEREOF, Assignor and Assignee have executed this Assignment as of the date cove written.
	ASSIGNOR:
	PALMYRA PARK HOSPITAL, INC.
	By: Gregg A. Gerken Vice President
	ASSIGNEE:
	HOSPITAL AUTHORITY OF ALBANY- DOUGHERTY COUNTY

Ralph Rosenberg Chairman EXHIBIT B

PARTIAL ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS PARTIAL ASSIGNMENT AND ASSUMPTION AGREEMENT ("Assignment") is made and entered into as of, 20 by and between [insert name of Seller or Seller Affiliate that is party to the agreement being partially assigned] ("Assignor"), and the Hospital Authority of Albany-Dougherty County, a hospital authority organized and existing pursuant to O.C.G.A. §§ 31-7-70 et seq. ("Assignee"). Capitalized terms not otherwise defined herein shall have the meanings given to such terms in the Asset Purchase Agreement (as defined below).
WITNESSETH:
WHEREAS, Assignor, an Affiliate of Assignor, Assignee, Phoebe Putney Health System, Inc., a Georgia nonprofit corporation, and Phoebe North, Inc., a Georgia nonprofit corporation, have entered into an Asset Purchase Agreement, dated December, 2010 (the "Asset Purchase Agreement"), pursuant to which, among other things, Assignee will purchase from Assignor and such Assignor Affiliates the Purchased Assets; and
WHEREAS, Assignor entered into that certain [insert name of agreement being partially assigned] effective, (such agreement, together with any other amendments and related arrangements, the "Agreement") with ("Vendor"); and
WHEREAS, pursuant to the Agreement, Vendor provides [and related products and supplies/the right to participate in Vendor's preferred provider network or other applicable reimbursement program] to Assignor in connection with its operation of the Facilities (the ["Subject Supplies"/"Subject Services"]); and
WHEREAS, pursuant to the Asset Purchase Agreement, Assignor agreed to transfer to Assignee all of Assignor's rights under the Agreement with respect to the [Subject Supplies/Subject Services] arising prior to the Closing, and Assignee agreed to assume all of Assignor's rights, obligations and liabilities under the Agreement with respect to the [Subject Supplies/Subject Services] arising from and after the Closing.
NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows:
1. <u>Assignment</u> . Assignor does hereby assign, grant, convey and transfer to Assignee, and its successors and assigns, all of Assignor's right, title and interest in and to the Agreement with respect to the [Subject Supplies/Subject Services] arising from and after the Closing.
2. <u>Assumption</u> . Assignee hereby accepts the foregoing assignment and assumes the future [payment and] performance of any and all obligations under the Agreement with respect to the [Subject Supplies/Subject Services] arising from and after the Closing.

3. <u>Irrevocable Assignment</u>. Assignor's assignment and Assignee's assumption pursuant to this Assignment are made for good and valuable consideration, the receipt and sufficiency of which are

acknowledged, and are coupled with an interest; therefore, the assignment and assumption contained herein are irrevocable.

- 4. <u>Further Assurances</u>. Assignor and Assignee each agrees to take any and all additional actions and to execute, acknowledge and deliver any and all documents which the other party hereto may reasonably request in order to effect the intent and purposes of this Assignment and the transactions contemplated hereby and/or by the Asset Purchase Agreement.
- 5. Remedies. Assignor's and Assignee's remedies with respect to any claim arising from a breach of this Assignment shall be as set forth in the Asset Purchase Agreement.
- 6. Waiver. Any term or condition of this Assignment may be waived at any time by the party which is entitled to the benefit thereof. Any such waiver must be in writing and must be duly executed by such party. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.
- 7. <u>Amendment</u>. This Assignment may be modified or amended only by a written instrument duly executed by each of the parties hereto.
- 8. <u>No Third-Party Beneficiaries</u>. The terms and provisions of this Assignment are intended solely for the benefit of the parties hereto and their respective successors or assigns, and it is not the intention of the parties to confer third party beneficiary rights upon any other Person.
- 9. Governing Law. THIS ASSIGNMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF GEORGIA WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.
- 10. <u>Disputes</u>. Any dispute under this Assignment shall be settled in accordance with the provisions of Section 12.3 of the Asset Purchase Agreement.
- 11. <u>No Inferences</u>. Inasmuch as this Assignment is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Assignment has been drafted by or on behalf of such party.
- 12. <u>Binding Effect</u>. This Assignment shall be binding upon and will inure to the benefit of the parties and their respective successors and permitted assigns.
- 13. <u>Headings</u>. The headings of this Assignment are for purposes of reference only and shall not limit or otherwise affect the meaning hereof.
- 14. Severability; Invalid Provisions. If any provision of this Assignment is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable, (b) this Assignment will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Assignment will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Assignment a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.

- 15. <u>Inconsistencies with Asset Purchase Agreement</u>. Notwithstanding anything to the contrary contained herein, the terms of this Assignment are subject to the terms, provisions, conditions and limitations set forth in the Asset Purchase Agreement, and this Assignment is not intended to alter the obligations of the parties to the Asset Purchase Agreement. In the event of any inconsistencies between the terms of this Assignment and the terms of the Asset Purchase Agreement, the parties hereto agree that the terms of the Asset Purchase Agreement shall control.
- 16. <u>Counterparts and Facsimile Signatures</u>. This Assignment may be executed simultaneously in any number of counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument. Facsimile signatures on this Assignment shall be deemed to be original signatures for all purposes.
- 17. <u>No Ratification, Extension or Renewal</u>. This Assignment is not intended to, and does not, in anyway, ratify, extend, or renew any of the Assumed Liabilities or other liabilities that have terminated or expired pursuant to their respective terms or otherwise.

[Remainder of Page Intentionally Blank]

IN WITNESS WHEREOF, Assignor and Assignee have executed this Assignment as of the date first above written.			
	ASSIGNOR:		
1	[Name of Assignor]		
· .	By; [Name] [Title]		
	ASSIGNEE:		
	HOSPITAL AUTHORITY OF ALBANY- DOUGHERTY COUNTY		

By:

Ralph Rosenberg Chairman EXHIBIT C

GENERAL BILL OF SALE AND ASSIGNMENT

THIS GENERAL BILL OF SALE AND ASSIGNMENT (this "Bill of Sale"), dated as of ________, 20___, is entered into by and between the HOSPITAL AUTHORITY OF ALBANY-DUOGHERTY COUNTY, a hospital authority organized and existing pursuant to O.C.G.A. §§ 31-7-70 et seq. ("Buyer"), and PALMYRA PARK HOSPITAL, INC., a Georgia corporation ("Seller").

WITNESSETH:

WHEREAS, Buyer, Seller, Phoebe Putney Health System, Inc., a Georgia nonprofit corporation, and Phoebe North, Inc., a Georgia nonprofit corporation, entered into an Asset Purchase Agreement, dated as of December _____, 2010 (the "Asset Purchase Agreement"), pursuant to which Seller agreed to sell, assign and transfer, and Buyer agreed to purchase, substantially all of the assets owned or leased by Seller that relate exclusively to Palmyra Medical Center, a general acute care hospital located in Albany, Georgia; and

WHEREAS, Buyer and Seller are entering into this Bill of Sale to satisfy a condition to the Closing of the transactions contemplated by the Asset Purchase Agreement;

NOW, THEREFORE, in consideration of the above premises and for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- 1. Definitions. Capitalized terms used herein and not otherwise defined herein shall have the meanings given to them in the Asset Purchase Agreement.
- 2. Sale and Assignment. Seller hereby irrevocably and unconditionally sells, assigns, conveys, transfers and delivers to Buyer, its successors and assigns forever, all of Seller's right, title and interest in and to the Purchased Assets, free and clear of all Encumbrances other than the Assumed Liabilities and the Permitted Encumbrances, to have and to hold the same and each and all thereof unto Buyer, its successors and assigns forever, to its and their own use and benefit forever. Seller hereby delivers possession of the Purchased Assets to Buyer as of the effective time of the Closing. Notwithstanding the foregoing, this Bill of Sale is not applicable to Seller's sale, assignment, conveyance, transfer or delivery of (a) the Owned Real Property, the conveyance of which is governed by the limited warranty deeds listed in Exhibit A hereto, or (b) the Contracts (including the leases relating to the Leased Real Property), the assignment of which is governed by the assignment agreements being delivered by the parties and their affiliates in connection with the Closing of the transactions contemplated by the Asset Purchase Agreement.
- 3. Disclaimer of Warranties. Except as expressly set forth in Article 4 of the Asset Purchase Agreement, the Facilities and the Purchased Assets are being sold, assigned, conveyed, transferred and delivered by Seller and purchased by Buyer in their physical condition as of the effective time of the Closing, "AS IS, WHERE IS AND WITH ALL FAULTS, DEFECTS, IMPERFECTIONS, LIABILITIES AND NONCOMPLIANCE WITH LAWS" WITH NO WARRANTY OF HABITABILITY OR FITNESS FOR HABITATION," with respect to the Real Property, and "WITH NO WARRANTIES, INCLUDING, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A

PARTICULAR PURPOSE," with respect to any personal property which is among the Purchased Assets, any and all of which warranties (both express and implied) Seller hereby disclaims.

- 4. Further Assurances. Seller hereby agrees to take any and all additional actions and to execute, acknowledge and deliver any and all documents that Buyer may reasonably request in order to effect the intent and purposes of this Bill of Sale and the transactions contemplated hereby and/or by the Asset Purchase Agreement.
- 5. Buyer Remedies. Buyer's remedies with respect to any claim arising from a breach of this Bill of Sale shall be as set forth in the Asset Purchase Agreement.
- 6. Waiver. Any term or condition of this Bill of Sale may be waived at any time by the party which is entitled to the benefit thereof. Any such waiver must be in writing and must be duly executed by such party. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.
- 7. Amendment. This Bill of Sale may be modified or amended only by a written instrument duly executed by each of the parties hereto.
- 8. No Third-Party Beneficiaries. The terms and provisions of this Bill of Sale are intended solely for the benefit of the parties hereto and their respective successors or assigns, and it is not the intention of the parties to confer third party beneficiary rights upon any other Person.
- 9. GOVERNING LAW. THIS BILL OF SALE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF GEORGIA WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.
- 10. Disputes. Any dispute under this Bill of Sale shall be settled in accordance with the provisions of Section 12.3 of the Asset Purchase Agreement.
- 11. No Inferences. Inasmuch as this Bill of Sale is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Bill of Sale has been drafted by or on behalf of such party.
- 12. Binding Effect. This Bill of Sale shall be binding upon and will inure to the benefit of the parties and their respective successors and permitted assigns.
- 13. Headings. The headings of this Bill of Sale are for purposes of reference only and shall not limit or otherwise affect the meaning hereof.
- 14. Severability; Invalid Provisions. If any provision of this Bill of Sale is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable, (b) this Bill of Sale will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Bill of Sale will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Bill of Sale a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.

- 15. Inconsistencies with Asset Purchase Agreement. Notwithstanding anything to the contrary contained herein, the terms of this Bill of Sale are subject to the terms, provisions, conditions and limitations set forth in the Asset Purchase Agreement, and this Bill of Sale is not intended to alter the obligations of the parties to the Asset Purchase Agreement. In the event of any inconsistencies between the terms of this Bill of Sale and the terms of the Asset Purchase Agreement, the parties hereto agree that the terms of the Asset Purchase Agreement shall control.
- 16. Counterparts and Facsimile Signatures. This Bill of Sale may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Facsimile signatures on this Bill of Sale shall be deemed to be original signatures for all purposes.

[Remainder of Page Intentionally Blank]

IN WITNESS WHEREOF, the parties hereto have caused this General Bill of Sale and Assignment to be executed in their names as of the date first above written.

SELLER:

PALMYRA PARK HOSPITAL, INC.

By:

Gregg A. Gerken Vice President

BUYER:

HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY

By:

Ralph Rosenberg Chairman

EXHIBIT A

Limited Warranty Deeds

EXHIBIT D

Exhibit D-1

Owned Real Property

- Tax Number 0000N-00016-001 [MOB]
 2100 Palmyra Rd,
 Albany, GA
- 2. Tax Number 000LL-00009-01D [Parking Lot] 808 14th Ave Albany, GA
- 3. Tax Number 000LL-00009-01E [MOB] 808 14th Ave Albany, GA
- 4. Tax Number 000LL-00009-001 [Main Hospital Parcel] 2000 Palmyra Rd Albany, GA
- 5. Tax Number 000LL-00009-025 [MOB] 1950 Palmyra Rd Albany, GA
- 6a. Tax Number 000LL-00009-4A2 [Condo] 810 13th Ave, Condo Unit A2 Albany, GA
- 6b. Tax Number 000LL-00009-04B [Condo] 810 13th Ave, Condo Unit B Albany, GA
- 6c. Tax Number 000LL-00009-04C [Condo] 810 13th Ave, Condo Unit C Albany, GA
- 6d. Tax Number 000LL-00009-04E [Condo] 810 13th Ave, Condo Unit E Albany, GA
- 7. Tax Number 000LL-00009-07B [MOB] 804 13th Ave Albany, GA
- 8. Tax Number 000LL-00009-007 [MOB] 1912 Arlington Lane Albany, GA

- 9. Tax Number 000LL-00009-07C [Unimproved Land]
 Aberdeen Rd
 Albany, GA
- 10a. Tax Number 000LL-00009-8A1 [Condo]1909 Aberdeen Rd, Condo Unit 1Albany, GA
- 10b. Tax Number 000LL-00009-8A2 [Condo] 1909 Aberdeen Rd, Condo Unit 2 Albany, GA
- 10c. Tax Number 000LL-00009-8A3 [Condo] 1909 Aberdeen Rd, Condo Unit 3 Albany, GA
- 10d. Tax Number 000LL-00009-8A4 [Condo] 1909 Aberdeen Rd, Condo Unit 4 Albany, GA
- 10e. Tax Number 000LL-00009-8A6 [Condo] 1909 Aberdeen Rd, Condo Unit 6 Albany, GA
- 10f. Tax Number 000LL-00009-8A7 [Condo] 1909 Aberdeen Rd, Condo Unit 7 Albany, GA
- 11. Tax Number 000LL-00009-008 [Unimproved Land]
 Aberdeen Rd
 Albany, GA
- 12. Tax Number 000LL-00009-005 [Unimproved Land]
 1901 Arlington Lane
 Albany, GA
- Tax Number 000LL-00009-006 [Unimproved Land]
 1900 Arlington Lane
 Albany, GA
- 14. Tax Number 000LL-00009-009 [Unimproved Land] 1900 Aberdeen Rd Albany, GA
- 15. Tax Number 00000-00039-003 [Unimproved Land]Dawson Rd.Albany, GA

- 16. Tax Number 00000-00039-004 [Unimproved Land]
 Dawson Rd.
 Albany, GA
- 17. Tax Number 00000-00039-005 [Unimproved Land]
 Dawson Rd.
 Albany, GA
- 18. Tax Number 00000-00039-002 [Unimproved Land]
 Dawson Rd.
 Albany, GA
- 19. Tax Number 042B-016 [Unimproved Land]
 Forrester Drive SE
 Dawson, GA

Exhibit D-2

Leased Real Property (Seller or its Affiliate is Tenant)

Lease of entire building located at 810 14th Avenue, Albany, GA ("Gillespie Building")

Lease of space (Suite # 100 and Suite #200) in building located at 2002 Palmyra Road, Albany, GA (the "Palmyra Park Building")

Lease of entire building located 2402 Osler Court, Albany, GA (the "AIM Building") [To be confirmed]

EXHIBIT G

IN THE UNITED STATES DISTRICT COURT MIDDLE DISTRICT OF GEORGIA ALBANY DIVISION

PALMYRA PARK HOSPITAL, INC., d/b/a PALMYRA MEDICAL CENTER,)))
PLAINTIFF,))
vs. PHOEBE PUTNEY MEMORIAL HOSPITAL, INC. AND PHOEBE PUTNEY HEALTH SYSTEM, INC.,)) CIVIL ACTION FILE NO.:) 1:08-CV-000102-WLS))
DEFENDANTS.)

PLAINTIFF'S MOTION FOR ORDER OF DISMISSAL WITHOUT PREJUDICE

NOW COMES Plaintiff Palmyra Park Hospital, Inc., d/b/a Palmyra Medical Center, pursuant to Fed. R. Civ. P. 41(a)(2), and hereby moves this Court for an Order dismissing the above-captioned matter without prejudice. Plaintiff is authorized to inform the Court that Defendants Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Health System, Inc. consent to this motion and the entry of an order of dismissal without prejudice. The parties have agreed to bear their own attorney's fees and costs incurred in this matter, and the parties consent to an Order of Dismissal Without Prejudice on that basis.

A proposed Order of Dismissal Without Prejudice is attached hereto for the Court's convenience.

William H. Hedrick Georgia Bar No. 343250 127 North Westover Boulevard Albany, Georgia 31707 (229) 883-7463 (229) 434-1581 (fax)

W. Ray Persons
Georgia Bar No. 573525
M. Russell Wofford, Jr.
Georgia Bar No. 773002
Christine A. Hopkinson
Georgia Bar No. 142106
KING & SPALDING, LLP
1180 Peachtree Street NE
Atlanta, Georgia 30309
(404) 572-4600
(404) 572-5100 (fax)
email of filer: chopkinson@kslaw.com

Attorneys for Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center

OF COUNSEL:

Jeffrey S. Spigel KING & SPALDING, LLP 1700 Pennsylvania Ave. NW, Suite 200 Washington, D.C. 20006-2706 (202) 737-0500 (202) 626-3737 (fax)

CERTIFICATE OF SERVICE

I hereby certify that on the _____ day of January, 2011, the foregoing PLAINTIFF'S MOTION FOR ORDER OF DISMISSAL WITHOUT PREJUDICE, was filed using the CM/ECF system, which will automatically send an electronic notice of such filing to the following counsel of record:

Karin A. Middleton BAUDINO LAW GROUP, PLC 2409 Westgate Drive Albany, Georgia 31707 middleton@baudino.com

Denise Cleveland-Leggett
Amy McCullough
BAUDINO LAW GROUP, PLC
1201 Peachtree Street, NE, Ste. 2020
Atlanta, Georgia 30361
leggett@baudino.com
mccullough@baudino.com

David J. Darrell
Ralph Walker
Jan D. Gibson
BAUDINO LAW GROUP, PLC
2600 Grand Avenue, Suite 300
Des Moines, Iowa 50312
darrell@baudino.com
walker@baudino.com
gibson@baudino.com

Attorneys for Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Hospital System, Inc. Edgar Baughn Wilkin, Jr.
PERRY & WALTERS, L.L.P.
P. O. Box 71209 / 212 N. Westover Blvd.
Albany, Georgia 31708-1209
ewilkin@perrywalters.com

James Sulentic
John Passarelli
KUTAK ROCK LLP
1650 Farnam Street
Omaha, Nebraska 68102
james.sulentic@kutakrock.com
john.passarelli@kutakrock.com

Attorneys for the Hospital Authority of Albany/Dougherty County

Christine A. Hopkinson Georgia Bar No. 142106

IN THE UNITED STATES DISTRICT COURT MIDDLE DISTRICT OF GEORGIA ALBANY DIVISION

PALMYRA PARK HOSPITAL, INC., d/b/a PALMYRA MEDICAL CENTER, PLAINTIFF,))))
vs. PHOEBE PUTNEY MEMORIAL HOSPITAL, INC. AND PHOEBE PUTNEY HEALTH SYSTEM, INC.;)) CIVIL ACTION FILE NO.:) 1:08-CV-000102-WLS
DEFENDANTS.	<u>}</u>
ORDER OF DISMISSA	L WITHOUT PREJUDICE
Upon the Plaintiff's Motion for Ord	ler of Dismissal without Prejudice, it is hereby
ORDERED that this action is DISMISSED W	VITHOUT PREJUDICE. Plaintiff and Defendants
shall each bear their own attorneys' fees and co	osts, and this case is hereby closed.
SO ORDERED, this day of Jan	uary, 2011.
	The Hon. W. Louis Sands Judge, United States District Court Middle District of Georgia, Albany Division

EXHIBIT H

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF)
COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC.) \
D/B/A PALMYRA MEDICAL CENTER,))
)
APPELLANTS,)
vs.	<i>)</i>)
•) APPEALS CASE NO.: [A10D0461]
PHOEBE SUMTER MEDICAL) [Appeals case number to be
CENTER, INC.) confirmed]
APPELLEE.)
APPELLANT PALMYRA	<u>PARK HOSPITAL, INC.'S</u> N TO WITHDRAW APPEAL
MOTION FOR PERMISSIO	N 10 WITHDRAW ALTEAL
COMES NOW Appellant Palmyra Park	Hospital, Inc. d/b/a Palmyra Medical Center
("Palmyra") and pursuant to Rule 41 (g)(1) of the	ne Rules of the Court of Appeals for the State of
Georgia and files this Motion for Permission to	Withdraw Appeal pending in the above
captioned matter before this Court. Appellant h	as decided not to pursue the pending appeal in
the above captioned matter and therefore respec	tfully requests that this Court grant its Motion for
Permission to Withdraw Appeal. For the Court'	s convenience, an Order granting this Motion is
attached hereto as Exhibit A.	
Respectfully submitted this day o	f, 2011.
	ATTORNEYS FOR APPELLANT PALMYRA
	PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER
	ATRACES A CORPORATE A AMERICAN

EXHIBIT A

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER,)))
APPELLANTS,)))
vs.) APPEALS CASE NO.: [A10D0461]
PHOEBE SUMTER MEDICAL CENTER, INC.) [Appeals case number to be confirmed]
APPELLEE.))
	RAWAL OF APPEAL Appellant Palmyra Park Hospital, Inc. d/b/a n to Withdraw Appeal and for good cause
It is hereby ORDERED that Appellant P	almyra Park Hospital, Inc. d/b/a Palmyra
Medical Center's appeal be withdrawn and dism	issed.
SO ORDERED this day of	, 20
	Judge Court of Appeals for the State of Georgia

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTE))) ER,)		
APPELLANTS,)		
vs.)	APPEALS CASI	E NO.: [A10D0461]
PHOEBE SUMTER MEDICAL CENTER, INC.) [[Appeals case number to be confirmed]	
APPELLEE.)	•	
CERT	IFICATE OF SEI	RVICE	
I hereby certify that on the	day of	, 20	the foregoing pleading
was sent to the following attorneys of	record via regula	ar first class U.S	S. Mail at the addresses
listed below:			

EXHIBIT I

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF COMMUNITY HEALTH AND PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER, APPELLANTS,))))
vs. PHOEBE PUTNEY MEMORIAL HOSPITAL, INC. APPELLEE.)) APPEALS CASE NO.: [A11A0733]) [Appeals case number to be confirmed])
	HOSPITAL, INC.'S MOTION FOR /ITHDRAW APPEAL
COMES NOW Appellant Palmyra Park	Hospital, Inc. d/b/a Palmyra Medical Center and
pursuant to Rule 41 (g)(1) of the Rules of the Co	ourt of Appeals for the State of Georgia and files
this Motion for Permission to Withdraw Appeal	pending in the above captioned matter before
this Court. Appellant has decided not to pursue	the pending appeal in the above captioned matter
and therefore respectfully requests that this Cou-	rt grant its Motion for Permission to Withdraw
Appeal. For the Court's convenience, an Order	granting this Motion is attached hereto as
Exhibit A.	
Respectfully submitted this day or	f, 2011.
	ATTORNEYS FOR APPELLANT PALMYRA PARK HOSPITAL, INC. D/B/A PALMYRA MEDICAL CENTER

EXHIBIT A

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF)	
COMMUNITY HEALTH AND)	
PALMYRA PARK HOSPITAL, INC.)	
D/B/A PALMYRA MEDICAL CENTER,)	
APPELLANTS,)))	
vs.) APPEALS CASE NO.: [A11A0733]	
PHOEBE PUTNEY MEMORIAL HOSPITAL, INC.) [Appeals case number to be confirmed]	
APPELLEE.)	
<u>O</u> R	<u>DER</u>	
HAVING READ AND CONSIDERED	Appellant Palmyra Park Hospital, Inc. d/b/a	
Palmyra Medical Center's Motion for Permission	on to Withdraw Appeal and for good cause	
shown,		
It is hereby ORDERED that Appellant F	almyra Park Hospital, Inc. d/b/a Palmyra	
Medical Center's appeal be withdrawn and disn	nissed.	
SO ORDERED this day of	, 2011.	
	Judge	
	Court of Appeals for the State of Georgia	

IN THE COURT OF APPEALS FOR THE STATE OF GEORGIA

GEORGIA DEPARTMENT OF)	
COMMUNITY HEALTH AND)	
PALMYRA PARK HOSPITAL, INC.)	
D/B/A PALMYRA MEDICAL CENTI	ER,)	
APPELLANTS,)))	
vs.)	ADDRAG GAGDAIO FAILANGGOI
2110722 21107 TELL 103 (02 1 1 1)	APPEALS CASE NO.: [A11A0733]
PHOEBE PUTNEY MEMORIAL)	[Appeals case number to be
HOSPITAL, INC.)	confirmed]
APPELLEE.)	
<u>CERT</u>	TIFICATE OF	SERVICE
I hereby certify that on the	_day of	, 20 the foregoing pleading
was sent to the following attorneys of	record via re	egular first class U.S. Mail at the addresses
listed below:		•

EXHIBIT J

Date

Marsha Hopkins Georgia Department of Community Health Division of Health Planning 2 Peachtree Street 5th Floor Atlanta, Georgia 30303-3159

RE: Withdrawal of Certificate of Need Application by Palmyra Medical Center, CON # 2008-081

Dear Ms. Hopkins:

Sincerely.

This letter serves as Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center's formal request to withdraw its Certificate of Need application, **CON # 2008-081**, which was approved by the Department of Community Health on January 9, 2009, and subsequently appealed by Phoebe Putney Memorial Hospital and Phoebe Sumter Medical Center on February 6, 2009.

Palmyra Medical Center has decided not to pursue its discretionary appeals granted by the Georgia Court of Appeals on August 2, 2010 of the decision rendered by the Superior Court of Sumter County on June 9, 2010 and the decision rendered by the Superior Court of Dougherty County on June 22, 2010. Accordingly, Palmyra Medical Center has filed Motions for Permission to Withdraw its appeal of case numbers [A11A0733] and [A10D0461] [Appeals case numbers to be confirmed] pending before the Georgia Court of Appeals.

garanteen Mil			
cc:	Karin A.	Middleton,	Esq.

EXHIBIT A

ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNM	ENT AND ASSUMPTION	ON AGREEMEN	VT (" <u>Assignment</u>	") is made and entered
into as of	, 20 by	and between Pa	almyra Park Hos	spital, Inc., a Georgia
corporation ("Assignor")	, and the Hospital Author	ority of Albany-I	Dougherty Count	y, a hospital authority
organized and existing	ursuant to O.C.G.A. §§	31-7-70 et seq	. ("Assignee").	Capitalized terms not
otherwise defined herein	shall have the meanings	s given to such t	erms in the Asse	t Purchase Agreement
(as defined below).				

WITNESSETH:

WHEREAS, Assignor, Assignee, Phoebe Putney Health System, Inc., a Georgia nonprofit corporation, and Phoebe North, Inc., a Georgia nonprofit corporation, have entered into an Asset Purchase Agreement, dated December _____, 2010 (the "Asset Purchase Agreement"), pursuant to which, among other things, Assignee will purchase from Assignor the Purchased Assets; and

WHEREAS, pursuant to the Asset Purchase Agreement, Assignor agreed to transfer to Assignee and Assignee agreed to accept from Assignor all of Assignor's right, title and interest in and to the Assumed Contracts and Assignee agreed to assume the future payment and performance of the Assumed Liabilities.

NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows:

- 1. <u>Assignment</u>. Assignor does hereby assign, grant, convey and transfer to Assignee, and its successors and assigns, all of Assignor's right, title and interest in and to the Assumed Contracts.
- 2. <u>Assumption</u>. Assignee hereby assumes the future payment and performance of the Assumed Liabilities.
- 3. <u>Irrevocable Assignment</u>. Assignor's assignment and Assignee's assumption pursuant to this Assignment are made for good and valuable consideration, the receipt and sufficiency of which are acknowledged, and are coupled with an interest; therefore, the assignment and assumption contained herein are irrevocable.
- 4. <u>Further Assurances</u>. Assignor and Assignee each agrees to take any and all additional actions and to execute, acknowledge and deliver any and all documents which the other party hereto may reasonably request in order to effect the intent and purposes of this Assignment and the transactions contemplated hereby and/or by the Asset Purchase Agreement.
- 5. <u>Remedies.</u> Assignor's and Assignee's remedies with respect to any claim arising from a breach of this Assignment shall be as set forth in the Asset Purchase Agreement.
- 6. Waiver. Any term or condition of this Assignment may be waived at any time by the party which is entitled to the benefit thereof. Any such waiver must be in writing and must be duly

executed by such party. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.

- 7. <u>Amendment</u>. This Assignment may be modified or amended only by a written instrument duly executed by each of the parties hereto.
- 8. No Third-Party Beneficiaries. The terms and provisions of this Assignment are intended solely for the benefit of the parties hereto and their respective successors or assigns, and it is not the intention of the parties to confer third party beneficiary rights upon any other Person.
- 9. <u>Governing Law</u>. THIS ASSIGNMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF GEORGIA WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.
- 10. <u>Disputes.</u> Any dispute under this Assignment shall be settled in accordance with the provisions of Section 12.3 of the Asset Purchase Agreement.
- 11. <u>No Inferences</u>. Inasmuch as this Assignment is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Assignment has been drafted by or on behalf of such party.
- 12. <u>Binding Effect</u>. This Assignment shall be binding upon and will inure to the benefit of the parties and their respective successors and permitted assigns.
- 13. <u>Headings</u>. The headings of this Assignment are for purposes of reference only and shall not limit or otherwise affect the meaning hereof.
- 14. Severability: Invalid Provisions. If any provision of this Assignment is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable, (b) this Assignment will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Assignment will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Assignment a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.
- 15. <u>Inconsistencies with Asset Purchase Agreement</u>. Notwithstanding anything to the contrary contained herein, the terms of this Assignment are subject to the terms, provisions, conditions and limitations set forth in the Asset Purchase Agreement, and this Assignment is not intended to alter the obligations of the parties to the Asset Purchase Agreement. In the event of any inconsistencies between the terms of this Assignment and the terms of the Asset Purchase Agreement, the parties hereto agree that the terms of the Asset Purchase Agreement shall control.
- 16. <u>Counterparts and Facsimile Signatures</u>. This Assignment may be executed simultaneously in any number of counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument. Facsimile signatures on this Assignment shall be deemed to be original signatures for all purposes.

17. No Ratification, Extension or Renewal. This Assignment is not intended to, and does not, in anyway, ratify, extend, or renew any of the Assumed Liabilities or other liabilities that have terminated or expired pursuant to their respective terms or otherwise.

IN WITNESS WHEREOF, Assignor irst above written.	r and Assignee have executed this Assignment as of the o
	ASSIGNOR:
	PALMYRA PARK HOSPITAL, INC.
	By: Gregg A. Gerken Vice President
	ASSIGNEE:
	HOSPITAL AUTHORITY OF ALBANY- DOUGHERTY COUNTY

Ralph Rosenberg Chairman EXHIBIT B

PARTIAL ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS PARTIAL ASSIGNMENT AND ASSUMPTION AGREEMENT ("Assignment") is made and entered into as of, 20 by and between [insert name of Seller or Seller Affiliate that is party to the agreement being partially assigned] ("Assignor"), and the Hospital Authority of Albany-Dougherty County, a hospital authority organized and existing pursuant to O.C.G.A. §§ 31-7-70 et seq. ("Assignee"). Capitalized terms not otherwise defined herein shall have the meanings given to such terms in the Asset Purchase Agreement (as defined below).		
WITNESSETH:		
WHEREAS, Assignor, an Affiliate of Assignor, Assignee, Phoebe Putney Health System, Inc., a Georgia nonprofit corporation, and Phoebe North, Inc., a Georgia nonprofit corporation, have entered into an Asset Purchase Agreement, dated December, 2010 (the "Asset Purchase Agreement"), pursuant to which, among other things, Assignee will purchase from Assignor and such Assignor Affiliates the Purchased Assets; and		
WHEREAS, Assignor entered into that certain [insert name of agreement being partially assigned] effective, (such agreement, together with any other amendments and related arrangements, the "Agreement") with ("Vendor"); and		
WHEREAS, pursuant to the Agreement, Vendor provides [and related products and supplies/the right to participate in Vendor's preferred provider network or other applicable reimbursement program] to Assignor in connection with its operation of the Facilities (the ["Subject Supplies"/"Subject Services"]); and		
WHEREAS, pursuant to the Asset Purchase Agreement, Assignor agreed to transfer to Assignee all of Assignor's rights under the Agreement with respect to the [Subject Supplies/Subject Services] arising prior to the Closing, and Assignee agreed to assume all of Assignor's rights, obligations and liabilities under the Agreement with respect to the [Subject Supplies/Subject Services] arising from and		
after the Closing.		
NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows:		
NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency		
NOW, THEREFORE, in consideration of the foregoing premises, the mutual covenants and other agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto covenant and agree as follows: 1. Assignment. Assignor does hereby assign, grant, convey and transfer to Assignee, and its successors and assigns, all of Assignor's right, title and interest in and to the Agreement with respect to		

acknowledged, and are coupled with an interest; therefore, the assignment and assumption contained herein are irrevocable.

- 4. <u>Further Assurances</u>. Assignor and Assignee each agrees to take any and all additional actions and to execute, acknowledge and deliver any and all documents which the other party hereto may reasonably request in order to effect the intent and purposes of this Assignment and the transactions contemplated hereby and/or by the Asset Purchase Agreement.
- 5. Remedies. Assignor's and Assignee's remedies with respect to any claim arising from a breach of this Assignment shall be as set forth in the Asset Purchase Agreement.
- 6. <u>Waiver</u>. Any term or condition of this Assignment may be waived at any time by the party which is entitled to the benefit thereof. Any such waiver must be in writing and must be duly executed by such party. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.
- 7. <u>Amendment.</u> This Assignment may be modified or amended only by a written instrument duly executed by each of the parties hereto.
- 8. <u>No Third-Party Beneficiaries.</u> The terms and provisions of this Assignment are intended solely for the benefit of the parties hereto and their respective successors or assigns, and it is not the intention of the parties to confer third party beneficiary rights upon any other Person.
- 9. Governing Law. THIS ASSIGNMENT SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF GEORGIA WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.
- 10. <u>Disputes</u>. Any dispute under this Assignment shall be settled in accordance with the provisions of Section 12.3 of the Asset Purchase Agreement.
- 11. <u>No Inferences</u>. Inasmuch as this Assignment is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Assignment has been drafted by or on behalf of such party.
- 12. <u>Binding Effect</u>. This Assignment shall be binding upon and will inure to the benefit of the parties and their respective successors and permitted assigns.
- 13. <u>Headings</u>. The headings of this Assignment are for purposes of reference only and shall not limit or otherwise affect the meaning hereof.
- 14. Severability; Invalid Provisions. If any provision of this Assignment is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable, (b) this Assignment will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Assignment will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Assignment a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.

- 15. <u>Inconsistencies with Asset Purchase Agreement</u>. Notwithstanding anything to the contrary contained herein, the terms of this Assignment are subject to the terms, provisions, conditions and limitations set forth in the Asset Purchase Agreement, and this Assignment is not intended to alter the obligations of the parties to the Asset Purchase Agreement. In the event of any inconsistencies between the terms of this Assignment and the terms of the Asset Purchase Agreement, the parties hereto agree that the terms of the Asset Purchase Agreement shall control.
- 16. <u>Counterparts and Facsimile Signatures</u>. This Assignment may be executed simultaneously in any number of counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same instrument. Facsimile signatures on this Assignment shall be deemed to be original signatures for all purposes.
- 17. No Ratification, Extension or Renewal. This Assignment is not intended to, and does not, in anyway, ratify, extend, or renew any of the Assumed Liabilities or other liabilities that have terminated or expired pursuant to their respective terms or otherwise.

[Remainder of Page Intentionally Blank]

IN WITNESS WHER first above written.	OF, Assignor and Assignee have executed this Assignment as of the date
	ASSIGNOR:
	[Name of Assignor]
·	By: [Name] [Title]
	ASSIGNEE:
	HOSPITAL AUTHORITY OF ALBANY- DOUGHERTY COUNTY

By:

Ralph Rosenberg Chairman EXHIBIT C

GENERAL BILL OF SALE AND ASSIGNMENT

THIS GENERAL BILL OF SALE AND ASSIGNMENT (this "Bill of Sale"), dated as of 20_, is entered into by and between the HOSPITAL AUTHORITY OF ALBANY-DUOGHERTY COUNTY, a hospital authority organized and existing pursuant to O.C.G.A. §§ 31-7-70 et seq. ("Buyer"), and PALMYRA PARK HOSPITAL, INC., a Georgia corporation ("Seller").

WITNESSETH:

WHEREAS, Buyer, Seller, Phoebe Putney Health System, Inc., a Georgia nonprofit corporation, and Phoebe North, Inc., a Georgia nonprofit corporation, entered into an Asset Purchase Agreement, dated as of December _____, 2010 (the "Asset Purchase Agreement"), pursuant to which Seller agreed to sell, assign and transfer, and Buyer agreed to purchase, substantially all of the assets owned or leased by Seller that relate exclusively to Palmyra Medical Center, a general acute care hospital located in Albany, Georgia; and

WHEREAS, Buyer and Seller are entering into this Bill of Sale to satisfy a condition to the Closing of the transactions contemplated by the Asset Purchase Agreement;

NOW, THEREFORE, in consideration of the above premises and for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

- 1. Definitions. Capitalized terms used herein and not otherwise defined herein shall have the meanings given to them in the Asset Purchase Agreement.
- 2. Sale and Assignment. Seller hereby irrevocably and unconditionally sells, assigns, conveys, transfers and delivers to Buyer, its successors and assigns forever, all of Seller's right, title and interest in and to the Purchased Assets, free and clear of all Encumbrances other than the Assumed Liabilities and the Permitted Encumbrances, to have and to hold the same and each and all thereof unto Buyer, its successors and assigns forever, to its and their own use and benefit forever. Seller hereby delivers possession of the Purchased Assets to Buyer as of the effective time of the Closing. Notwithstanding the foregoing, this Bill of Sale is not applicable to Seller's sale, assignment, conveyance, transfer or delivery of (a) the Owned Real Property, the conveyance of which is governed by the limited warranty deeds listed in Exhibit A hereto, or (b) the Contracts (including the leases relating to the Leased Real Property), the assignment of which is governed by the assignment agreements being delivered by the parties and their affiliates in connection with the Closing of the transactions contemplated by the Asset Purchase Agreement.
- 3. Disclaimer of Warranties. Except as expressly set forth in Article 4 of the Asset Purchase Agreement, the Facilities and the Purchased Assets are being sold, assigned, conveyed, transferred and delivered by Seller and purchased by Buyer in their physical condition as of the effective time of the Closing, "AS IS, WHERE IS AND WITH ALL FAULTS, DEFECTS, IMPERFECTIONS, LIABILITIES AND NONCOMPLIANCE WITH LAWS" WITH NO WARRANTY OF HABITABILITY OR FITNESS FOR HABITATION," with respect to the Real Property, and "WITH NO WARRANTIES, INCLUDING, THE WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A

PARTICULAR PURPOSE," with respect to any personal property which is among the Purchased Assets, any and all of which warranties (both express and implied) Seller hereby disclaims.

- 4. Further Assurances. Seller hereby agrees to take any and all additional actions and to execute, acknowledge and deliver any and all documents that Buyer may reasonably request in order to effect the intent and purposes of this Bill of Sale and the transactions contemplated hereby and/or by the Asset Purchase Agreement.
- 5. Buyer Remedies. Buyer's remedies with respect to any claim arising from a breach of this Bill of Sale shall be as set forth in the Asset Purchase Agreement.
- 6. Waiver. Any term or condition of this Bill of Sale may be waived at any time by the party which is entitled to the benefit thereof. Any such waiver must be in writing and must be duly executed by such party. A waiver on one occasion shall not be deemed to be a waiver of the same or any other breach, provision or requirement on any other occasion.
- 7. Amendment. This Bill of Sale may be modified or amended only by a written instrument duly executed by each of the parties hereto.
- 8. No Third-Party Beneficiaries. The terms and provisions of this Bill of Sale are intended solely for the benefit of the parties hereto and their respective successors or assigns, and it is not the intention of the parties to confer third party beneficiary rights upon any other Person.
- 9. GOVERNING LAW. THIS BILL OF SALE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF GEORGIA WITHOUT GIVING EFFECT TO ANY CHOICE OR CONFLICT OF LAW PROVISION OR RULE THAT WOULD CAUSE THE APPLICATION OF THE LAWS OF ANY OTHER JURISDICTION.
- 10. Disputes. Any dispute under this Bill of Sale shall be settled in accordance with the provisions of Section 12.3 of the Asset Purchase Agreement.
- 11. No Inferences. Inasmuch as this Bill of Sale is the result of negotiations between sophisticated parties of equal bargaining power represented by counsel, no inference in favor of, or against, either party shall be drawn from the fact that any portion of this Bill of Sale has been drafted by or on behalf of such party.
- 12. Binding Effect. This Bill of Sale shall be binding upon and will inure to the benefit of the parties and their respective successors and permitted assigns.
- 13. Headings. The headings of this Bill of Sale are for purposes of reference only and shall not limit or otherwise affect the meaning hereof.
- 14. Severability; Invalid Provisions. If any provision of this Bill of Sale is held to be illegal, invalid or unenforceable under any present or future Law, (a) such provisions will be fully severable, (b) this Bill of Sale will be construed and enforced as if such illegal, invalid or unenforceable provision had never comprised a part hereof, (c) the remaining provisions of this Bill of Sale will remain in full force and effect and will not be affected by the illegal, invalid or unenforceable provision or by its severance herefrom; and (d) in lieu of such illegal, invalid or unenforceable provision, there will be added automatically as a part of this Bill of Sale a legal, valid and enforceable provision as similar in terms to such illegal, invalid or unenforceable provision as may be possible.

- 15. Inconsistencies with Asset Purchase Agreement. Notwithstanding anything to the contrary contained herein, the terms of this Bill of Sale are subject to the terms, provisions, conditions and limitations set forth in the Asset Purchase Agreement, and this Bill of Sale is not intended to alter the obligations of the parties to the Asset Purchase Agreement. In the event of any inconsistencies between the terms of this Bill of Sale and the terms of the Asset Purchase Agreement, the parties hereto agree that the terms of the Asset Purchase Agreement shall control.
- 16. Counterparts and Facsimile Signatures. This Bill of Sale may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Facsimile signatures on this Bill of Sale shall be deemed to be original signatures for all purposes.

[Remainder of Page Intentionally Blank]

IN WITNESS WHEREOF, the parties hereto have caused this General Bill of Sale and Assignment to be executed in their names as of the date first above written.

SELLER:

PALMYRA PARK HOSPITAL, INC.

By: Gregg A. Gerken Vice President

BUYER:

HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY

By:

Ralph Rosenberg Chairman

EXHIBIT A

Limited Warranty Deeds

EXHIBIT D

Exhibit D-1

Owned Real Property

- Tax Number 0000N-00016-001 [MOB]
 2100 Palmyra Rd,
 Albany, GA
- Tax Number 000LL-00009-01D [Parking Lot]
 808 14th Ave
 Albany, GA
- 3. Tax Number 000LL-00009-01E [MOB] 808 14th Ave Albany, GA
- 4. Tax Number 000LL-00009-001 [Main Hospital Parcel] 2000 Palmyra Rd Albany, GA
- 5. Tax Number 000LL-00009-025 [MOB] 1950 Palmyra Rd Albany, GA
- 6a. Tax Number 000LL-00009-4A2 [Condo] 810 13th Ave, Condo Unit A2 Albany, GA
- 6b. Tax Number 000LL-00009-04B [Condo] 810 13th Ave, Condo Unit B Albany, GA
- 6c. Tax Number 000LL-00009-04C [Condo] 810 13th Ave, Condo Unit C Albany, GA
- 6d. Tax Number 000LL-00009-04E [Condo] 810 13th Ave, Condo Unit E Albany, GA
- 7. Tax Number 000LL-00009-07B [MOB] 804 13th Ave Albany, GA
- 8. Tax Number 000LL-00009-007 [MOB]
 1912 Arlington Lane
 Albany, GA

- 9. Tax Number 000LL-00009-07C [Unimproved Land]
 Aberdeen Rd
 Albany, GA
- 10a. Tax Number 000LL-00009-8A1 [Condo] 1909 Aberdeen Rd, Condo Unit 1 Albany, GA
- 10b. Tax Number 000LL-00009-8A2 [Condo] 1909 Aberdeen Rd, Condo Unit 2 Albany, GA
- 10c. Tax Number 000LL-00009-8A3 [Condo] 1909 Aberdeen Rd, Condo Unit 3 Albany, GA
- 10d. Tax Number 000LL-00009-8A4 [Condo] 1909 Aberdeen Rd, Condo Unit 4 Albany, GA
- 10e. Tax Number 000LL-00009-8A6 [Condo] 1909 Aberdeen Rd, Condo Unit 6 Albany, GA
- 10f. Tax Number 000LL-00009-8A7 [Condo] 1909 Aberdeen Rd, Condo Unit 7 Albany, GA
- 11. Tax Number 000LL-00009-008 [Unimproved Land]
 Aberdeen Rd
 Albany, GA
- 12. Tax Number 000LL-00009-005 [Unimproved Land]
 1901 Arlington Lane
 Albany, GA
- Tax Number 000LL-00009-006 [Unimproved Land]
 1900 Arlington Lane
 Albany, GA
- 14. Tax Number 000LL-00009-009 [Unimproved Land] 1900 Aberdeen Rd Albany, GA
- 15. Tax Number 00000-00039-003 [Unimproved Land]Dawson Rd.Albany, GA

- 16. Tax Number 00000-00039-004 [Unimproved Land]
 Dawson Rd.
 Albany, GA
- 17. Tax Number 00000-00039-005 [Unimproved Land]
 Dawson Rd.
 Albany, GA
- 18. Tax Number 00000-00039-002 [Unimproved Land]
 Dawson Rd.
 Albany, GA
- 19. Tax Number 042B-016 [Unimproved Land]
 Forrester Drive SE
 Dawson, GA

Exhibit D-2

Leased Real Property (Seller or its Affiliate is Tenant)

Lease of entire building located at 810 14th Avenue, Albany, GA ("Gillespie Building")

Lease of space (Suite # 100 and Suite #200) in building located at 2002 Palmyra Road, Albany, GA (the "Palmyra Park Building")

Lease of entire building located 2402 Osler Court, Albany, GA (the "AIM Building") [To be confirmed]

EXHIBIT G

IN THE UNITED STATES DISTRICT COURT MIDDLE DISTRICT OF GEORGIA ALBANY DIVISION

PALMYRA PARK HOSPITAL, INC., d/b/a PALMYRA MEDICAL CENTER, PLAINTIFF,))))
vs. PHOEBE PUTNEY MEMORIAL HOSPITAL, INC. AND PHOEBE PUTNEY HEALTH SYSTEM, INC.,)) CIVIL ACTION FILE NO.:) 1:08-CV-000102-WLS))
DEFENDANTS.)

PLAINTIFF'S MOTION FOR ORDER OF DISMISSAL WITHOUT PREJUDICE

NOW COMES Plaintiff Palmyra Park Hospital, Inc., d/b/a Palmyra Medical Center, pursuant to Fed. R. Civ. P. 41(a)(2), and hereby moves this Court for an Order dismissing the above-captioned matter without prejudice. Plaintiff is authorized to inform the Court that Defendants Phoebe Putney Memorial Hospital, Inc. and Phoebe Putney Health System, Inc. consent to this motion and the entry of an order of dismissal without prejudice. The parties have agreed to bear their own attorney's fees and costs incurred in this matter, and the parties consent to an Order of Dismissal Without Prejudice on that basis.

A proposed Order of Dismissal Without Prejudice is attached hereto for the Court's convenience.

William H. Hedrick Georgia Bar No. 343250 127 North Westover Boulevard Albany, Georgia 31707 (229) 883-7463 (229) 434-1581 (fax)

W. Ray Persons
Georgia Bar No. 573525
M. Russell Wofford, Jr.
Georgia Bar No. 773002
Christine A. Hopkinson
Georgia Bar No. 142106
KING & SPALDING, LLP
1180 Peachtree Street NE
Atlanta, Georgia 30309
(404) 572-4600
(404) 572-5100 (fax)
email of filer: chopkinson@kslaw.com

Attorneys for Palmyra Park Hospital, Inc. d/b/a Palmyra Medical Center

OF COUNSEL:

Jeffrey S. Spigel KING & SPALDING, LLP 1700 Pennsylvania Ave. NW, Suite 200 Washington, D.C. 20006-2706 (202) 737-0500 (202) 626-3737 (fax)

TERMINATION FEE AGREEMENT

THIS TERMINATION FEE AGREEMENT (this "Agreement") is made and entered into as of December 20, 2010 by and among PHOEBE PUTNEY HEALTH SYSTEM, INC., a Georgia nonprofit corporation ("PPHS"), and PALMYRA PARK HOSPITAL, INC., a Georgia corporation ("Palmyra"). PHOEBE NORTH, INC., a Georgia nonprofit corporation ("PNI"), joins this Agreement solely for the purposes set forth in Section 3.2.

WITNESSETH:

- A. Palmyra owns the assets and operations of Palmyra Medical Center, Albany, Georgia (the "Hospital"), and certain related businesses.
- B. Certain affiliates of PPHS lease (or own) and operate certain hospitals and other health care facilities in Dougherty County, Georgia and other counties in southwest Georgia, including Phoebe Putney Medical Center in Albany, Georgia. The Hospital Authority of Albany-Dougherty County (the "Authority"), a hospital Authority organized and existing pursuant to Georgia Code Annotated §§31-7-70 et seq. (the "Act") owns the hospitals currently operated by affiliates of PPHS and leases such hospitals to PPHS affiliates as contemplated by the Act.
- C. PPHS and Palmyra have engaged in negotiations regarding a transaction under which the Authority would acquire substantially all of the assets of the Hospital and certain related businesses, and assume certain liabilities relating to the Hospital and such businesses, all in accordance with the terms and conditions of an asset purchase agreement (the "Asset Purchase Agreement") that would be entered into by and among the Authority, as buyer, Palmyra, as seller, PPHS, as guarantor of the obligations of the Authority and PNI, and PNI, which would agree to employ Palmyra's employees at the closing of the transaction (the "Closing") and manage the Hospital for the benefit of the Authority. The final execution form of the Asset Purchase Agreement is attached hereto as Exhibit A.
- D. As of the date of this Agreement, (1) neither the form of the Asset Purchase Agreement nor the transactions contemplated thereby has been presented to, or approved by, the Authority, (2) the respective boards of directors of PPHS and PNI have authorized such entities' execution and delivery of the Asset Purchase Agreement in the form attached hereto as Exhibit A following the Authority's due authorization, execution and delivery of the Asset Purchase Agreement, but neither PPHS nor PNI has executed the Asset Purchase Agreement, and (3) Palmyra has authorized the Asset Purchase Agreement, Palmyra has executed the Asset Purchase Agreement, and Palmyra has delivered original copies of such executed Asset Purchase Agreement to the Escrow Agent (as defined below) to be held by it in escrow pending the due authorization, execution and delivery of the Asset Purchase Agreement by the Authority and the execution and delivery of the Asset Purchase Agreement by PPHS and PNI.
- E. The parties acknowledge that neither PPHS nor PNI has the authority to negotiate on behalf of the Authority or to bind the Authority, that the form of Asset Purchase Agreement must be presented to and approved by the Authority in order to become a binding obligation of the Authority, and that no assurance can be given that the Authority will approve the Asset Purchase Agreement or the transactions contemplated thereby.
- F. PPHS and PNI have represented to Palmyra that neither of such entities is subject to the Open Meetings Act or the Open Records Act (as defined below). Additionally, pursuant to a

confidentiality agreement, dated as of December 13, 2010, between PPHS and Palmyra (the "Confidentiality Agreement"), PPHS has agreed not to disclose "Company Evaluation Material" (as defined in the Confidentiality Agreement) or the existence of the negotiations or discussions between PPHS and Palmyra. PPHS reiterates its acknowledgement in the Confidentiality Agreement that the disclosure of Company Evaluation Material or the existence of such negotiations or discussions could have a material adverse impact on the Hospital and its related businesses.

- G. PPHS now desires to present the Asset Purchase Agreement to the Authority for its consideration, authorization, approval and, if approved, execution and delivery. PPHS and PNI acknowledge that the Authority is subject to the Open Meetings Act and the Open Records Act and that the presentation of the Asset Purchase Agreement to the Authority could require the public disclosure of such Asset Purchase Agreement, Company Evaluation Material and the existence of the relevant negotiations and discussions between PPHS and Palmyra. Consequently, as a material inducement to Seller to enter into this Agreement and to permit PPHS to present the Asset Purchase Agreement to the Authority, (1) pursuant to Section 3.4, and subject to the terms and provisions of Section 3.4, PPHS will compensate Seller in the event the Authority does not duly authorize, execute and deliver the Asset Purchase Agreement following the Authority's authorization, execution and delivery of the Asset Purchase Agreement, and (2) pursuant to Section 3.3, and subject to the terms and provisions of Section 3.3, PPHS and PNI will commit to execute and deliver the Asset Purchase Agreement promptly following the Authority's due authorization, execution and delivery thereof.
- H. The form of the Asset Purchase Agreement includes provisions under which (1) PPHS agrees to compensate Palmyra (under the circumstances, and subject to the conditions, described in the Asset Purchase Agreement) in the event the transactions contemplated by the Asset Purchase Agreement are not consummated or if the parties to the Asset Purchase Agreement are required to rescind the transactions contemplated by the Asset Purchase Agreement after the consummation of such transactions, and (2) PPHS agrees to guarantee the obligations of the Authority under the Asset Purchase Agreement. As an additional material inducement to Seller to permit PPHS to present the Asset Purchase Agreement to the Authority and to enter into this Agreement, PPHS wishes to confirm that in the event that the Authority, PPHS, PNI and Palmyra execute and deliver the Asset Purchase Agreement, PPHS will be bound by those terms and provisions of the Asset Purchase Agreement applicable to PPHS regardless of whether the Authority shall have duly authorized the Asset Purchase Agreement and regardless of whether the Authority shall have complied with all provisions of the Open Records Act, the Open Meetings Act, its constituent documents, the Authority's rules and procedures regarding open meetings or similar matters and regardless of whether the Asset Purchase Agreement (or any portion thereof) shall be deemed not to be a binding and enforceable obligation of the Authority for any reason.

NOW, THEREFORE, for and in consideration of the premises, and the agreements, covenants, representations and warranties hereinafter set forth, and other good and valuable consideration, the receipt and adequacy of which are forever acknowledged and confessed, the parties hereto agree as follows:

1. <u>REPRESENTATIONS AND WARRANTIES OF PALMYRA</u>.

Palmyra represents and warrants to PPHS the following:

1.1 <u>Incorporation: Qualification and Capacity</u>. Palmyra is a corporation validly existing, duly organized and in good standing under the Laws of the State of Georgia. Palmyra has the requisite corporate power and authority to enter into this Agreement, the Asset Purchase Agreement and the documents described in the Asset Purchase Agreement to which Palmyra would be required to become a party pursuant to the Asset Purchase Agreement (the "Palmyra Ancillary Agreements"), to perform its

IN WITNESS WHEREOF, the parties have caused this Termination Fee Agreement to be executed in multiple originals by their authorized officers, all as of the date and year first above written.

	PPHS:
_	PHOEBE PUTNEY HEALTH SYSTEM, INC. By:
	Lemuel V. Griffin Chairman
	PNI:
	PHOEBE NORTH, INC.
	By: Xe Wsorw
	Joel Wernick President and CEO
	PALMYRA:
	PALMYRA PARK HOSPITAL, INC.
	Ву:
	Gregg Gerken
	Vice President

MANAGEMENT SERVICES AGREEMENT

This Management Services Agreement ("Agreement" or "Management Agreement") is dated as of _______, 2011 by and between the Hospital Authority of Albany-Dougherty County, Georgia, a hospital authority organized and existing pursuant to O.C.G.A. §§ 31-7-70 et seq. ("Authority") and Phoebe North, Inc., a Georgia nonprofit corporation, ("Manager").

WITNESSETH:

WHEREAS, Authority is acquiring substantially all of the assets used exclusively in the operation of [hospital], [address] (the "Hospital") and the other assets used in the support of the Hospital and the Authority will assume certain liabilities relating to the Hospital pursuant to the terms of a certain Asset Purchase Agreement under date of December 21, 2010;

WHEREAS, Phoebe Putney Health System, Inc., ("PPHS") has formed Manager as its affiliate corporation and Manager will employ, effective February 1, 2011, substantially all employees employed in the operation of Hospital; and

WHEREAS, Authority desires to engage Manager to manage the day-to-day operations of Hospital, to provide all employees necessary to operate Hospital and deliver non-physician patient care to patients of Hospital and to provide all services necessary for the operation of Hospital.

NOW, THEREFORE, in consideration of the mutual benefits accruing to the parties hereto, the mutual covenants and agreements herein and other valuable consideration, the adequacy of which the parties hereto acknowledge, the parties hereto agree as follows:

DEFINITIONS

- "Administrative Managers" shall have the meaning set forth in Section in 3.01(a).
- "Administrative Manager Reimbursement" shall have the meaning set forth in Section 3.01(a) hereof.
- "Asset Purchase Agreement" shall mean the agreement entered into on December 21, 2010, by and among Authority, PPHS, Manager, and Palmyra Park Hospital, Inc.
 - "Authority" shall have the meaning set forth in the Preamble.
 - "Authority Accounts" shall have the meaning set forth in Section 5.01(a).
 - "Consultants" shall have the meaning set forth in Section 3.02.
 - "CPAs shall have the meaning set forth in Section 3.04(a).
 - "Employee Compensation" shall have the meaning set forth in Section 3.01(a) hereof.

"GAAP" means United States generally accepted accounting principles and practices as in effect from time to time applied consistently by Manager in connection with the Hospital throughout the periods involved.

"Governmental Entity" means any government or any agency, bureau, board, directorate, commission, court, department, official, political subdivision, tribunal or other instrumentality of any government, whether federal, state or local, domestic or foreign.

"Hospital Authorities Law" means the Hospital Authorities Law, O.C.G.A. §§ 31-7-70 et seq.

"HIPAA" has the meaning set forth in Section 3.13 (f).

"Law" means any constitutional provision, statute, ordinance or other law, rule, regulation, interpretation, judgment, decree or order of any Governmental Entity or any settlement agreement or compliance agreement with any Governmental Entity.

"Manager" has the meaning set forth in the Preamble.

"Operating Period" shall have the meaning set forth in Section 1.01.

"TJC" has the meaning set forth in Section 4.04.

1. TERM

1.01 <u>Term.</u> This Agreement shall be effective from 12:01 a.m. on February 1, 2011 ("Effective Date") until 11:59 a.m. January 31, 2013, unless terminated earlier pursuant to Section 7.03 hereof (the "Operating Period").

2. GENERAL RESPONSIBILITIES OF PARTIES

- 2.01 Appointment and Fiduciary Duty. Authority hereby appoints Manager as the sole and exclusive manager of the day-to-day operations of Hospital during the Operating Period, subject to Authority's control described in Section 2.02 and as required by law. Manager hereby accepts such appointment and shall have authority for the day-to-day operations of Hospital, subject to this Agreement. Manager acknowledges that it shall have a fiduciary responsibility to Authority with regard to the performance of Manager's responsibilities under this Agreement.
- 2.02 <u>General Control of Authority</u>. Authority shall at all times during the Operating Period have ultimate control over the assets and operations of the Hospital. Manager shall perform its duties as described in this Agreement pursuant to policies, procedures, rules and directives, adopted and amended from time to time, by Authority and in accordance with all applicable Laws.
- 2.03 <u>Medical and Professional Matters</u>. All matters relating to the care and treatment of patients requiring professional medical judgment shall remain the responsibility of the Hospital's Medical Staff, and Manager shall not be responsible for nor have any right to make medical

judgments regarding the delivery of medical care by physicians or allied health professionals. However, the Manager and the Medical Staff shall consult with each other regarding such matters where consultation is feasible and in the best interest of the patients of the Hospital. Authority hereby delegates control and authority for overseeing Medical Staff affairs, including monitoring the performance of professional services by the Medical Staff and other licensed personnel to ensure that Hospital maintains high standards of patient care, treatment and related functions to Manager. Authority may at any time revoke Manager's control and authority for overseeing Medical Staff affairs, appointments and actions. The foregoing notwithstanding, the Authority retains control and authority over all appointments to Hospital's Medical staff, the granting of clinical privileges at the Hospital, and any actions taken with respect to Medical Staff members, including appeals of actions taken.

3. DUTIES OF MANAGER

3.01 Personnel

- Manager shall provide or arrange for the provision of all employees necessary for the day-today operation of the Hospital together with certain administrators designated by Manager (the persons identified in such written documents are referred to herein as the "Administrative Managers") necessary for the management of the Hospital. The Administrative Managers as of the Effective Date of this Agreement are listed in Schedule A attached hereto. Any Administrative Manager may be changed from time to time in the sole discretion of Manager. All employees necessary for the day-to-day operation of the Hospital together with the Administrative Managers shall be employees of Manager. Manager shall inform Authority of all compensation and related expenses related to all employees provided by Manager for the day-to-day operation of the Hospital ("Employee Compensation") together with all compensation and related expenses incurred or paid by Manager to or on behalf of the Administrative Managers (the "Administrative Manager Reimbursement"). For purposes of this Section 3.01, both Administrative Manager Reimbursement and Employee Compensation shall be inclusive of salary, fringe benefits, bonuses, severance benefits and approved reimbursable business expenses paid to the Administrative Managers or employees. "Fringe benefits" shall include the employer's contribution to F.I.C.A., unemployment compensation and other employment taxes, workmen's compensation, group life, accident and health insurance premiums, disability coverage insurance premiums, retirement matching, car allowance and any other benefits. Manager shall reasonably allocate Administrative Manager Reimbursement among Hospital and other entities Administrative Managers manage for Manager, PPHS or other affiliates of PPHS.
- (b) Manager shall have sole responsibility for supervising all personnel at the Hospital, and, except as otherwise provided by this Agreement, shall be entitled to take such actions with respect to such personnel as Manager, in its sole discretion, believes are necessary to manage Hospital in accordance with this Agreement. Manager shall have complete authority as to its employees and the Administrative Managers, including but not limited to the following:
- (i) Determining such personnel's wages, benefits and other terms and conditions of employment, including reasonable severance agreements;

- (ii) Supervising all such personnel, including responsibility for all hiring, counseling, disciplinary and termination decisions; and
- (iii) After consulting with and obtaining the approval of Authority with regard to all actions relating to labor organizations, including but not limited to recognition of such organizations and the negotiation, settlement, execution and administration of any labor agreements, to the extent permitted by the terms of such agreements and by the Authority.
- 3.02 <u>Consultants</u>. Manager may hire or retain any consultants, accountants, attorneys or other professional personnel (the "Consultants") which Manager, in its sole discretion, determines are necessary or appropriate to assist Manager in carrying out its duties and responsibilities in accordance with this Agreement; provided however should the total cost to hire or retain any Consultant exceed \$10,000, Manager shall obtain Authority's prior approval of such hire or retention..

3.03 Management

- (a) Manager shall supervise and manage all purchasing, billings, collections, payables, data processing, accounting, cost reporting and other financial matters related to the day-to-day operation of Hospital. Manager shall supervise and manage the business affairs of Hospital to ensure that funds are collected and expended for and on behalf of Authority and in accordance with the terms of this Agreement. Manager shall ensure that the Hospital bond payments are made in a full and in a timely manner to ensure that the bonds do not go into default assuming sufficient funds exist to make such payments. Manager shall notify Authority if insufficient funds exist. Manager shall also comply with all covenants of the bonds issued to support the operations and facilities of the Hospital.
- (b) Manager, acting in Authority's name and as agent for Authority, shall make or direct to be made timely deposits in a bank account that is separate and apart from all other funds and accounts of Authority all receipts and moneys arising from the operation of Hospital in accordance with Section 5.01 of this Agreement, and shall make disbursements from such accounts on behalf of Authority in such amounts and at such times as the same are required. Signatories and approvals as to the amounts on all checks shall be in accordance with the duly adopted policies and procedures of Authority.
- (c) Manager shall be responsible for the performance of all acts reasonably necessary or required in connection with the operation of the Hospital in accordance with Authority's directions.
- (d) Manager shall be responsible for the negotiation and preparation of service and all other contracts determined by Manager in its sole discretion to be necessary or desirable in connection with the operation of Hospital in the usual course of business, including the execution of all physician agreements, medical director agreements, joint venture agreements and provider agreements necessary and appropriate to operate Hospital. All such service contracts shall be in compliance with Internal Revenue Service Revenue Procedure 97-13 or similar procedures or Law

- (e) Manager acknowledges that Authority is a Georgia hospital authority organized and operating under the Hospital Authorities Law with, inter alia, a charitable mission and it has under the Hospital Authorities Law certain responsibilities and obligations, including but not limited to, obligations to provide charity care and indigent care. At all times, Manager in managing Hospital shall follow the charity and indigent care policies of Authority and shall assist Authority in meeting all of Authority's required obligations under the Hospital Authorities Law.
- (f) Manager shall not engage in any financial lending, financing or banking actions that result in a liens, mortgages, lines of credit, security interest or financial obligations in the name of the Hospital without the prior written consent of the Authority. Prior to requesting consent for approval, Manager shall provide a detailed proposal to Authority describing as a minimum the amount of required funding, the purpose of the financing, the strategic plan to generate sufficient revenue to repay such financing and all other alternatives evaluated to obtain sufficient funding.

3.04 Financial Reports, Books and Records.

- (a) Manager shall retain, on behalf of Authority, a firm of independent certified public accountants (the "CPAs") approved by the Authority.
- (b) Manager shall prepare or cause to be prepared for Authority the following financial statements:
- (i) Within ninety (90) days following the close of Hospital's fiscal year, audited annual financial statements for the Hospital for that fiscal year prepared by the CPAs in accordance with GAAP, which statement shall include a balance sheet, a statement of operations and changes in net assets and a statement of cash flow; and
- (ii) Unaudited monthly financial statements including a balance sheet, a statement of operations and changes in net assets and a statement of cash flow.
- (c) Manager shall compile, on a monthly basis, a report which highlights the different cost and revenue centers of Hospital.
- (d) Pursuant to and in compliance with Section 952 of the Omnibus Reconciliation Act of 1980 (PL96-499) to the extent it applies to Authority, Manager covenants to maintain, or cause the maintenance of for a period of at least four (4) years after the rendering of any management services, all books, documents and records of Manager as the same pertain to Manager's actual cost of providing the services contemplated herein and, additionally, to make said documents available to the Secretary of the United States Department of Health and Human Services, the Comptroller General of the United States Government, or any of their duly authorized representatives upon written request from said officials or their designees.
- (e) Manager shall keep all books, accounts and records maintained for the operation of the Hospital open at all reasonable hours and upon appropriate notice for inspection and audit by Authority or any accountants selected by Authority for that purpose.

- (f) Manager shall cause the preparation and filing of all State and Federal cost reports necessary for reimbursement of Authority. Manager shall report to Authority at the end of each fiscal year quarter in reasonable detail on Hospital's participation in Medicare and Medicaid programs.
- (g) Manager shall create, implement and review annual budgets (capital and operating) for the Hospital, which shall be approved annually by Authority and then take all actions determined by Manager in its sole discretion to be necessary to implement such budgets.
- (h) Within ninety (90) days following the close of Hospital's fiscal year, Manager shall cause to be prepared for Authority's review and approval the Community Benefit Report required by the Hospital Authorities Law and upon approval by the Authority cause same to be filed with the Clerk of the Dougherty Superior Court.
- 3.05 Contracts. Subject to Section 3.03 (e) above, Manager may enter into, modify, discharge, make any settlements with respect to or terminate contracts, leases or other documents in the name of Manager or Authority with vendors, physicians or any other persons or entities as may be necessary and appropriate to manage and operate the Hospital pursuant to this Agreement; provided however, if the contract obligates Authority in the amount of \$10,000.00 or more, Manager shall obtain prior approval from a designee of the Authority before obligating the Authority. Manager shall make available to Hospital for the benefit of Authority all those services Manager has contracted specifically for Hospital under the terms any contract it enters into in its own name for the term of this Agreement. In the event of a termination of this Agreement which results in no further relationship between the Authority and Manager, Manager shall cause any contracts it has entered into in its own name to manage and operate the Hospital pursuant to this Agreement for the benefit of Authority to be assigned to Authority. Manager shall perform its obligations and responsibilities under the Asset Purchase Agreement and shall assist Authority in performing its obligations and responsibilities under the Asset Purchase Agreement.
- 3.06 <u>Litigation</u>. Subject to any rights of indemnity which the parties may have under this Agreement, Manager may, in its sole discretion, defend, assert, settle or otherwise dispose of any claims, litigation, judgments or liabilities in connection with the operations of the Hospital during the Operating Period, provided any settlement that impacts Hospital's participation in any state or federal healthcare program shall be subject to the approval and involvement of the Authority.
- 3.07 <u>Manager Liability</u>. Manager shall cause to be maintained policies of insurance or self-insurance as are, in its opinion, appropriate to insure Authority, Manager and Manager's employees and agents against any liability resulting from Manager or Manager's employees' acts or omissions during the Operating Period. Insurance shall include general liability, professional liability, premises liability and Directors and Officers insurance.
- 3.08 <u>Repairs and Maintenance</u>. Manager shall, in the name of and for the account of Authority, as determined by Manager in its sole discretion to be appropriate, negotiate, contract for and supervise the repair and maintenance of the physical property and equipment of the Hospital as shall be necessary to keep and maintain Hospital in good working order and condition.

- 3.09 Government Regulations. On behalf of Authority, Manager shall use its best efforts and all due diligence to ensure that (a) Authority continuously complies with all applicable Laws, including without limitation the state and federal False Claims Act, Civil Monetary Penalty Law, state and federal Anti-Kickback statutes, state and federal self referral prohibitions and applicable Medicare conditions of coverage and/or participation, including ensuring that none of Manager's employees have been sanctioned, debarred or suspended by the Medicare or Medicaid program and (b) Authority retains and maintains in good standing all necessary accreditations, licenses, permits, approvals and authorizations required for the ongoing operation of the Hospital.
- 3.10 <u>Confidentiality of Records</u>. Manager shall protect to the extent permitted by the Hospital Authorities Law and other Laws, the confidentiality of all records of Authority and assist Authority in complying with all applicable Federal, State and local laws and regulations relating to the records of Authority. Manager shall notify Authority of any violation of such Laws upon becoming aware of them. During and following the term of this Agreement, all records of Authority shall remain the property of Authority.
- 3.11 <u>Quality Control</u>. Manager shall cause to be maintained a quality program at the Hospital designed to provide for quality of patient care at the Hospital and Manager shall take all steps necessary to assist Authority to continue meeting Joint Commission accreditation standards.

3.12 Marketing. Manager shall:

- (a) Cause to be maintained a marketing program designed to inform and educate health care professionals and the general public served by Authority of the existence of all services offered by the Hospital.
- (b) Cause to be prepared and distributed such descriptive booklets, brochures or pamphlets as may be necessary to inform health care professionals and members of the public of the nature and requirements of State and Federal reimbursement programs for patients and how the same relate to the services offered at Hospital.

3.13 <u>Legal and Risk Management Services</u>. Manager shall:

- (a) Cause to be prepared or maintained a Patient's Rights Policy and a Patient Care Policy for the Hospital which complies with all applicable Federal and State standards, subject to Authority's approval.
- (b) Cause to be maintained a risk management program at the Hospital, including periodic safety checks, the installation of and review of the performance of functions of safety committees designed to monitor all incidents involving patients, staff or visitors at Authority and the implementation of corrective procedures designed to remedy such incidents:
 - (c) Caused to be reviewed insurance and other liability claims.

- (d) Manager agrees either with its personnel or through the retention of an insurance consultant at least annually to review the insurance requirements of Hospital and prepare a report of his or her findings for the Authority. Manager shall review such report and attempt to obtain, at commercially reasonable premiums, the amount and types of insurance coverage determined in the sole discretion of Manager to be necessary for Authority to maintain.
- (e) As part of such insurance coverage, Manager shall maintain a workers' compensation insurance or self insurance program regarding claims by employees of Manager under any worker's compensation or similar laws, and also from any other claims for personal injury, including death or property damage, which may be made by or on behalf of agents or employees of Manager and the general public, in amounts which are customarily carried, subject to customary deductibles, and against such risks as are customarily insured against by other corporations in connection with the ownership and operation of facilities of similar character and size. All such policies of insurance shall name Authority as an additional insured.
- (f) Cause to be implemented or maintained policies and procedures to comply with the Healthcare Information Portability and Accountability Act ("HIPAA"). Authority and Manager agree to abide by HIPAA, its regulations and the policies implemented by Manager as well as any other applicable federal, state, or local privacy or patient confidentiality Laws. Manager shall recommend from time to time those actions Manager deems necessary for compliance with HIPAA and its applicable regulations, and Authority will promptly cooperate with Manager in implementing such actions. In the course of performing this Agreement Manager will receive or create protected health information (as defined in HIPAA) of/for Hospital and in that regard Manager will be acting as a business associate (as defined in HIPAA) of Authority and Manager will execute a business associate agreement in the form of Exhibit C hereto.
- (g) Cause to be implemented or maintained a compliance program for Hospital in accordance with the state and federal regulations governing compliance program requirements. Manager agrees to refrain, and to cause its employees to refrain, from any activities that could reasonably be expected to cause Manager or Hospital to act in a manner that is inconsistent with such compliance program. Specifically, but without limitation, Manager agrees to abide, and to cause its employees to abide, by Medicare and Medicaid billing and coding requirements, including proper documentation of services. Manager will from time to time undertake actions that Manager deems necessary to ensure that Authority, Hospital and Manager in the operation of Hospital comply with applicable Laws, and, as required Authority will promptly cooperate with Manager's actions under Manager's compliance program for Hospital, including annual internal and external audit to detect and prevent improper activities. Manager's Chief Compliance Officer (CCO) shall be the CCO of Hospital, and during the term of this Agreement will have reporting access and responsibility directly to the Authority.
- 3.14 Other Action. Except as otherwise limited by this Agreement, Manager may take any other actions necessary and appropriate, in its sole opinion and discretion, to manage the Hospital as long as such action is under the authority of the Manager and does not in any material manner obligate Authority beyond the end of the Operating Period. During the initial Operating Period of this Agreement Manager shall not discontinue or eliminate any services offered at Hospital as of the Effective Date of this Agreement and shall ensure that the emergency room is open and operating twenty-four hours, seven days a week during the Operating Period. Any change

in the licensure, payment model, classification or operations of the hospital other than as an acute care PPS hospital shall be subject to the prior written approval of the Authority.

- 3.15 <u>Visits to Hospital</u>. Manager shall cause the Administrative Managers of the Hospital to be on site at Hospital as needed to coordinate, review and supervise the operations of the Hospital and Manager shall cause its other executive personnel to visit Hospital as necessary.
- 3.16 <u>Additional Reports</u>. Manager agrees to prepare, cause to be prepared or otherwise make available reports regarding the operation of Authority as follows:
- (a) Within forty-five (45) days after the end of each fiscal quarter, Manager shall make available to Authority the following information:
- (i) Relevant utilization statistics for Authority for the prior quarter, including but not limited to patient volume for each department or clinical service, payor mix on a departmental or clinical service basis; and
- (ii) Financial statements and budget analysis for Authority for the prior quarter, including but not limited to a report in reasonable detail on Authority Accounts transactions..
- (b) Manager shall provide to Authority, on at least thirty (30) days' notice, or sooner if necessary, to implement such required action, a description of any needed or discontinued services, refinancing proposals, expansion plans or material changes in operating procedures. The proposal shall explain the reasons for the proposed activity.
- (c) Manager agrees to respond, in writing, to any and all recommendations made by Authority's CPAs, which response shall either acknowledge that an audit proposal or recommendation has been implemented or, if not, the reasons why not.

4. LIMITATIONS ON RESPONSIBILITIES OF MANAGER

- 4.01 <u>Risk of Loss</u>. At all times during and following the term of this Agreement, the risk of loss with respect to Hospital operations and assets shall be borne by Authority, subject to the indemnification section below Section 8.
- 4.02 <u>Liabilities</u>. Manager shall not be responsible for liabilities incurred by Authority, either in the name of or on behalf of Authority, in connection with or as a result of the performance by Manager of Manager's duties under this Agreement. All such liabilities shall be solely the obligations of Authority, subject to the indemnification section below Section 8.
- 4.03 <u>Expenses</u>. Manager shall not be responsible for the payment of any expenses incurred by Manager and/or the Administrative Managers on behalf of Authority in the performance of the duties of the Manager and/or the Administrative Managers under this Agreement. All such expenses shall be solely the obligations of Authority.

4.04 <u>Licensure</u>. Manager shall not, without the prior written consent of Authority, act in a manner which could adversely affect the licensure of the Hospital by the State of Georgia as an acute care PPS hospital or the accreditation of the Hospital by The Joint Commission ("TJC"). Within thirty (30) days of receipt by Hospital or Manager of any final report or written assessment concerning the licensure of Hospital by the State of Georgia as an acute care hospital or the accreditation of the Hospital by TJC, Manager shall furnish a copy of such report to the Authority.

5. FINANCIAL MATTERS

- 5.01 <u>Handling and Disposition of Funds</u>. Funds originating from the operation of the Hospital and received by Manager shall be received, handled, managed and disposed as set forth in this Section 5.01.
- (a) Manager shall deposit all funds actually received by it from the operation of Hospital, and all working capital loans advanced for and on behalf of Authority, in a bank account or accounts bearing the name of Authority (hereinafter the "Authority Accounts") in a bank or banks approved by Authority and Manager. Only moneys derived from the operation of the Hospital shall be deposited into the Authority Accounts. The funds on deposit in the Authority Accounts shall not be commingled with any other funds of the Authority or any other accounts or funds of the Manager. Manager shall have no liability or responsibility for any loss resulting from the insolvency, malfeasance or nonfeasance of the bank or banks in which such funds are deposited.
- (b) Authority and Manager shall control Authority Accounts and shall each have the right to make withdrawals from and use Authority Accounts for the purposes of operating the Hospital and performing their obligations hereunder, paying all necessary and reasonable expenses and debts incurred in acquiring and operating Hospital (including but not limited to paying directly to each employee of Manager that employee's individual Employee Compensation), performing the terms and conditions of the Asset Purchase Agreement and paying Manager's fees, as set forth in Section 6.01 of this Agreement, until the expiration or termination of this Agreement, at which time Manager shall resign as co-signatory or authorized signatory for the account. Manager shall document all withdrawals from Authority Accounts and shall provide to the Authority documentation, reasonably acceptable to Authority, of the purpose and amount of each withdrawal. From such accounts, Manager may pay to itself all its fees, as set forth in Section 6.01 of this Agreement, and reimbursements due hereunder.
- 5.02 <u>Manager Not to Pledge Authority Credit</u>. Manager shall not, under any circumstance, pledge the credit of Authority or the revenues of the Hospital, nor shall Manager in the name of, or on behalf of, Authority borrow any money or execute any promissory note, bill of exchange or other obligation, or dispose of any asset of Authority not in the ordinary course of business, without the consent of Authority.
- 5.03 Reimbursement for Advances. Everything done by Manager in the performance of and pursuant to its obligations and duties hereunder, and all its costs and expenses incurred under and pursuant to this Agreement, including but not limited to the provision of all employees necessary for the operation of Hospital and the Administrative Managers, shall be for Authority's account and shall be reimbursed by Authority from the Authority Accounts. In the event that

Manager shall have advanced any funds of its own to third parties in payment of expenses for Authority, Authority agrees to reimburse Manager within fifteen (15) days after receipt of written notice of the advance or advances. Manager shall notify Authority of such advances and shall furnish Authority with receipts evidencing such advancements.

6. COMPENSATION TO MANAGER

Compensation to Manager. During the term of this Agreement, Authority shall pay Manager for its management services a management fee of One Dollar and No Cents (\$1.00) per year as well as a monthly reimbursement to Manager of a sum which is the equivalent of the Administrative Manager Reimbursement and Employee Compensation as defined in Section 3.01. Authority will also reimburse Manager for other general operating expenses and for incidental expenses of employees of Manager (or its affiliated entities) which are directly related to the performance of Manager's management services, including, but not limited to, mileage reimbursement and occasional lodging and meals. To the extent Manager may provide other services not provided for in this Agreement, such as dietary, clinical laboratory, laundry, waste removal, materials management, environmental services, information systems, educational services, biomedical engineering, maintenance and the like, such are not included in the compensation set forth above, but, rather, shall be subject to payment as additional amounts as may be agreed upon in the normal purchasing process of Hospital.

7. **DEFAULT AND TERMINATION**

- 7.01 Non-Financial Default. In the event either party to this Agreement deems the other party to be in default of its obligations hereunder, then said party shall be required to provide notice of the alleged default to the other party, which notice shall contain detailed specifications (which shall be deemed to have occurred on the date the specifications were mailed to the party by certified mail, return receipt requested). The party being charged with the default shall have thirty (30) days in which to:
- (a) Correct the alleged default or provide appropriate assurances to the charging party that the default will be timely corrected; or
- (b) File notice with the charging party that the party against whom the default has been charged denies that the factual matters alleged constitute a default under this Agreement. In the event this latter course is chosen, the parties do hereby covenant and agree to submit such matter to binding arbitration in Dougherty County, Georgia, in accordance with the American Health Lawyers Association Alternative dispute Resolution Service Rules of Procedure for Arbitration and applying the laws of Georgia. Any determination by the arbitrator shall be final and binding upon each of the parties, and judgment thereon may be entered in any court having jurisdiction thereof. The costs shall be borne equally by both parties. During the pendency of any such arbitration and until final judgment thereon has been entered, this Agreement shall remain in full force and effect unless otherwise terminated as provided hereunder.
- 7.02 <u>Financial Default</u>. In the event Authority fails to make the payments or reimbursements due to Manager pursuant to this Agreement, then (but only after thirty (30) days'

written notice of said default, during which period Authority may cure the default), Manager shall be authorized, at its option, to terminate this Agreement.

7.03 <u>Termination</u>.

- (a) In the event of a party's failure to cure a default within the time allowed for curing a default, the non-defaulting party may immediately terminate this Agreement and neither party shall have any further obligations under this Agreement, except those provided for by this Agreement.
 - (b) The parties may mutually agree at anytime to terminate this Agreement.

7.04 Effect of Termination.

- (a) In the event of termination of this Agreement, Manager shall remove itself as a signature to the Authority Accounts and turn over to Authority within ten (10) days following the expiration or termination of this Agreement all business records of the Authority pertaining to Hospital and all medical records shall be maintained by Hospital and shall remain the property of Hospital.
- (b) Within ten (10) days upon the termination or expiration of this Agreement, Administrative Managers shall cease employment with Manager and Authority shall employ the Administrative Managers based upon the terms and conditions of the Administrative Manager's compensation and benefit packages in effect on the Effective Date of this Agreement.

8. INDEMNITY

Except for acts or omissions for which Authority is responsible in whole or in part, Manager agrees to protect, indemnify, defend by counsel reasonably acceptable to Authority and hold Authority, its affiliates and its successors and assigns, and the members, directors, officers, employees, agents and independent contractors (collectively "Indemnitees") free and wholly harmless from and against any and all losses, claims, liens, encumbrances, charges, obligations, damages, remediation, liabilities, demands, suits, causes of action, proceedings, costs and expenses whatsoever (including without limitation consequential damages, interest, fines, claims for the recovery of response costs and reasonable attorneys', engineering consultants', accounting, and other necessary professional fees and expenses), in each case whether known or unknown (to Authority) absolute or contingent, accrued or unaccrued, and including without limitation unasserted claims incurred or suffered by Indemnitees, or any of them, directly or indirectly, caused by acts or omissions of Manager or Manager's employees which occurred during the term of this Agreement or any breach of any representation, warranty, covenant, term or agreement of Manager contained in this Agreement.

9. MISCELLANEOUS

create any obligations, on behalf of the other party, except as otherwise expressly provided herein. Authority and Manager agree to be solely and entirely responsible for their respective acts and for the acts of any of its employees and agents, except as otherwise expressly provided herein.

Compliance with Law, Revenue Procedure 97-13 and Changes in Law. The parties hereto have made all reasonable efforts to ensure that this Agreement represents and memorializes the arrangement between the parties hereto and that it complies with all applicable Law, including, but not limited to, all applicable health care Laws and Internal Revenue Procedure 97-13. The parties are aware that there may be no clear or definitive guidance with respect to the applicability of various health care Laws to the arrangement contained herein. While the parties have attempted to review and analyze all applicable Law in an effort to comply with same, the parties acknowledge that the applicable health care Laws and interpretations thereof, are often vague and are constantly changing. The parties therefore understand and acknowledge that as applicable Law, and interpretations thereof, become more settled including, but not limited to, interpretations of Internal Revenue Service Procedure 97-13, this Agreement may need to be amended in order to comply with such Law, including, but not limited to, Internal Revenue Service Procedure 97-13, or interpretations thereof, as the case may be. In the event there is such a change in Law, or the interpretations thereof, whether by statute, regulation, agency or judicial decision, or otherwise, that has any material effect on any term of this Agreement, or in the event that a party's reputable counsel with experience in health care Law or tax-exempt financing Law determines that any term of this Agreement poses a material risk of violating such Laws, including, but not limited to, Internal Revenue Service Procedure 97-13, then the applicable term(s) of this Agreement shall be subject to renegotiation and either party may request renegotiation of the affected term or terms of this Agreement, upon written notice to the other party, to remedy such condition. In the interim, the parties shall perform their obligations hereunder in full compliance with applicable Law. The parties expressly recognize that upon request for renegotiation, each party has a duty and obligation to the other only to renegotiate the affected term(s) in good faith and, further, the parties expressly agree that their consent to proposals submitted by the other party during renegotiation efforts shall The parties further expressly recognize that in any such not be unreasonably withheld. renegotiation, the relative economics to each of the parties shall be preserved. Should the parties be unable to renegotiate the term or terms so affected so as to bring it/them into compliance with the statute, regulation, decision or interpretation that rendered it/them unlawful or unenforceable within thirty (30) days of the date on which written notice of a desired renegotiation is given, then either party shall be entitled, after the expiration of said thirty (30) day period, to terminate this Agreement upon sixty (60) additional days prior written notice to the other party, provided that such party has received an opinion of reputable legal counsel, which legal counsel and opinion are reasonably acceptable to the other party, that it is more likely than not that this Agreement violates applicable Law, including, but not limited to, Internal Revenue Service Procedure 97-13.

IN WITNESS WHEREOF, the undersigned parties have entered into this Management Services Agreement effective as of the date first written above.

Hospital Authority of Albany-Dougherty County

By:		
Ralph	Rosenberg	

Chairman

Phoebe North, Inc.

By:
Joel Wernick
President and Chief Executive Officer

RESOLUTION OF THE HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY, GEORGIA REGARDING VOLUNTARY PAYMENTS IN LIEU OF TAXES

WHEREAS, the Hospital Authority of Albany-Dougherty County, Georgia (the "Authority") authorized the execution of an Asset Purchase Agreement (the "APA") whereby the Authority will purchase certain assets, including real estate and the existing Palmyra Medical Center ("PMC"), such assets being defined in the APA and referred to in this Resolution as the "Purchased Assets"; and

WHEREAS, Palmyra Park Hospital or PMC has operated in this community for many years as a for-profit business entity and has been subject to payment of taxes to the Tax Director of Dougherty County, Georgia; and

WHEREAS, as a result of the Purchased Assets being acquired by the Authority, the taxes referred to above will no longer be payable; and

WHEREAS, it is the desire of the Authority to have Phoebe Putney Health System, Inc. ("Health System"), or one or more of its affiliates, offset the loss of taxes affecting Dougherty County, the City of Albany and the Dougherty County School System, by entering into an appropriate agreement to make voluntary payments in lieu of taxes;

NOW THEREFORE, be it Resolved, that the Authority hereby urges Health System or one or more of its affiliated entities to enter into an agreement with appropriate authorities for the purpose of making payments to the Tax Director of Dougherty County, Georgia, so as to offset, in an amount equivalent to or greater, the corresponding loss of taxes as a result of the consummation of the transaction described above. A copy of this Declaration is to be promptly transmitted to officials of Health System.

SO RESOLVED this 21st day of December, 2010.

HOSPITAL AUTHORITY OF ALBANY-DOUGHERTY COUNTY, GEORGIA

RALPH S. ROSENBERG, CHAIRMAN